



SIXTEENTH NURSES MASTER AND COMPONENT AGREEMENTS

Between

HER MAJESTY IN RIGHT OF THE PROVINCE OF BRITISH COLUMBIA

and

THE BRITISH COLUMBIA NURSES' UNION

April 1, 2014 – March 31, 2019 Signed – September 3, 2017

UNION CONTACT INFORMATION



TABLE OF CONTENTS

<u>Article</u>	<u>Pa</u>	ige
ARTICLI	E 1 - PREAMBLE AND DEFINITIONS	1
1.01 *1.02 *1.03	Preamble Definitions Misuse of Managerial/Supervisory Authority	1 5
*1.04 *1.05 *1.06	Human Rights Code Sexual Harassment Discrimination and Sexual Harassment Complaint Procedures	9 10
1.07	Respectful Workplace	
2.01 2.02 2.03	Bargaining Agent Bargaining Unit Defined Scope of Agreement	12
2.04 2.05	No Discrimination for Union Activity Individual Contracts	13 13
2.06 2.07 2.08	Contracting Out Notice Boards	13
2.08 2.09 2.10	Accessibility Precedence of Agreement Accessibility of Employer's Regulations	14
2.11 2.12	Strikes and Picket Lines Rights of Witnesses	14 15
2.13 *2.14 2.15	Ambiguity with Master Agreement Amendments Invalid Articles	15
	E 3 - UNION AND PROFESSIONAL SECURITY	
*3.01 *3.02	Membership Membership in Professional Bodies	
*ARTICI	LE 4 - CHECK-OFF OF UNION DUES	16
ARTICLI	E 5 - UNION TO INTERVIEW NEW OR TERMINATING EMPLOYEES	18
ARTICLI	E 6 - EMPLOYER'S RIGHTS	18
6.01 6.02	Rights Supervisory Officials	18 18
ARTICLI	E 7 - STEWARDS	18
7.01 7.02 7.03	Recognition of Stewards Area of Responsibility Duties and Responsibilities	18

ARTICLI	E 8 - GRIEVANCES	20
8.01	Definitions	20
8.02	Failure to Act	20
8.03	Step 1	
8.04	Time Limit	
8.05	Step 2	
8.06	Time Limit	
8.07	Step 3	
8.08	Time Limit	
8.09	Amending Time Limits	
*8.10	Administrative Provisions	
8.11	Dismissal or Suspension Grievance	
8.12	Group Grievance	
8.13	General Application Dispute	
8.14	Investigations and Documents	
8.15	Deviation from Grievance Procedure	
8.16	Technical Objections	
8.17	Nullification	
8.18	Suspensions Pending Investigation	
0.10	Suspensions renaing investigation	27
ARTICLI	E 9 - ARBITRATION	24
9.01	Notification	24
9.02	Composition of the Board of Arbitration	25
9.03	Failure to Appoint	25
9.04	Board Procedure	25
9.05	Decision of the Board	25
9.06	Disagreement on Meaning of Decision	25
9.07	Expenses of Board	
9.08	Amending Time Limits	26
9.09	Resubmission to Arbitration	
9.10	Expedited Arbitration	26
ARTICLI	E 10 - DISMISSAL, SUSPENSION, AND DISCIPLINE	27
10.01	Dismissal	27
10.02	Suspension	
10.03	Rejection During Probation	
10.04	Return to Former Classification	
10.05	Unauthorized Absence	
10.06	Assessment of Professional Competence.	
10.07	Disciplinary Procedure	
10.08	Unjust Dismissal or Suspension	
10.00	Disciplinary Records	
10.10	Appraisal Reports	
AKIIULI	E 11 - SENIORITY	30
*11.01	Definitions	30

*11.02	Seniority List	. 31
11.03	Loss of Seniority	
11.04	Re-employment	
11.05	Bridging of Service	
11.06	Training Credits	
11.07	Service Seniority Tie Breaker	
11.07	Service Semonty The Dreaker	55
ARTICL	E 12 - POSTINGS, TRANSFERS AND SECONDMENT	34
*12.01	Postings	34
*12.02	Selection Panels	
12.03	Interview Expenses	35
12.04	Union Observer	
12.05	Notice of Promotions	36
12.06	Notification of Unsuccessful Applicants	36
*12.07	Review of Staffing Decisions	
12.08	Relocations	
12.09	Transfers Without Posting	
12.10	Demotions and Transfers	
12.11	Secondment	
ARTICL	E 13 - LAYOFF AND RECALL OF REGULAR EMPLOYEES	39
13.01	Principles	39
13.02	Changing Status	40
*13.03	Jurisdictional Units and Seniority Blocks	40
13.04	Redundant Positions	41
13.05	Pre Layoff	41
*13.06	Identification of a Supernumerary Employee	42
13.07	Transfer Within Geographic Location	43
13.08	Identification of Employee to Receive Notice of Layoff	44
13.09	Notice of Layoff and Employee Options Upon Receipt of Notice	44
13.10	Severance Pay	47
13.11	Orientation Period	47
*13.12	Relocation Expenses	48
13.13	Joint Committee	
13.14	Savings Clause	48
13.15	Copies of Correspondence	49
ARTICL	E 14 - HOURS OF WORK	49
14.01	Hours of Work	49
14.02	Work Day	
14.02	Meal Breaks	
14.03	Rest Periods	
14.04	Consecutive Days Off	
14.05	Attendance Security	
17.00		50
ARTICL	E 15 - SHIFT AND ROTATION	51

15.01	Shift Designation	51
15.02	Shift System	51
*15.03	Shift Differential	51
15.04	Hours and Overtime Rates	52
15.05	Changes in Schedules	52
ARTICL	E 16 - OVERTIME	52
16.01	Definitions	52
16.02	Authorization and Application of Overtime	53
16.03	Overtime Entitlement	
16.04	Recording of Overtime	54
16.05	Sharing of Overtime	54
16.06	Overtime On Designated Paid Holidays	54
16.07	Overtime On Travel Status	
16.08	Overtime Meal Allowance	55
16.09	No Layoff to Compensate for Overtime	
16.10	Right to Refuse Overtime	
16.11	Call-out Provision	
16.12	Pre-Arranged Overtime	
16.13	Rest Interval	
16.14	Transportation	
*16.15	Contact While Off-Duty	56
ARTICL	E 17 - PAID STATUTORY HOLIDAYS	56
17.01	Designated Paid Holidays	56
17.02	Holidays Falling on Saturday or Sunday	
17.03	Holiday Falling on a Day of Rest	
17.04	Holiday Falling on a Scheduled Work Day	
17.05	Holiday Coinciding with a Day of Vacation	
17.06	Christmas or New Year's Day Off	58
17.07	Shift Workers	58
17.08	Conversion of Hours and Part-Time Entitlement	58
ARTICL	E 18 - ANNUAL VACATION	59
*18.01	Entitlement	59
18.02	Scheduling of Vacation	60
18.03	Salary During Vacation Period	62
18.04	Approved Leave of Absence With Pay During Vacation	62
*18.05	Vacation Leave on Retirement	62
18.06	Separation Allowance	63
ARTICL	E 19 - ILLNESS, INJURY, REHABILITATION AND WORKERS	
COMPI	ENSATION	63
19.01	Short-Term Illness and Injury and Long Term Disability	63
19.02		
	Rehabilitation Committee	05

ARTICLE	E 20 - SPECIAL LEAVE	64
20.01	Bereavement Leave	64
20.02	Leave for Professional Association/College Duties	64
*20.03	Union Business or Public Duties	
20.04	Leave for Court Appearances	67
20.05	Reduction of Benefits	67
20.06	Leave for Committee Meetings	
20.07	Her Majesty's Forces	68
20.08	Civil Emergency	69
20.09	Elections	69
20.10	General Leave	69
20.11	Household Emergency	69
20.12	Family Illness	
20.13	Leave for Medical and Dental Care	70
*20.14	Special Leave	71
20.15	Special Leave Limitation	72
20.16	Service Breaks	72
20.17	Donor Leave73	
20.18	Extended Child Care Leave	73
20.19	Compassionate Care Leave	73
ARTICLI	E 21 - MATERNITY/PARENTAL/ADOPTION LEAVE	74
21.01	Maternity Leave	74
21.01	Parental Leave	
21.02	Maximum Combined Entitlement	
21.03	Benefit Waiting Period Allowance	
21.01	Maternity Leave Allowance	
21.05	Parental Leave Allowance	
21.00	Pre-Placement Adoption Leave	
21.08	Benefits Continuation	
21.09	Deemed Resignation	
21.10	Entitlements Upon Return to Work	
21.11	Maternity and/or Parental Leave &/or Pre-Adoption Leave Allowance Repayment.	
21.12	Benefits Upon Layoff	
ARTICLI	E 22 - OCCUPATIONAL HEALTH AND SAFETY	78
*22.01	Copies of Regulations	
*22.01	Safe Workplace	
*22.02	Right to Refuse Unsafe Work	
*22.03	Communicable Disease	
*22.04 *22.05	Occupational Health and Safety Committees	
*22.03 22.06	Disagreement Over Recommendations	
22.00	Injury Pay Provisions	
22.07	Transportation of Accident Victims	
*22.08	Provincial Joint Occupational Health and Safety Committee (PJOSH)	
22.09	Strain Injury Prevention	
22.10		04

*22.11	Occupational First Aid Requirements and Courses	82
* 22.12	Violence In The Workplace	84
*22.13	Investigation of Accidents	86
ARTICLI	E 23 - EDUCATION POLICY	86
23.01	Educational Leave and Assistance	86
23.02	Joint Education Committee and Jurisdiction	88
*23.03	Ancillary Bonuses	
23.04	Academic Bonuses	90
ARTICLI	E 24 - NO EMPLOYEE LIABILITY	91
24.01	Actions of Patients	91
24.02	Personal Property Damage	91
ARTICLI	E 25 - HEALTH AND WELFARE	91
25.01	Basic Medical Insurance	91
*25.02	Extended Health Care Plan	92
25.03	Dental Plan	92
*25.04	Group Life	92
25.05	Air Travel Insurance	
25.06	Medical Examination	94
25.07	Employment Insurance	
25.08	Employee and Family Assistance Program	94
ARTICLI	E 26 - ACCUMULATION OF TIME	95
ARTICLI 26.01	Compensation for Statutory Holidays, Overtime and Standby	95
26.01 *26.02	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time	95 95
26.01	Compensation for Statutory Holidays, Overtime and Standby	95 95
26.01 *26.02 26.03	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time	95 95 95
26.01 *26.02 26.03 ARTICLI *27.01	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures	95 95 95 96
26.01 *26.02 26.03 ARTICLI	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES	95 95 95 96 96
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates	95 95 95 96 96 97 98
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility	95 95 95 96 96 97 98 98
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures 2 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay	95 95 95 96 96 97 98 98 99
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification	95 95 95 96 96 97 98 98 99 99
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion	95 95 95 96 96 97 98 98 98 99 99 100
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution	95 95 95 96 96 97 98 98 98 99 99 100 100
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures 2 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Appointments	95 95 95 96 96 97 98 98 99 99 100 100 101
26.01 *26.02 26.03 ARTICLJ *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09 *27.10	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures E 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Appointments Vehicle Allowance	95 95 95 96 96 97 98 98 99 100 100 101 101
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09 *27.10 *27.11	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures. 27 - PAYMENT OF SALARIES AND ALLOWANCES . Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Appointments Vehicle Allowance	95 95 95 96 96 97 98 98 98 99 100 100 101 101 101
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09 *27.10 *27.11 27.12	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Appointments Vehicle Allowance Return to Higher-Paid Classification	95 95 95 96 96 97 98 98 99 99 100 100 101 101 102 102
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09 *27.10 *27.11 27.12 *27.13	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures 2 7 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Appointments Vehicle Allowance Meal Allowance Return to Higher-Paid Classification Accommodation, Board and Lodging Allowance	95 95 95 96 97 98 98 99 100 100 101 101 101 102 102 103
26.01 *26.02 26.03 ARTICLI *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09 *27.10 *27.10 *27.11 27.12 *27.13 27.14	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures 27 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Substitution Temporary Appointments Vehicle Allowance Return to Higher-Paid Classification Accommodation, Board and Lodging Allowance Standby Provisions	95 95 95 96 96 97 98 98 98 99 100 100 101 101 101 102 102 103
26.01 *26.02 26.03 ARTICLI *27.01 *27.02 *27.03 *27.04 *27.05 27.06 27.07 27.08 27.09 *27.10 *27.11 27.12 *27.13	Compensation for Statutory Holidays, Overtime and Standby Compensation for Surplus Time Requests for Time Off Procedures 2 7 - PAYMENT OF SALARIES AND ALLOWANCES Salaries Pay Period Increment Dates Increments - Eligibility In-hiring Rates of Pay Salary on Promotion or Reclassification Salary on Demotion Temporary Substitution Temporary Appointments Vehicle Allowance Meal Allowance Return to Higher-Paid Classification Accommodation, Board and Lodging Allowance	95 95 95 96 96 97 98 98 99 99 100 101 101 101 102 102 103 103

27.17	Death Benefit	104
27.18	Formula for Hourly, Daily and Partial Month Calculations	105
27.19	Child Care Expenses	106
27.20	Travel Expense Reimbursement	
ARTICL	E 28 - CLASSIFICATION AND RECLASSIFICATION	107
28.01	Classification Grade Descriptors	107
*28.02	Job Evaluation Plan	
28.03	New Classifications or Proposed Changes in Existing Classifications	
*28.04	Classification Appeal Procedure	
*28.05	Classification Referee	
ARTICL	E 29 - AUXILIARY EMPLOYEES	111
29.01	Terms of Employment	
*29.02	Seniority on Applying for Regular Positions	
29.03	Seniority	
29.04	Loss of Seniority	
29.05	Layoff and Recall	
*29.06	Application of Agreement	
*29.07	Health and Welfare	
29.08	Paid Holidays	
29.09	Annual Vacation	
29.10	Eligibility Requirements for Benefits	
29.11	Weekly Indemnity	
*29.12	Increments	
29.13	Changing Status	121
*29.14	Conversion of Auxiliary Employees	
ARTICL	E 30 - GENERAL CONDITIONS	123
30.01	Parking	123
30.02	Termination of Employment	
30.03	Standing Joint Committee	
*30.04	Indemnity	
30.05	Payroll Deductions	
30.06	Political Activity	
*30.07	Personnel Files	
30.08	Employment of Relatives	
30.09	Copies of Agreement	127
30.10	Technical Information	
30.11	Excessive Work Load	127
30.12	Previous Policies and Letters of Understanding	
30.13	Statement of Duties and Responsibilities	127
30.14	Clothing and Uniforms	128
ARTICL	E 31 - TERM OF AGREEMENT	128
*31.01	Expiration of Agreement	128
-	1 U	

*31.02	Notice to Bargain	129
31.03	Commencement of Bargaining	
31.04	Changes in Agreement	
*31.05	Effective Date of Agreement	129
*APPEN	NDIX 1 COMPONENT STRUCTURE & JURISDICTIONAL AREAS	131
*APPEN	DIX 2 EXCLUSIONS	131
	NDIX 3 SHORT TERM ILLNESS AND INJURY PLAN AND LONG TERM BILITY PLAN	133
Part I – S	Short Term Illness and Injury Plan	133
Part II - I	Long Term Disability Plan	139
Part III -	Rehabilitation Committee	149
*APPEN	DIX 4 MINISTRIES, BOARDS AND AGENCIES	152
*APPEN	NDIX 5 SIXTEENTH NURSES MASTER WAGE SCHEDULES	153
APPENI	DIX 6 ADDRESSING WORKPLACE VIOLENCE	172
*INFOR	RMATION APPENDIX A APPOINTMENT TO THE PUBLIC SERVICE	172
INFORM	MATION APPENDIX B ADVANCE PAYMENT OF GROUP LIFE BENEFI'	Г 173
*INFOR	RMATION APPENDIX C JOB SHARING	174
*INFOR	RMATION APPENDIX D DEFERRED SALARY LEAVE PROGRAM	177
	RMATION APPENDIX E DISCRIMINATION AND HARASSMENT IN THE	
	RMATION APPENDIX F RE: 15 th MASTER AGREEMENT MOU#21 TANT LEADER	184
*INFOR	RMATION APPENDIX G RE: 15 th MASTER AGREEMENT APPENDIX 6.	186
*MEMC	DRANDUM OF UNDERSTANDING #1 NEW FACILITIES	188
MEMO	RANDUM OF UNDERSTANDING #2 RESIGNATION PROMPTING	188
	DRANDUM OF UNDERSTANDING #3 SAFEGUARDING VULNERABLE LE	188
-	ORANDUM OF UNDERSTANDING #4 PRE-EMPLOYMENT ELIGIBILITY CHECK-OFF ADMINISTRATION	
*MEMC	DRANDUM OF UNDERSTANDING #5 TRANSFERS WITHOUT POSTING	190
*MEMC	ORANDUM OF UNDERSTANDING #6 RE: HUMAN RIGHTS CODE	190
*MEMC	DRANDUM OF UNDERSTANDING #7 REVISIONS OF BENEFITS	192

MEMORANDUM OF UNDERSTANDING #8 EMPLOYMENT EQUITY 1	93
*MEMORANDUM OF UNDERSTANDING #9 NURSING ADVOCACY COMMITTEE PROFESSIONAL RESPONSIBILITY 1	
MEMORANDUM OF UNDERSTANDING #10 AUXILIARY EMPLOYEES HIRED FO DOWNSIZING/CLOSING FACILITY1	
*MEMORANDUM OF UNDERSTANDING #11 LABOUR ADJUSTMENT 1	.97
*MEMORANDUM OF UNDERSTANDING #12 TRANSITION COMMITTEE 1	.99
*MEMORANDUM OF UNDERSTANDING #13 ROLE OF THE REHABILITATION	
COMMITTEE	00
*MEMORANDUM OF UNDERSTANDING #14 BOARD AND LODGING AND RELOCATION EXPENSES	201
Part I Board and Lodging Regulations	.03
Part II Relocation Expenses	
*MEMORANDUM OF UNDERSTANDING #15 RE: RECRUITMENT AND RETENTION INCENTIVE ADJUSTMENT	11
*MEMORANDUM OF UNDERSTANDING #16 REVIEW OF WORKPLACE SAFETY TRAINING AND GUIDANCE	
*MEMORANDUM OF UNDERSTANDING #17 RE: EXTENDED WORK DAY/COMPRESSED WORK WEEK 2	14
*MEMORANDUM OF UNDERSTANDING #18 BASELINE STAFFING INFORMATION	18
*MEMORANDUM OF UNDERSTANDING #19 RECRUITMENT AND RETENTION COMMITTEE	18
*MEMORANDUM OF UNDERSTANDING #20 RE: ECONOMIC STABILITY DIVIDEND	19
SUBJECT INDEX - MASTER AND COMPONENT AGREEMENTS	22

HOSPITAL SERVICES NURSES COMPONENT

Note: Changes to the **Fifteenth** Master Agreement are indicated in the **Sixteenth** Master Agreement with **bold type** (except for **Headings**) and an asterisk (*); or an asterisk (*) only where language has been deleted.

ARTICLE 1 - PREAMBLE AND DEFINITIONS

1.01 Preamble

(a) The principals to this Agreement recognize the right of an individual to uninterrupted, skillful, and efficient care. It is obligatory upon the Employer and its employees to strive for the efficient operation and maintenance of our hospitals, clinics, and other institutions or facilities. Also, in consideration of the maintenance of harmonious relations and settled conditions of employment, and recognizing the mutual value of joint discussions on all matters pertaining to working conditions, hours of work, and salaries, the Employer and the Union agree to abide by the terms set out in this Agreement. The Union further agrees that it will at all times instruct its members to act in accordance with the terms contained in this Agreement. The Employer agrees, in the exercise of the functions of management, that the provisions of this Agreement will be carried out. The Union and the Employer agree that the two parties will be bound by the *Human Rights Code* of British Columbia.

(b) Wherever the masculine or singular is used, the same shall be construed as meaning the feminine or plural unless otherwise specifically stated.

*1.02 Definitions

"Aboriginal Community Government" means an Indian Band Council duly constituted under the federal Indian Act or an aboriginal, or Métis governing body authorized under the terms of a treaty duly ratified by the provincial and/or federal governments or federal legislated self-government arrangements.

"bargaining unit" as referred to hereafter in this Agreement shall be understood to be the unit for collective bargaining described in Section 4(a) of the *Public Service Labour Relations Act* and includes all employees licensed or registered under the *Health Professions Act*, including those employees eligible for license or registration who are employed in that professional capacity in the ministries, boards and agencies of the Government of British Columbia.

"basic pay" means the same pay as an employee would receive if they were on vacation, e.g., they would receive their salary as set forth in the Appendix **5** plus any applicable educational bonus and/or isolation allowance.

"Component" means an occupational bargaining unit as specified in Appendix 1 and means a Subsidiary Collective Agreement as that term is used in the *Public Service Labour Relations Act*.

"continuous employment and continuous service" means uninterrupted employment in the Public Service of British Columbia subject to the provisions of Article 11.

"demotion" means a change from an employee's position to one with a lower maximum salary.

"Deputy Minister" means a Deputy of a Minister or the Head of a Ministry as defined in this Agreement.

"employee" means a member of the bargaining unit and includes:

(a) "regular employee" - means an employee who is employed for work which is of a continuous full-time or part-time nature.

(b) "casual auxiliary employee" means an employee who is employed for work which is not of a continuous nature, such as:

(1) seasonal positions;

(2) positions created to carry out special projects or work which is not continuous;

(3) temporary positions created to cover employees on vacation, short term illness and injury leave, education leave, compassionate leave, or other leave;

(4) temporary positions created by special programs such as the summer student employment program, winter works programs for the unemployed, emergencies such as floods or other special temporary programs;

(c) "on-call auxiliary employee" means an employee who is employed at irregular intervals and specifically to meet operational requirements during depleted staff situations.

"employee" does not include:

(a) persons excluded by Section 1 of the *Public Service Labour Relations Act*;

(b) incumbents of managerial or confidential positions mutually excluded by the parties to this Agreement and recorded in Appendix 2.

"Employer" means the Government of British Columbia represented by the BC Public Service Agency (BCPSA) including any agency or person authorized to exercise the authority of the BCPSA.

"Interim Permit Nurse" means a nurse who is a graduate of an approved nursing program but is not registered with the College of Registered Psychiatric Nurses of British Columbia or the **College of** Registered Nurses of British Columbia.

"headquarters or geographic location" is that area within a radius of 32 kilometers of where an employee ordinarily performs their duties. When employees are relocated, this definition may be redefined where exceptional circumstances such as unusual road conditions exist.

"holiday" means the 24-hour period commencing at 00:01 hours of a day designated as a paid holiday in this Agreement.

"layoff" means the termination of an employee's employment because of lack of work or because of a discontinuation of a function or program.

"leave of absence without pay" means to be absent from duty with permission but without pay.

"Ministry" means a part of the Government of British Columbia specified in Appendix 4.

"pay" means rate of compensation for the job.

"professional association" as referred to hereafter in this Agreement shall be understood to be the College of Registered Psychiatric Nurses of British Columbia and/or the **College of** Registered Nurses of British Columbia.

"promotion" means a change from an employee's position to one with a higher maximum salary level.

"resignation" means a voluntary written notice that an employee is terminating their services on the day specified therein.

"rest day" (scheduled day off) in relation to an employee, means a day other than a holiday on which an employee is not ordinarily required to perform the duties of their position. This does not include employees on leave of absence.

"relocation" refers to the movement of an employee from one geographic location to another.

"shift" means the normal consecutive working hours scheduled for each nurse (regular full-time, regular part-time, or auxiliary) which occur in any 24-hour period. In each 24-hour period there will normally be three shifts, namely, day, afternoon, and night shifts.

(a) "Day shift" means a shift in which the major portion occurs between 0700 and 1500 hours;

(b) "Afternoon shift" means a shift in which the major portion occurs between 1500 and 2300 hours;

(c) "Night shift" means a shift in which the major portion occurs between 2300 and 0700 hours.

"shift employee" is one who regularly operates within a pattern of various work periods with designated and regular specific change-over (duty-relief) times, and/or an employee who works a shift other than the day shift.

"spouse" shall include common-law (both same sex and opposite sex), husband or wife.

"Steward" shall mean an employee within the Employer's service elected or appointed by the Union or its members to represent the Union and its members.

"transfer" (lateral) refers to the movement of an employee from one position to another which does not constitute a demotion or promotion.

"travel status" with respect to an employee means absence of the employee from their headquarters or geographic location on Government business with the approval of the Employer, but travel status does not apply to an employee temporarily assigned outside their headquarters or geographic location and who is in receipt of board and lodging allowance.

"Union" as referred to hereafter in this Agreement shall be understood to mean the British Columbia Nurses' Union holding certification for this bargaining unit.

"Union representative" shall mean a member of the staff of the Union or an employee approved for absence under Clause 20.03 who is designated, due to the absence of the Union representative, to substitute and perform the duties normally performed by a member of the staff of the Union.

"work day" is a period of 24 consecutive hours commencing with the starting time of an employee's shift.

"workplace" means the mutually agreed vicinity and/or facility in which the employee's duties are generally performed.

"work schedule" means the roster of work hours and days to meet the annual hours of work.

*1.03 Misuse of Managerial/Supervisory Authority

Misuse of managerial/supervisory authority takes place when a person who supervises or is in a position of authority exercises that authority in a manner which serves no legitimate work purpose and which ought reasonably be known to be inappropriate.

Misuse of managerial/supervisory authority does not include action occasioned through the exercise, in good faith, of the Employer's managerial/supervisory rights and responsibilities. Nor does it include a single incident of a minor nature where the harm, by any objective standard is minimal.

Where the allegation is based on a matter for which another dispute resolution mechanism exists, then this process shall not be utilized.

If an employee does not present a complaint within the prescribed time limits, or if the President of the Union or their designate does not present a complaint to the next higher level within the prescribed time limits, the complaint will be deemed to have been abandoned.

Procedures

(a) If there is an allegation of misuse of managerial/supervisory authority, the employee will approach their supervisor or the first level of excluded manager, not involved in the matter, for assistance in resolving the issue within 30 days of the alleged occurrence. The supervisor/manager will investigate the allegation and take steps to resolve the concern as appropriate within 30 days of the issue being raised by the employee. The supervisor/manager will discuss the proposed resolution with the employees directly involved. The employees directly involved may have a steward present during these discussions.

(b) If the proposed resolution is not acceptable, the complainant may refer the matter through the Union in writing to the Deputy Minister or their designate within 30 days of receiving the supervisor's/manager's response or when the response was due. The written statement will provide full particulars of the allegation including:

- the name(s) of individual(s) involved; and
- the specific actions and dates of the alleged misuse of managerial/supervisory authority; and
- names of witnesses; and
- an explanation as to why it should be considered misuse of authority; and
- the remedy sought; and
- an outline of the steps which have been taken to resolve the matter in (a) above.

These particulars will form the basis of the Deputy Minister's consideration and/or investigation and will be those which are placed before the panel should the matter proceed pursuant to (d). The Deputy Minister shall provide the respondent with a copy of the complaint.

(c) The Deputy Minister or their designate will acknowledge, in writing, receipt of the written statement, including the particulars, and when required, will have the matter investigated and will take such steps as may be required to resolve the matter. The Union and the employees involved in the allegation shall be advised in writing of any proposed resolution or other response within 30 days of providing notice to the Deputy Minister.

(d) Where the matter is not resolved pursuant to (c), the Union may refer the matter to the Joint Mediation/Arbitration Panel within 30 days of receiving the Deputy Minister's response or when the response was due. The Panel will be comprised of one member each from the Employer and the Union, and a Chairperson who shall be appointed jointly by the Parties. By mutual agreement, the Parties may appoint two members each to the Panel.

The referral to the Panel will include the written statement presented at step (b) above and the Deputy Minister's response.

(e) The Panel will review the **written statement** and the **Deputy Minister's** response. The Panel may make a decision based on these documents or if it determines that there is no basis for the complaint or there are insufficient particulars, the panel will dismiss the case.

Where the Panel determines there is sufficient reason to conduct a hearing, the Panel shall hear and determine any dispute between the Parties over interpretation, application or any alleged violation of this clause.

Hearings shall be conducted **so as to give those involved a fair hearing. The Panel may admit any evidence deemed necessary or appropriate.** The Panel may:

(1) make findings of fact;

(2) decide if, on the facts, misuse of managerial/supervisory authority has occurred;

- (3) attempt to mediate a resolve;
- (4) dismiss the complaint.

The decision of the Panel shall be final and binding and consistent with the terms of the collective agreement.

(f) Where the complaint is found to be frivolous, vindictive or vexatious, the Employer may take appropriate action which may include discipline.

(g) Disciplinary action taken by the Employer which is consistent with the recommendations of the majority of the Panel shall not form the basis of a grievance.

(h) Pending the determination of the complaint, the Deputy Minister(s) may take interim measures to separate the employees concerned, if deemed necessary. Any such action taken under this section will not be deemed disciplinary in nature, or seen as presumption of guilt or innocence.

*1.04 Human Rights Code

The Government of British Columbia, in cooperation with the Union, will promote a work environment that is free from discrimination where all employees are treated with respect and dignity.

Discrimination relates to any of the prohibited grounds contained in the B.C. *Human Rights Code*. Prohibited conduct may be verbal, non-verbal, physical, deliberate or unintended, unsolicited or unwelcome, as determined by a reasonable person. It may be one incident or a series of incidents depending on the context.

Employees have the right to employment without discrimination because of race, colour, ancestry, place of origin, religion, family status, marital status, physical disability, mental disability, sex, age, sexual orientation, **gender identity or expression**, political beliefs, and criminal or summary offense unrelated to their employment.

Discrimination does not include actions occasioned through exercising in good faith the Employer's managerial/supervisory rights and responsibilities.

Protection against discrimination extends to incidents occurring at or away from the workplace during or outside working hours provided the acts are committed within the course of the employment relationship.

This Clause does not preclude an employee from filing a complaint under the B.C. *Human Rights Code*, however, an employee shall not be entitled to duplication of process. An employee making a complaint of discrimination must choose to direct a complaint to either the **British Columbia Human Rights Tribunal** or to the process specified in Clause 1.06. In either event a complaint of discrimination, if included as an element of a grievance, shall not be pursued through the process identified in Clause 1.06.

An employee who files a written complaint which would be seen by a reasonable person to be frivolous, vindictive or vexatious may be subject to disciplinary action. Disciplinary action taken may be grieved pursuant to Article 8 - Grievances.

*1.05 Sexual Harassment

The Government of British Columbia, in cooperation with the Union, will promote a work environment that is free from sexual harassment where all employees are treated with respect and dignity.

Sexual harassment is one form of discrimination and is defined as any unwelcome comment or conduct of a sexual nature that may detrimentally affect the work environment or lead to adverse job-related consequences for the victim of the harassment. Prohibited conduct may be verbal, non-verbal, physical, deliberate or unintended, unsolicited or unwelcome, as determined by a reasonable person. It may be one incident or a series of incidents depending on the context.

Sexual harassment does not include actions occasioned through exercising in good faith the Employer's managerial/supervisory rights and responsibilities.

Protection against harassment extends to incidents occurring at or away from the workplace during or outside working hours provided the acts are committed within the course of the employment relationship.

This Clause does not preclude an employee from filing a complaint under the B.C. *Human Rights Code*, however, an employee shall not be entitled to duplication of process. An employee making a complaint of sexual harassment must choose to direct a complaint to either the **British Columbia Human Rights Tribunal** or to the process specified in Clause 1.06. In either event a complaint of sexual harassment, if included as an element of a grievance shall not be pursued through the process identified in Clause 1.06.

An employee who files a written complaint which would be seen by a reasonable person to be frivolous, vindictive or vexatious may be subject to disciplinary action. Disciplinary action taken may be grieved pursuant to Article 8 - Grievances.

Examples of sexual harassment include but are not limited to:

- a person in authority asking an employee for sexual favours in return for being hired or receiving promotions or other employment benefits;
- sexual advances with actual or implied work related consequences;
- unwelcome remarks, questions, jokes or innuendo of a sexual nature; including sexist comments or sexual invitations;

- verbal abuse, intimidation, or threats of a sexual nature;
- leering, staring or making sexual gestures;
- display of pornographic or other sexual materials;
- offensive pictures, graffiti, cartoons or sayings;
- unwanted physical contact such as touching, patting, pinching, hugging;
- physical assault of a sexual nature.

This definition of sexual harassment is not meant to inhibit interactions or relationships based on mutual consent or normal social contact between employees.

*1.06 Discrimination and Sexual Harassment Complaint Procedures

(a) All persons involved in the handling of a discrimination or sexual harassment complaint under Clause 1.04 or 1.05 shall hold in the strictest confidence all information of which they become aware; however, it is recognized that various officials of the constituent group(s) and the Employer will be made aware of all or part of the proceedings on a *"need to know"* basis.

(b) Before proceeding to the formal complaint mechanism an employee who believes they have a complaint of harassment or discrimination may approach their supervisory personnel, union steward, or other contact person to discuss potential means of resolving a complaint and to request assistance in resolving the matter. If the matter is resolved to the complainant's satisfaction the matter is deemed to be resolved.

(c) If the matter is not resolved to the employee's satisfaction, then the employee will approach the first excluded level of management not involved in the matter, for assistance in resolving the issue within six months of the alleged occurrence. The manager will investigate the allegation and take steps to resolve the concern as appropriate within 30 days of the issue being raised by the employee. The manager will discuss the proposed resolution with the employee. The employee may have a union representative present during these discussions. Where the first excluded level of management is the respondent, the employee shall approach the respondent's supervisor.

(d) If the proposed resolution is not acceptable, the employee may refer the matter through the Union in writing to the Deputy Minister or their designate

within 30 days of receiving the manager's response or when the response was due.

A written complaint shall specify the details of the allegation(s) including:

- name, title and Ministry of the respondent;
- a description of the action(s), conduct, events or circumstances involved in the complaint;
- the specific remedy sought to satisfy the complaint;
- date(s) of incidents;
- name(s) of witnesses (if any);
- prior attempts to resolve (if any).

(e) The Deputy Minister or their designate will acknowledge, in writing, receipt of the Union's notice and will have the matter investigated and will take such steps as may be required to resolve the matter. The Union and the employees involved shall be advised in writing of the proposed resolution within 30 days of providing notice to the Deputy Minister or such later date as may be mutually agreed by the Ministry and the Union.

(f) Where the matter is not resolved pursuant to (e), the Union may refer the matter to adjudication in accordance with the Human Resources policy on Discrimination and Harassment in the Workplace.

(g) Any action taken by the Employer, including discipline, which is consistent with the findings of fact of the Adjudicator shall be considered by all parties to be determinative of the complaint and shall not form the basis of a grievance.

(h) If the Adjudicator determines that discrimination and/or harassment has occurred, the Employer must document the personnel file of the respondent accordingly.

(i) Pending the determination of the complaint, the Deputy Minister(s) may take interim measures to separate the employees concerned, if deemed necessary. Any such action taken under this section will not be deemed disciplinary in nature, or seen as presumption of guilt or innocence.

(j) The complainant will not be relocated without their agreement.

1.07 Respectful Workplace

The Employer is committed to promoting a work environment in which employees, students, medical staff, physicians, residents, fellows, volunteers, contractors, visitors, patients and clients conduct themselves in a civil, respectful and cooperative manner.

The Employer will publish a clear policy for promoting and maintaining a working environment in which all persons are treated with respect and dignity. These policies will be accessible to staff and users of the mental health care system regarding expectations and consequences of inappropriate behaviour and violence.

ARTICLE 2 - RECOGNITION AND RIGHTS

2.01 Bargaining Agent

The Employer recognizes the Union as the exclusive bargaining agent for all employees for whom the Union has been certified as bargaining agent.

2.02 Bargaining Unit Defined

The bargaining unit shall comprise all employees included in the Nurses' bargaining unit as defined in this Agreement except those employees in positions mutually agreed to between the parties as managerial and/or confidential exclusions.

The parties to this Agreement acknowledge the difficulty in establishing a service-wide policy for determining managerial and/or confidential exclusions. The parties further agree that cognizance shall be given to the type of hospital, facility, or unit and to the degree to which employees, at varying levels, are involved either in the formation of Government policy or in the process of Employer-employee relations.

The guideline to be considered in negotiating exclusions shall be:

(1) position incumbents employed for the primary purpose of exercising senior management functions;

(2) a sufficient number of position incumbents to represent management in matters relating to labour relations taking into account both operational and geographical considerations.

Incumbents of new positions established by the Employer shall automatically be included in the bargaining unit unless specifically excluded by mutual agreement or by virtue of their being covered by another bargaining unit as specified in Section 4(b) or 4(c) of the *Public Service Labour Relations Act*.

2.03 Scope of Agreement

This Agreement applies to all employees in the bargaining unit as defined in this Agreement but does not apply to the incumbents of those positions listed in Appendix 2 of this Agreement. It is understood that any revision to Appendix 2 will be subject to negotiations between the Employer and the Union.

2.04 No Discrimination for Union Activity

The Employer and the Union agree that there shall be no discrimination, interference, restriction or coercion exercised or practised with respect to any employee for reason of membership or activity in the Union.

2.05 Individual Contracts

The Employer agrees not to enter into any agreement or contract with the employees covered by this Agreement individually or collectively which in any way conflicts with the terms and provisions of this Agreement.

2.06 Contracting Out

The Employer agrees not to contract out any work presently performed by employees covered by this Agreement which would result in the laying off of such employees.

2.07 Notice Boards

The Employer will provide adequate space on bulletin boards located in each unit the sites to be determined by mutual agreement, for the posting of such notices as the Union may from time to time wish to post.

2.08 Accessibility

Union representatives and elected representatives of the Union shall have access to the Government's hospitals, clinics and other facilities by first notifying the designated supervisory official in advance of their intention and their purpose for entering and shall not interfere with the operation of the Ministry or section concerned.

The Union will provide the Employer with names of Union representatives and elected representatives for dealing with Employer representatives. The Employer will provide the Union with names and positions of their designated representatives for dealing with the Union.

2.09 Precedence of Agreement

In the event that there is a conflict between the content of this Agreement and any regulation made by the Employer, or on behalf of the Employer, this Agreement shall take precedence over the said regulation. Written rules and regulations governing the conduct of employees covered by this Agreement shall be sent to the Union.

2.10 Accessibility of Employer's Regulations

All employees covered by this Agreement shall be given access, upon request, to any regulations or circulars which pertain to or affect this Agreement.

2.11 Strikes and Picket Lines

(a) It is mutually agreed that there shall be no strike or lockout, whether sympathetic or otherwise, during the term that this Agreement shall be in force, except as provided in the *Public Service Labour Relations Act* in respect of a dispute arising in the bargaining of a subsidiary collective agreement.

(b) All employees covered by this Agreement shall have the right to refuse to cross a picket line arising out of a dispute as defined in the Labour Relations Code. Failure to cross a picket line shall not be considered a violation of this Agreement nor shall it be grounds for disciplinary action.

(c) Any employee refusing to cross a picket line shall be considered to be absent without pay unless such refusal does not unduly interfere with the proper discharge of their duties, in which case the employee shall not suffer loss of pay.

(d) The Employer agrees not to instruct employees to perform tasks other than they would expect to perform in the normal course of their duties.

(e) In the event of a strike, proposed strike or apprehended action by a Union other than the signatory to this Agreement, the Employer and the Union agree to meet in order to determine the essential and emergent services which will be continued by the employees.

2.12 **Rights of Witnesses**

With respect to disciplinary matters, the Union and the Employer have the right to question witnesses and potential witnesses, and agree to the following:

(a) No employee shall be coerced by either party in respect to any investigation in which they may be a witness or potential witness.

(b) An employee has the right to be accompanied by a representative of the Union when being questioned by the Employer providing that this does not result in an undue delay of action being taken.

(c) Every effort shall be made to provide suitable temporary office accommodation, where required, when interviews take place.

(d) The Employer and the Union are obligated to provide each other, when asked, with the names of those who they may consider, or know, to have witnessed the event or events, which give rise to a particular grievance.

2.13 Ambiguity with Master Agreement

In the event that there is an ambiguity between a component agreement and the Master Agreement, the Master Agreement shall prevail.

*2.14 Amendments

The Deputy Minister of the BC Public Service Agency shall advise the Union of any proposal to amend, repeal, or revise the *Public Service Act*, the *Public Service*

Labour Relations Act, the **Public Sector Pension Plans Act or Pension Benefits Standards Act** which would affect the terms and conditions of employment of employees covered by this Agreement. The Union shall be given notice in writing immediately after the introduction of the above proposals for first reading, of the nature of the proposals so that representation may be made by the Union.

2.15 Invalid Articles

In the event that legislation or a court of competent jurisdiction renders null and void or materially alters any provisions of this Agreement, the remaining provisions of the Agreement shall remain in effect for the term of the Agreement, and the parties hereto shall negotiate a mutually agreeable provision to be substituted for the provisions so rendered null and void or materially altered.

ARTICLE 3 - UNION AND PROFESSIONAL SECURITY

*3.01 Membership

All employees shall as a condition of continued employment become and remain members of the Union.

***3.02** Membership in Professional Bodies

As a condition of continued employment, it is the responsibility of the employee to obtain and maintain membership in those licensing bodies or associations as are necessary to maintain professional standing as a Nurse. Regular full-time employees who have completed their initial probationary period will be entitled to reimbursement of their annual licensing fee to a maximum of \$250, effective April 1, 2018 to a maximum of \$300, prorated for regular part-time employees upon application and presentation of a receipt.

*ARTICLE 4 - CHECK-OFF OF UNION DUES

(a) At the time of initial employment, and upon each transfer to a new facility identified in Appendix 1 the Employer shall determine or confirm the registration

status of every new and transferred employee. Such employees shall be identified as having:

(1) Practicing registration as a registered nurse in British Columbia;

(2) Practicing registration as a registered psychiatric nurse in British Columbia;

(3) Practicing registration as both a registered nurse and a registered psychiatric nurse in British Columbia.

(b) New employees shall sign an authorization of dues deductions form, described in Memorandum of Understanding **#4 authorizing remittance of dues to the Union.** A copy of such authorization will be forwarded to the Union.

(c) The Employer agrees to deduct from the wages of each employee in the bargaining unit, whether or not such employees are members of the Union, the amount of the regular membership dues payable to the Union by a member of the Union.

(d) The Employer shall deduct from each employee who is a member of the Union any assessments levied in accordance with the By-Laws of the Union and owing by the employee to the Union.

(e) Dues shall be remitted to the British Columbia Nurses Union.

(f) Deductions shall be made in each payroll period of each month and membership dues or payment in lieu thereof shall be considered as owing in the pay period for which they were deducted.

(g) All deductions shall be remitted to the Union within 30 calendar days after the date of deduction and the Employer shall provide a list of names of those employees from whose salary deductions have been made. Each list will be divided to indicate employees in the Hospital Services Nurses Component and the Community Services Nurses Component. The list will also indicate additions and deletions to the list, and the amounts deducted from each employee. All lists will be in alphabetical order.

(h) The Employer shall supply each employee without charge a receipt for income tax purposes in the amount of the deductions paid to the Union by the

employee in the previous year. Such receipts shall be provided to the employee prior to March 1 of the succeeding year.

(i) No employee organization other than the Union shall be permitted to have membership dues or other monies deducted by the Employer from the pay of the employee in the bargaining unit.

ARTICLE 5 - UNION TO INTERVIEW NEW OR TERMINATING EMPLOYEES

Where operational requirements permit, the steward and/or the elected representative will be given an opportunity to meet new or terminating employees as a group, within regular hours and without loss of pay, for not more than 30 minutes once each month.

ARTICLE 6 - EMPLOYER'S RIGHTS

6.01 Rights

The Union acknowledges that the management and direction of employees in the bargaining unit is retained by the Employer except as this Agreement otherwise specifies.

6.02 Supervisory Officials

The Employer shall submit the names of the supervisory officials designated to deal with the Union.

ARTICLE 7 - STEWARDS

7.01 Recognition of Stewards

The Employer acknowledges the right of the Union to appoint stewards.

7.02 Area of Responsibility

The Employer and the Union shall by mutual agreement determine the area and jurisdiction of each steward having regard to the plan of organization, the

distribution of employees at the workplace, and the administrative structure implied by the grievance procedure. The jurisdictional areas shall be defined in Appendix 1 of this Agreement. The Union shall notify the Employer in writing of the names of such appointments.

7.03 Duties and Responsibilities

The duties of stewards include the investigation of complaints of an urgent nature, investigation of grievances, and assisting employees in preparing and presenting a grievance in accordance with the grievance procedure, supervision of ballot boxes and other related functions during ratification votes, attending meetings called by the Employer's representative, to accompany employees at meetings of a disciplinary nature, to attend selection panels when nominated by their Union, and to handle other related duties normally accorded to Union stewards.

Stewards shall be entitled to reasonable time without loss of basic pay to perform these duties when:

(a) the steward does not leave their workplace; or when

(b) attending as an observer on a selection panel away from their workplace but within their headquarters' geographic location, and when the selection panel is conducting an interview for a position at the employee's workplace; or when

(c) meeting with representatives of the Employer away from the employee's workplace, at the request of the Employer.

A steward must seek permission from their immediate supervisor before leaving their work station. Such permission will not be unreasonably withheld. On resuming their normal duties, the steward shall notify the supervisor.

A steward who, while off duty, wishes access to their designated area or jurisdiction shall first notify the designated supervisory official. No steward shall be relocated from one unit, sub-unit or jurisdictional area to another as a disciplinary measure because of such legitimate Union activities as are described in this section.

ARTICLE 8 - GRIEVANCES

8.01 **Definitions**

A grievance is:

(a) differences between the parties respecting the interpretation, application, operation, or any alleged violation of a provision of this Agreement, Component Agreement or arbitral award, including a question as to whether or not a matter is subject to arbitration; or

(b) A dispute involving the dismissal, suspension, or discipline of an employee bound by this Agreement.

8.02 Failure to Act

If the Union fails to present a grievance to the next step within the prescribed time limits the grievance shall be deemed to be abandoned.

8.03 Step 1

The Employer and the Union agree that every effort shall be made to settle the dispute at the local level. The aggrieved employee with or without their Union steward shall discuss the matter with their local designated supervisor and shall request a resolution of the matter from the supervisor. If the supervisor fails to resolve the matter to the satisfaction of the employee, or fails to respond to the employee's request for resolution, within the prescribed time, the employee may submit a written grievance, through the Union steward, to Step 2 of the grievance procedure.

8.04 Time Limit

An employee who wishes to present a grievance at Step 2 of the grievance procedure must do so no later than 30 days after the date:

(a) on which they were notified orally or in writing of the action or circumstances giving rise to the grievance; or

(b) on which they first became aware of the action or circumstances giving rise to the grievance.

8.05 Step 2

Subject to the time in Clause 8.04 the employee may present a grievance at this level by:

(a) recording the grievance on the appropriate grievance form setting out the nature of the grievance and the circumstances from which it arose;

(b) stating the Article or Articles of the Agreement infringed upon or alleged to have been violated and the remedy or correction required; and

(c) transmitting the grievance through the Union representative or steward to the representative designated by the Employer with the authority to handle grievances at Step 2 who shall provide the employee with a dated receipt.

8.06 Time Limit

The representative designated by the Employer shall reply in writing to an employee's grievance within 14 days of receiving the grievance at Step 2.

8.07 Step 3

The Union may present a grievance at Step 3:

(a) within 14 days after the decision has been conveyed by the representative designated by the Employer to handle grievances at Step 2;

(b) within 14 days after the Employer's reply was due.

8.08 Time Limit

(a) At this level of the grievance procedure, the grievance shall be submitted to the Employer's Step 3 designate. The Employer's Step 3 designate will discuss the grievance with a representative of the Union with a view to settlement of the grievance. In any event the Employer's Step 3 designate shall reply in writing to the grievance within 30 days of receipt of the grievance.

(b) This step may be waived, by mutual agreement, by the Step 3 designate and such shall be in writing.

8.09 Amending Time Limits

The time limits fixed in this grievance procedure may be altered by mutual consent of the parties, but the same must be in writing.

***8.10** Administrative Provisions

Grievances and replies at Step 3 of the grievance procedure and notification to arbitrate shall be by **email**, registered mail, courier or any other means mutually agreed to by the parties. Where the matter is presented by **email**, registered mail, courier, or other mutually agreed to means, it shall be deemed to be presented on the day on which it is registered or otherwise recorded with the transmitting body and it shall be deemed to be received on the day it was delivered to the appropriate office of the Employer and the Union.

8.11 Dismissal or Suspension Grievance

(a) In the case of a dispute arising out of an employee's suspension with or without intent to dismiss, the grievance shall commence at Step 2 of the grievance procedure. At Step 2 the Employer shall respond to the grievance within 7 days instead of 14 days as indicated in Clause 8.06.

(b) In the case of a dispute arising from an employee's dismissal, the grievance may be filed directly at arbitration in accordance with the provisions of Article 9. Notification of intent to arbitrate and a copy of the grievance shall be submitted to the BC Public Service Agency, Labour Relations Branch with a copy to the Deputy Minister of the appropriate Ministry within 30 days of the date on which the dismissal occurred, or within 30 days of the employee receiving notice of dismissal.

8.12 Group Grievance

Where a grievance as defined in Clause 8.01(a) appears to affect two or more employees and is limited to a single jurisdictional area as listed in Appendix 1 of this Agreement a group grievance may be initiated at Step 1 by one of the affected employees. Group grievances presented in writing at Step 2 shall include a list of all known employees who are party to the grievance.

8.13 General Application Dispute

Where either party disputes the general application, interpretation or alleged violation of an Article of this or a component Agreement, they shall notify the other party in writing within:

(a) 60 days of the date on which the Union or the BC Public Service Agency were notified orally or in writing of the action or circumstances giving rise to the grievance, or

(b) 60 days of the date on which the employees or the local employer first become aware of the action or circumstances giving rise to the grievance;

whichever occurs first.

Notwithstanding the foregoing:

Where either party disputes the general application, interpretation or alleged violation of an ongoing matter they shall notify the other party and the dispute may be designated a "general application dispute" by either party.

The dispute shall be discussed initially with the BC Public Service Agency, or the Union, as the case may be within 30 days of either party receiving notice. Where no satisfactory agreement is reached, either party may submit the dispute to arbitration as set out in Article 9 of this Agreement.

8.14 Investigations and Documents

(a) Union representatives and stewards shall have the right to investigate including questioning witnesses, and shall have access to all documents directly related to the grievance including statements of witnesses, and upon request shall receive copies of all such documents.

(b) Official notes, minutes, and/or transcripts taken at any step of the grievance procedure by either party to the grievance shall be provided to both parties.

8.15 Deviation from Grievance Procedure

The Employer agrees that, after a grievance has been initiated by the Union, the employer's representatives will not initiate any discussion or negotiations, with

respect to the grievance, either directly or indirectly with the aggrieved employee without the consent of the Union.

8.16 Technical Objections

It is the intent of both parties to this Agreement that no grievance shall be defeated merely because of a technical error other than time limitations in processing the grievance through the grievance procedure. To this end, an arbitration board shall have the power to allow all necessary amendments to the grievance and the power to waive formal procedural irregularities.

8.17 Nullification

In the event that after having initiated a grievance through the grievance procedure, an employee endeavours to pursue the same grievance through any other channel, then the Union agrees that pursuant to this Article, the grievance shall be considered to have been abandoned.

8.18 Suspensions Pending Investigation

When the Employer suspends an employee without pay for the purposes of investigating an allegation of any form of misconduct, the Employer shall advise the employee of the expected approximate length of time the Employer may require to carry out such investigation. The Employer shall be prudent and expeditious in its investigation.

ARTICLE 9 - ARBITRATION

9.01 Notification

Where a dispute between the parties arises related to the interpretation, application, or administration of this Agreement or a Component Agreement, including any questions as to whether a matter is arbitrable or where an allegation is made that a term or condition of employment has been violated, either of the parties may, after exhausting the grievance procedure in Article 8 and within 60 days of the receipt of the reply at the third step or within 60 days after the reply was due, advise the other party of its desire to submit the matter to arbitration.

9.02 Composition of the Board of Arbitration

(a) The party requesting arbitration shall, in its notification under Clause 9.01, propose the name of a single arbitrator to act as the Board of Arbitration. The recipient of the notice shall, within 14 days, notify the other party of its agreement to the proposed arbitrator, or propose the name of another arbitrator.

(b) Notwithstanding (a) above, either party may indicate to the other party, within 14 days of receipt of written notice pursuant to Clause 9.01, if it chooses to have the matter heard by a three-person arbitration board. Both parties shall then have 14 days to name their appointee to the three-person board. The two appointees shall then meet to select an impartial chairperson.

9.03 Failure to Appoint

If the parties fail to agree to a single arbitrator under Clause 9.02(a) or if the two appointees fail to select a chairperson within 14 days under Clause 9.02(b), the appointment of a single arbitrator or Chairperson, as the case may be, shall be made by the Director of the Collective Agreement Arbitration Bureau.

9.04 Board Procedure

The Board may determine its own procedure in accordance with the relevant legislation and shall sit, hear the parties, and settle the terms of the question to be arbitrated and make every effort to conclude its award expeditiously.

9.05 Decision of the Board

The Board shall deliver its award in writing to each of the parties and the award of the majority of the Board shall be final and binding on all parties. Where there is no majority, the decision of the Chairman shall be the decision of the Board.

9.06 Disagreement on Meaning of Decision

Should the parties disagree as to the meaning of the Board's decision, either party may apply to the Chairman of the Arbitration Board to reconvene the Board to clarify the decision, which it shall make every effort to do within seven days.

9.07 Expenses of Board

Each party shall pay the costs and expenses of its appointee to the Arbitration Board and share equally the costs and expenses of the Chairman.

9.08 Amending Time Limits

The time limits established for the arbitration procedure in this Article may be extended by mutual consent of both parties.

9.09 **Resubmission to Arbitration**

If the award of the Arbitration Board is subsequently set aside by a court of competent jurisdiction the question shall, at the request of either party, be submitted to another Arbitration Board appointed under the provisions of this Article.

9.10 Expedited Arbitration

(a) The parties may, by mutual agreement, refer to Expedited Arbitration any outstanding exclusion request, or outstanding grievance filed at arbitration, with the exception of group grievances or general application disputes.

(b) An expedited arbitration decision respecting an exclusion request will be deemed to be an agreement between the parties.

(c) The parties shall mutually agree upon a single arbitrator(s) who shall be appointed to hear the grievance and render a decision within two working days of such hearings. No written reasons for the decision shall be provided beyond that which the arbitrator deems appropriate to convey a decision.

(d) An expedited arbitration decision respecting any matter shall be of no precedential value and shall not thereafter be referred to by the parties in respect of any other matter.

(e) All settlements of expedited arbitration cases prior to hearing shall be without prejudice.

(f) Notwithstanding (a) above, either party may remove from the expedited arbitration process any matter at any time prior to hearing and forward the matter through the usual process.

ARTICLE 10 - DISMISSAL, SUSPENSION, AND DISCIPLINE

10.01 Dismissal

A Deputy Minister, or any other person authorized in accordance with the *Public Service Act*, may dismiss an employee for just cause. Notice of dismissal shall be in writing stating the reasons for the dismissal. A copy of the dismissal notice shall be sent to the Union.

10.02 Suspension

A Deputy Minister, or any other person authorized in accordance with the *Public Service Act*, may suspend an employee for just cause. Notice of suspension shall be in writing stating the reasons for the suspension. A copy of the suspension notice shall be sent to the Union.

The Deputy Minister has the right to assign anyone, including a member of a bargaining unit, the authority to suspend. A list of designates shall be provided to the Union.

10.03 Rejection During Probation

(a) A Deputy Minister, or any other person authorized in accordance with the *Public Service Act*, may reject any probationary employee for just cause. A rejection during probation shall not be considered a dismissal for the purpose of Clause 8.11 of this Agreement. The test of just cause for rejection shall be a test of suitability of the probationary employee for continued employment in the position to which they have been appointed, provided that the factors involved in suitability could reasonably be expected to affect work performance.

(b) Where an employee believes they have been aggrieved by the decision of a Deputy Minister or any other person authorized in accordance with the *Public Service Act* to reject the employee during the probationary period, the employee may grieve the decision and the grievance may be filed directly at arbitration in accordance with Article 9.

10.04 Return to Former Classification

(a) When an employee who has been promoted, including a promotion to an excluded position, is rejected by a Deputy Minister or any other person authorized in accordance with the *Public Service Act* during the probationary

period, or where, during the probationary period, the employee decides that the position to which they have been promoted is unsuitable, the Employer will make every effort to return them to a position at their former classification, if one is available, within their geographic location, without loss of seniority. If a position at their former classification, within their geographic location is not available, the Employer shall provide other suitable employee in the first position at their former classification, within their geographic location, when a vacancy occurs.

(b) Where an employee feels they have been aggrieved by a decision of a Deputy Minister or any other person authorized in accordance with the *Public Service Act*, under provision (a) above, they may initiate action through the grievance procedure, at Step 2.

10.05 Unauthorized Absence

Where an employee fails to report for duty for 10 consecutive working days, the employee shall be deemed to have abandoned their position. An employee shall be afforded the opportunity to rebut such presumption and demonstrate that there were reasonable grounds for not informing the Employer.

10.06 Assessment of Professional Competence

Where disciplinary action against an employee is related to professional competence, such matters shall be assessed by supervisory or management employees who are members of the nursing profession.

10.07 Disciplinary Procedure

(a) In all cases where the Employer considers the employee's conduct may warrant disciplinary action (such as suspension with the intent to dismiss, suspension without the intent to dismiss, demotion, reprimand) the Employer shall make every effort to take such action at a meeting with the employee. Where such a meeting is conducted the employee shall be informed in advance of the nature of the complaint(s) being considered. At any meeting with the Employer, which may result in disciplinary action, or where the employee has reason to believe they will face a disciplinary action, the employee shall have the right to have a steward present and to state their side of the case.

(b) Should an employee refuse or fail to represent themselves at a meeting called for the purpose of disciplinary action, or refuse or fail to request to be accompanied by a steward it may be considered by both the Employer and the Union that the employee has abandoned their rights to further representation on the matter by the Union. The refusal or failure of the employee to represent themselves shall not prejudice the Employer from proceeding with disciplinary action.

(c) A steward shall have the right to consult with and/or to have a Union representative present and/or participate in any discussion with supervisory personnel which the steward believes might be the basis of disciplinary action against the steward, providing that this does not result in an undue delay of the action being taken.

10.08 Unjust Dismissal or Suspension

Should an employee be dismissed or suspended, and it is later established that such dismissal or suspension was unjust they shall be returned immediately to their former status in all respects and shall be compensated for any loss of earning they may have suffered by reason of such dismissal or suspension unless otherwise determined by arbitration.

10.09 Disciplinary Records

An employee will receive a copy of all written censures, letters of reprimand, adverse reports and adverse evaluations/performance appraisals. Should an employee dispute any such entry on their file, they shall be entitled to recourse through the grievance procedure and the eventual resolution thereof shall become part of their personnel record. Upon the employee's request, any such document, other than performance appraisals shall be removed from the employee's file after the expiration of 18 months from the date it was issued, provided there has not been a further infraction. The Employer agrees not to introduce as evidence in any hearing any document from the file of an employee, the existence of which the employee was not aware at the time of filing.

10.10 Appraisal Reports

Where a formal appraisal of an employee's performance is carried out, the employee shall be given sufficient opportunity to read and review the appraisal. Provisions shall be made on the appraisal form for an employee to sign it. The form shall

provide for the employee's signature in two places; one indicating that the employee has read and accepts the appraisal, and the other indicating that the employee has read and disagrees with the appraisal. The employee shall sign in only one of the places provided. No employee may initiate a grievance regarding the contents of an appraisal report unless the signature indicates disagreement with the appraisal. An employee shall receive a copy of their appraisal at the time of signing.

Any additions or changes to the appraisal shall be brought to the attention of the employee. Such additions or changes shall be subject to the grievance procedure of this Agreement.

An employee who disagrees with the appraisal and so signifies in the appropriate place, shall have the right to amplify the reasons for their objections in writing, and such amplifications shall be attached to and become part of that appraisal.

ARTICLE 11 - SENIORITY

*11.01 Definitions

(a) <u>Service Seniority</u>

A regular employee's service seniority shall be the length of continuous service in the Public Service of British Columbia. Service seniority for part-time employees shall be pro-rated on the basis of one year's service seniority for every 1827 hours completed.

(b) <u>Classification Seniority</u>

A regular employee's classification seniority means:

(1) The length of service in their present classification from that date upon which an employee is last appointed to their present classification with the status of a regular employee and includes time spent by an employee substituting in or temporarily appointed to a higher classification;

(2) relationship between one classification of employees and another as determined by the salary of the classification.

(c) Notwithstanding the provisions of (b) above, a regular employee who is demoted shall have time previously spent at the level to which they are demoted included in their classification seniority, other than in cases where an employee takes a voluntary demotion in accordance with Clause 12.09 (Transfers Without Posting) or 19.02 (Rehabilitation Committee) or is demoted through no fault of their own. In the latter cases, the employee shall have classification seniority equivalent to all time previously spent at the level to which they are demoted, together with all time spent in any higher classification.

(d) A regular employee of a Health Authority or another health care employer who had been employed as a regular employee in the public service prior to their transfer by the Employer to such an agency, who is appointed to a regular position in the public service shall be credited with service seniority equivalent to their length of service with the agency and the Employer subject to the following conditions:

(1) the employee must successfully complete their probation period on appointment to the Public Service; and

(2) there must be no break in regular service between the two employers.

(e) <u>Recognition of Seniority</u>

When an external applicant is successful for a regular position and the previous Employer was unionized by BCNU the Employer will recognize the employee's previous seniority.

*11.02 Seniority List

(a) On the last date of the payroll period immediately prior to April 1 and October 1 of each calendar year, the Employer shall provide lists showing the seniority of all employees at **their** worksite. The lists **will** be in seniority order.

(b) The Employer will post copies in each worksite.

(c) On the last date of the payroll period immediately prior to April 1 and October 1 of each calendar year, the Employer shall provide the Union a master list showing the seniority of all employees covered by this agreement. The list shall be in seniority order. The seniority lists shall contain the following information:

- (i) name;
- (ii) status (regular full-time, regular part-time, auxiliary);
- (iii) grid level;
- (iv) start date in the Public Service;
- (v) total hours for auxiliary employees;
- (vi) worksite.

(d) Where such lists are produced in electronic format, the Employer will provide them to the Union in this format, provided that it can be done so at no additional cost to the Employer.

11.03 Loss of Seniority

(a) A regular employee on leave of absence without pay, other than a leave of absence for an elected or appointed position in the Union, or leaves granted pursuant to Article 21 (Maternity/Parental/Adoption Leave) shall not accrue seniority for leave periods over 30 calendar days.

(b) A regular employee on a claim recognized by the Workers' Compensation Board shall be credited with seniority equivalent to what they would have earned had they not been absent and had been able to work.

- (c) A regular employee shall lose their seniority in the event that:
 - (1) they are discharged for just cause;
 - (2) they abandon their position;
 - (3) they are on layoff for more than one year;
 - (4) they become an auxiliary employee.

However for the purposes of Articles 12 and 13, with the exception of Article 13.10, an employee who changes their status from regular to auxiliary, or from auxiliary to regular, will maintain their previous seniority providing the break in service has not exceeded 30 days. This provision applies only to employees who change their status after January 1, 2011.

11.04 Re-employment

A regular employee who resigns their position and within 90 days is re-employed as a regular employee shall be granted a leave of absence covering those days absent and shall retain all rights in relation to seniority and other benefits, provided they have not withdrawn their superannuation contributions.

11.05 Bridging of Service

If a regular employee terminates their employment, as a result of a decision to raise a dependent child or dependent children, or to care for a member of the employee's immediate family as defined in Article 20, and is re-employed as a regular employee, upon application they shall be credited with length of service accumulated at time of termination for the purposes of benefits based on service seniority.

The following conditions shall apply:

(1) The employee must have been a regular employee for at least two continuous years at the time of termination.

(2) The resignation must indicate that the reason for termination is to raise a dependent child or dependent children, or to care for a member of the employee's immediate family as defined in Article 20.

(3) The break in service shall be for no longer than six years from the date of resignation.

(4) The previous length of service shall not be reinstated until successful completion of the probationary period on re-employment.

11.06 Training Credits

Employees who are currently credited with service seniority for training course time taken through the old Department of Nursing Education, Mental Health Branch, shall retain such seniority until such time as a break in service occurs.

11.07 Service Seniority Tie Breaker

A regular employee who had previously been an auxiliary employee and who had not lost their auxiliary seniority at the time of becoming a regular employee shall be entitled to use the equivalency of their previously accrued auxiliary seniority as the tie breaker where two or more regular employees have the same regular service seniority.

ARTICLE 12 - POSTINGS, TRANSFERS AND SECONDMENT

*12.01 Postings

(a) Vacancies of a regular nature that are to be filled, for positions in the Bargaining Unit, shall be posted throughout the Public Service or in Regions or Ministries as deemed necessary by the appropriate Deputy Minister. Ministries are not precluded from filling more than one vacancy from each posting.

(b) The notice of postings shall contain the following information:

Nature of position, qualification, skills, whether shift work is involved, wage or salary rate or range and where applicable, specific location. Such qualifications may not be established in an arbitrary or discriminatory manner.

(c) Eligibility lists may be established through the posting process and used to fill vacancies. When eligibility lists are established it shall be stated on the posting. Eligibility lists shall be in effect for a maximum of 12 months from the establishment of the list.

(d) Vacancies of a temporary nature which are known to exceed seven months shall be posted within 30 days. Such postings will normally be limited to the seniority block within each jurisdictional unit as described in Clause 13.03. In those circumstances where the posting is not limited, the provisions of Memorandum of Understanding **#14**, Board and Lodging and Relocation Expenses, will not apply.

(e) Notices shall be posted at least 14 days prior to the closing date of the competition except as provided for in Clauses 12.08 (Relocations), 12.09 (Transfer Without Posting) and 19.02 (Rehabilitation Committee).

(f) On posted competitions, an employee is ineligible for transfer or demotion from one geographic location to another within two years at the previous location. The closing date of the competition shall determine eligibility. A selection panel may waive this restriction with the approval of the applicant's Deputy Minister or designate. This restriction does not apply to supernumerary employees or to promotions. (g) Employees who are absent from their places of employment for periods of longer than 14 calendar days may make a preliminary application for, and in anticipation of, vacancies or new positions which may be posted in their absence.

(h) Temporary vacancies of not more than seven months in duration shall be filled in accordance with Clause 27.08(b) (Temporary Substitution).

(i) Any nurse wishing a lateral transfer should make this known to their respective Personnel Department.

(j) Copies of all nursing postings shall be sent to the Union.

***12.02** Selection Panels

(a) Selection Panels shall be convened in accordance with the *Public Service Act* and regulations/directives pursuant thereto.

(b) The Employer shall give whatever notice is reasonable under the prevailing circumstances to any employee selected to appear before a Selection Panel. Such similar notice will also be provided to the Union **and steward**.

12.03 Interview Expenses

An in-service applicant for a posted position who is not on leave of absence without pay and who has been called for a panel interview shall be granted leave of absence with base pay and shall have their authorized expenses paid. An employee granted leave under this section shall notify their supervisor as soon as they are notified of their requirement to appear for an interview.

12.04 Union Observer

The Union may designate a person to sit as an observer on a selection panel in all selections within the scope of this Agreement. The observer shall be a disinterested party of the panel being convened. The panel chairperson must be given reasonable advance notice that an observer will be attending. This Clause shall not apply to excluded positions.

12.05 Notice of Promotions

The appropriate Deputy Minister or their designate shall notify the Union of any promotions made to positions in which the incumbent would be within the Bargaining Unit.

12.06 Notification of Unsuccessful Applicants

Unsuccessful in-service applicants to posted positions will be notified of the name and classification of the successful in-service applicant, or, upon request, either the name or a summary of the qualifications, skills and experience of the successful outof-service applicant.

*12.07 Review of Staffing Decisions

(a) An employee who is an unsuccessful applicant for an appointment to the public service may request from the individual responsible for the appointment an explanation of the reasons why he or she was not appointed.

(b) The responsible individual must provide an explanation as soon as practicable after receiving a request under subsection (a).

(c) An employee who has made a request under (a) above may request an inquiry into the application of section 8(1) of the *Public Service Act* with respect to the appointment. Any such request must include a detailed statement specifying the grounds on which the request is made and be directed to the deputy minister responsible for the position.

(d) The deputy minister, or a person designated by the deputy minister, who receives an application under (c) above must inquire into the appointment and confirm the appointment or proposed appointment or direct that the appointment or proposed appointment be reconsidered. The Deputy Minister will reply within thirty (30) days.

(e) An employee who is an unsuccessful applicant for an appointment to a position and who has made a request pursuant to (c) above and disagrees with the decision made in (d) above to confirm the appointment or proposed appointment may request a review of the appointment by the merit commissioner on the grounds that section 8(1) of the *Public Service Act* has not been complied with.

(f) A request for a review pursuant to (e) above must be in writing and may only be based upon the grounds submitted to the deputy minister under (c) above.

(g) All requests for reasons, inquiry or review and submissions must be within the time period prescribed by Regulation made pursuant to the *Public Service Act*.

12.08 Relocations

It is understood by the parties that as a general policy employees shall not be required to relocate from one geographic location to another against their will. However, the Employer and the Union recognize that in certain cases relocations may be in the interests of the Public Service and/or the employee. In such cases, an employee will be fully advised of the reasons for their relocation, as well as the possible result of refusal to be relocated.

12.09 Transfers Without Posting

(a) Lateral transfers or voluntary demotions may be granted, without posting, for:

(1) compassionate or medical grounds to regular employees who have completed their probationary periods; or

(2) all employees who have become incapacitated by industrial injury or industrial illness.

In such cases the Rehabilitation Committee outlined in Clause 19.02 shall consider any applications or request presented to the Committee. Each request for special considerations shall be judged solely on its merit.

(b) In addition, lateral transfers or voluntary demotions may also be granted where such transfer is in the best interest of the employee and the Public Service.

(c) An employee whose spouse is also an employee and whose spouse is relocated or transferred pursuant to Articles 12.08 or 13 shall be considered for a lateral transfer or voluntary demotion to available vacancies.

(d) Where an employee whose spouse is also an employee of the Government of the Province of British Columbia but not in this bargaining unit, and where the spouse is relocated or transferred, the Employer may consider a request from the employee for a lateral transfer or voluntary demotion in the context of operational requirements and the interests of the Public Service.

12.10 Demotions and Transfers

Where an employee feels they have been aggrieved by any decision of the Employer relating to a demotion or a transfer, the employee may proceed to Step 2 of the grievance procedure.

12.11 Secondment

(a) Definition - Secondment is a process by which the Employer may assign an employee to another agency, board, society, commission, or employer not subject to the *Public Service Labour Relations Act*.

(b) Notice - The Employer shall make every effort to provide an employee with four weeks' notice of secondment. Where possible, the written notice of secondment shall indicate the term of secondment.

(c) Provisions of Agreement to apply - The provisions of the applicable current Union-Employer collective agreement shall apply to all seconded employees. The agency, board, society, commission, or employer to which the employee is seconded shall receive written notice of this section and shall be provided with copies of the relevant agreements.

(d) Grievance procedure - The grievance procedure prescribed in Article 8 of this Agreement shall apply to seconded employees with the following modifications:

(1) The Employer, in conjunction with the agency, board, society, or commission to which an employee is seconded shall designate an official who shall be considered the locally designated supervisor for purposes of Step 1.

(2) The Employer shall appoint a representative within the Public Service to deal with grievances of seconded employees at Step 2 and shall notify the Union and seconded employees of such designation.

(3) The Union shall appoint a steward for seconded employees and shall notify the Employer of such appointments.

(e) Professional consultations - Employees seconded to agencies, boards, societies, commissions, or employers which do not employ any other employees

in senior nursing or nursing consultative positions shall have access to nursing consultants in the Public Service for advice and consultation on professional nursing matters.

ARTICLE 13 - LAYOFF AND RECALL OF REGULAR EMPLOYEES

13.01 Principles

(a) In the event that the Employer foresees that a layoff may occur the Employer will provide the Union with as much advance notice as is reasonably possible with a view to expedite a staff reduction with the least disruption to the delivery of services and health care and to the work force. The Employer will advise the Union and employees of the number of individuals and classifications likely to be affected by a prospective layoff.

(b) It is understood that all staffing actions described in this Article are subject to the requirements of the *Public Service Act*, however Article 12 of this Agreement may not apply should employees be made offers of placement into vacancies pursuant to this Article.

(c) It is understood that employees to be retained must possess the skills, knowledge and abilities to perform the work which is available, and in the event of a layoff those employees possessing the greatest seniority will have preference to job placement, if a position is available, subject to the provisions as described in this Article.

(d) In the event a position held by a regular employee is relocated outside the originating geographic location, the incumbent shall be offered the position and the opportunity to relocate. An employee may decline an offer pursuant to this section and such a decline will initiate the Article 13 process.

(e) Employees subject to displacement or placement activity pursuant to Article 13 shall be entitled to decline, without penalty, positions with dissimilar hours of work.

(f) For the purposes of Article 13 supernumerary employee means that employee who is identified as being in excess of the number of employees required by the Employer.

13.02 Changing Status

For the purposes of this Article, with the exception of Clause 13.10, an employee who changes their status from regular to auxiliary, or from auxiliary to regular, will maintain their previous seniority providing the break in service has not exceeded 30 days. This applies only to an employee who changes their status after January 1, 2011.

*13.03 Jurisdictional Units and Seniority Blocks

(a) For the purposes of the operation of this Article there shall be jurisdictional units and seniority blocks. Each employee shall be employed within the jurisdictional unit as outlined in their letter of appointment.

(b) If eligible, an employee may only exercise displacement options pursuant to Clause 13.09 within the jurisdictional unit in which they are employed.

(c) Pursuant to the *Public Service Act* requirements an employee may be offered placement into a vacancy into a jurisdictional unit other than the jurisdictional unit in which they are employed.

- (d) There are two jurisdictional units being:
 - (i) Hospital
 - (ii) Community
- (e) <u>Hospital</u>

The seniority blocks of the Hospital Jurisdictional Unit shall be:

- 1 The Maples Adolescent Treatment Centre
- 2 Burnaby Youth Custody Services
- 3 Youth Forensic Psychiatric Services/In-Patient Assessment Unit
- 4 Prince George Youth Custody Services
- (f) <u>Community</u>

The seniority blocks of the Community Jurisdictional Unit shall be by ministry by geographic location.

(g) An employee exercising displacement options within the above seniority blocks shall not be entitled to relocation expenses as per Clause 13.13 for the purposes of this Article. The identified seniority blocks shall be considered headquarters or geographic location as defined in Clause 1.02.

13.04 Redundant Positions

In the event positions become redundant, or in the event that the Employer foresees that a layoff may occur for any reason, the Employer shall identify and advise the Union of those positions the Employer considers redundant, and of vacant positions, which are to be filled by regular employees, elsewhere in the jurisdictional unit in the same and lesser classifications as the redundant positions.

13.05 Pre Layoff

(a) Prior to the identification of supernumerary regular employee(s) or the layoff of regular employee(s) the Employer may within any work area, seniority block, geographic location, or jurisdictional unit canvass any employee or group of employees to invite any one or combination of the following:

(i) a voluntary transfer or a voluntary demotion into a vacant position to be filled by a regular employee; or

- (ii) a voluntary resignation with severance as set out in Clause 13.10; or
- (iii) where eligible, early retirement as set out in Clause 13.09(b)(1).

An employee is under no obligation to participate in the pre layoff canvass.

(b) Should an employee, in writing elect to participate in the pre layoff canvass by accepting the option(s) offered by the Employer pursuant to Clause 13.05(a), such acceptance is final and binding upon the employee, subject to agreement of the Employer. In the event of a placement offer, the employee with the greatest service seniority shall have preference. In the event of a severance offer, the employee with the least service seniority shall receive preference.

(c) The Employer may establish reasonable time periods, not less than five work days, in which responses from employees will be received for consideration.

(d) Where the pending layoffs are a result of a substantial reorganization the Ministry will conduct a pre-layoff canvas pursuant to (a) above.

*13.06 Identification of a Supernumerary Employee

(a) <u>Hospital Jurisdictional Unit</u>

(i) In the event there is a requirement to identify a supernumerary employee as a result of the Employer identifying redundant positions or in the event that the Employer foresees that a layoff will occur for any reason, the supernumerary employee shall be the employee with the least classification seniority within the affected classification within the seniority block.

(ii) Such employee shall be notified in writing of their supernumerary status by the Employer and will be advised of their options as follows:

(1) to be placed in a vacancy to be filled by a regular employee in the same or a lesser classification in the same jurisdictional unit, or;

(2) if no vacancy exists in the supernumerary employee's last previously held classification in their present seniority block, to displace in their present seniority block the employee with the least classification seniority in that classification, provided the supernumerary employee has greater service seniority than the employee identified for displacement.

(3) In the event no displacement can be affected by the supernumerary employee pursuant to (2) above, the provisions of (2) above will be reapplied to each next lower previously held classification(s) if required.

(iii) An employee displaced pursuant to (ii)(2) above shall be advised in writing that they are supernumerary and will be entitled to the provisions of (ii) above.

(iv) Employees identified as supernumerary shall be provided a minimum of five work days to inform the Employer of the option selected pursuant to (ii) above.

(b) <u>Other Jurisdictional Units</u>

(i) In the event there is a requirement to identify a supernumerary employee as a result of the Employer identifying redundant positions or in the event that the Employer foresees that a layoff will occur for any reason, the supernumerary employee shall be the employee with the least service seniority within the affected classification within the seniority block.

(ii) Such employee shall be notified in writing of their supernumerary status and will be advised of their option to be placed in a vacancy to be filled by a regular employee in the same or lesser classification in the same jurisdictional unit. The employee with the greatest service seniority shall have preference in the event of a placement.

(iii) If no vacancy exists in the supernumerary employee's last previously held classification in their present seniority block, to displace in their present seniority block the employee with the least service seniority in that classification, provided the supernumerary employee has greater service seniority than the employee identified for displacement.

(iv) Employees identified as supernumerary shall be provided a minimum of five days to inform the Employer of the option selected pursuant to (ii) above.

(c) <u>Relocation Expenses</u>

For the purposes of Clause 13.06 only, employees who exercise option (a)(ii)(1) or (b)(ii) shall not be entitled to relocation expenses pursuant to Clause 27.16.

13.07 Transfer Within Geographic Location

Should Clauses 13.05 and 13.06 above not result in the placement of a supernumerary employee who has not received notice of layoff, they may be laterally transferred to vacant regular positions within the geographic location. Refusal to accept a lateral transfer shall be deemed a resignation and there shall be no entitlement to severance pay.

13.08 Identification of Employee to Receive Notice of Layoff

The employee to receive a notice of layoff shall be the supernumerary employee who is unable to or elects not to exercise the options in Clause 13.06 above.

13.09 Notice of Layoff and Employee Options Upon Receipt of Notice

(a) <u>Notice of Layoff</u>

(i) The employee with three or more years of service seniority shall be given a minimum of six weeks written notice prior to the effective date of layoff. If the employee has not had the opportunity to work their regularly scheduled shifts during the six week period after notice of layoff, they shall be paid in lieu of work for that part of the regularly scheduled shifts during which work was not made available.

(ii) The employee with less than three years' service seniority shall be given a minimum of 20 work days' written notice prior to the effective date of layoff, or pay in lieu of notice or portion thereof.

(iii) The employee will also be provided a list of options and identification of vacant positions to be filled by regular employees, where applicable.

(b) <u>Employee Options</u>

Within five work days of receipt of notice of layoff an employee shall opt for one of the following:

(1) severance pay effective the date the layoff was scheduled to occur, in which case the employee shall be deemed to have resigned.

A regular employee who is eligible to immediately receive a pension as of the effective date of layoff, and who has opted for and is entitled to severance pay shall, upon application, be entitled to additional pensionable service equivalent in value, as determined by the BC Pension Corporation, to the severance pay compensation. Benefits under this provision shall not exceed the time that would be required to reach the employee's maximum retirement age; (2) to be placed, effective the scheduled date of layoff, on a recall list for a period of up to one year, for the purpose of recall to a regular position within the seniority block from which the employee has been laid off. Recall of employees from this recall list shall be in order of regular service seniority as of the scheduled date of layoff. Seniority accumulated as an auxiliary employee accepting option (b)(3) below shall not apply. Should an employee decline a recall or fail to respond to a recall within five days (or to a maximum of 28 calendar days where an employee is required to provide notice of resignation to another employer), then such employee shall be deemed to have abandoned all rights under the Collective Agreement;

(3) to be placed onto the auxiliary recall list(s) of the seniority block. This option may be elected along with the option described in (b)(2) above. Such regular employees shall be considered more senior than auxiliary employees for the purpose of the assignment of available auxiliary work. Employees electing this option will be considered to have regular status only for purposes of the benefits contained within Articles 19 and 25. For all other contractual purposes, such employees will be considered to have "on-call" auxiliary status pursuant to Clauses 29.01(b) and 29.06(b). In order to maintain the Articles 19 and 25 benefits, employees must work 1200 hours at the straight time rate for each preceding 12 month period;

(4) a regular employee who has been promoted within six months of the effective date of layoff from another position in the same geographic location may opt to displace the employee currently filling the position originally held by the employee designated for layoff, providing the employee exercising such a displacement option has greater service seniority.

(5) an employee with three or more years seniority as a regular employee may opt to be placed into a vacancy to be filled by a regular

	VACANCY / DISPLACEMENT	CLASSIFICATION	SENIORITY BLOCK	GEOGRAPHIC LOCATION	JURISDICTIONAL UNIT
(i)	vacancy	same	different	same	same
	or vacancy	lesser	same	same	same
	or vacancy	lesser	different	same	same
(ii)	displace	lesser	same	same	same
(iii)	displace	same	different	same	same
(iv)	displace	lesser	different	same	same
(v)	vacancy	same	different	different	same
(vi)	vacancy	lesser	different	different	same
(vii)	displace	same	different	different	same
(viii)	displace	lesser	different	different	same
(ix)	vacancy	same	different	same	different
(x)	vacancy	lesser	different	same	different
(xi)	vacancy	same	different	different	different
(xii)	vacancy	lesser	different	different	different

employee or to displace a junior employee with less than three years' service seniority in the following manner in sequence, (i) to (xii) below:

For a period of up to one year, an employee who has been placed at a lesser classification level than that from which they were laid off, shall be recalled to a vacancy, to be filled by a regular employee, in their former classification in their current seniority block.

(6) to accept an available temporary assignment within the geographic location. Upon conclusion of the temporary assignment the employee will be required to select an option pursuant to (b) above.

(c) In order to facilitate the administration of (b)(5) above, an employee is required to advise the Employer within five work days of receipt of notice of layoff if it is their intention to displace. Displacement shall be voluntary and if declined by the employee it shall not count as a placement offer pursuant to (d) below. Should an employee wish to displace, the Employer will identify the employee with the least service seniority within the classification, seniority block or geographic location.

(d) Employees who are offered placement to a vacancy shall be given 10 calendar days to accept or decline such offers. Should an employee decline one

job offer at their same classification level and same geographic location they shall claim severance pursuant to Clause 13.10. Should an employee decline two job offers at their same classification level outside their geographic location and/or at a lesser classification level in their same or outside their geographic location they shall claim severance pursuant to Clause 13.10.

(e) Employees who are to displace shall be given 14 calendar days' notice to report to work. Should an employee fail to report to work pursuant to this Clause they will be deemed to have resigned without further recourse.

(f) Employees who are to be displaced pursuant to the provisions of Clause 13.09 shall receive notice of layoff and shall be entitled to the options contained in Clause 13.09.

(g) An employee shall not accumulate seniority as a regular employee while on layoff.

13.10 Severance Pay

(a) Severance pay for employees with less than three years' service seniority shall be an amount equal to one week's pay for every year of service as a regular employee or major part thereof to a maximum payment equivalent to three week's pay.

(b) Severance pay for employees with three or more years of service seniority shall be three weeks current salary for each year of service or major part thereof.

The employee will not receive an amount greater than 12 months' current salary. If the employee's severance entitlement is the result of voluntary resignation pursuant to Clause 13.05, the maximum amount will be twelve months' current salary.

Severance pay, if converted to time, shall not exceed the time that would be required to reach the employee's maximum retirement age.

13.11 Orientation Period

Employees placed into vacancies, auxiliary positions, recalled to regular positions, or placed into positions as a result of exercising displacement rights will be provided with job orientation.

***13.12** Relocation Expenses

Employees with three or more years' service seniority who relocate pursuant to Clauses 13.09 and 13.01(d) shall be entitled to relocation expenses in accordance with Clause 27.16.

13.13 Joint Committee

(a) A Joint Committee shall be constituted to provide for continuing consultation and cooperation between the parties with respect to the relocation, orientation and placement of employees who have three or more years of seniority and who have grieved the application of Article 13.

- (b) (1) The Joint Committee shall consist of four representatives, two appointed by the Union, two appointed by the Employer, and a Chairperson.
 - (2) The Chairperson shall be appointed jointly by the parties.

(3) The Committee shall meet at the call of either party. Minutes shall be taken of all meetings and copies of such minutes shall be provided to the Employer and the Union.

(c) The Union and the Employer representatives on the Committee shall have the authority to waive by mutual agreement any portion of Article 13 where it is considered by them to be fair and equitable.

(d) Notwithstanding Article 9, the Chairperson of the Committee shall, at the request of either party, sit as an arbitrator over any disputes pertaining to the application, interpretation or alleged violation of Article 13.

13.14 Savings Clause

(a) No provision of the Master or Component Agreements shall act as a bar which may inhibit or prevent the Employer from transferring, placing, or displacing employees pursuant to this Article.

(b) Failure to exercise an option within prescribed time limits will result in a loss of entitlements set out in this Article and an employee will be deemed to have resigned without further recourse.

13.15 Copies of Correspondence

The Union shall be provided copies of notices of supernumerary status, notices of layoff and letters of placement given to employees pursuant to the operation of Article 13.

ARTICLE 14 - HOURS OF WORK

14.01 Hours of Work

The annual hours of work exclusive of meal periods taken away from the work station but including paid holidays will be 1827, which is the equivalent to an average of 35 hours per week. Due to varying lengths of the calendar and work years and the varying times that employees may begin and end their work schedules, an individual employee may not work the exact annual hours, however, all work schedules shall be based on an average of 1827 hours.

A week shall be deemed to be any seven consecutive calendar days.

The daily hours of work for each employee shall be consecutive unless otherwise provided in the Component Agreements.

14.02 Work Day

Application of hours of work are set forth in the Component Agreements.

14.03 Meal Breaks

Meal breaks as set forth in the Component Agreements shall be scheduled during an employee's work day. Should an employee be recalled to duty during a meal break, additional time shall be provided later in the day, or repaid at overtime rates.

14.04 Rest Periods

A rest period is a paid interval which is included in the work day or shift and is intended to give the employee an opportunity to have refreshments and/or a rest:

(a) Two rest periods of 15 minutes each will be granted during each full work day or shift. One rest period of 15 minutes will be granted during a work day of three and one-half to six hours duration.

(b) These rest periods will be taken away from the area of direct patient care, subject to the employee being readily available for recall.

(c) Where an employee is called back during a rest period, or is unable to take their rest period(s) according to the foregoing provisions, the rest period will be granted at another time during the work day or shift.

(d) Subject to local arrangements, employees on a full afternoon or night shift may take their rest periods consecutively.

(e) Employees assigned to a night shift in the Hospital Component, shall have two, 15 minute rest periods in each work period in excess of six hours, one rest period to be granted before and one after the meal period. Employees working a shift of three and one-half hours, but not more than six hours, shall receive one rest period during such shift. Rest periods shall not begin until one hour after the commencement of work or not later than one hour before either the meal period or the end of the shift.

14.05 Consecutive Days Off

Two days off a week will be scheduled consecutively and no employee shall be required to work more than five consecutive days without two days off, unless otherwise provided in the Component Agreements. This Clause shall not apply to a casual auxiliary employee hired expressly to replace employees who are on sick leave or vacation.

14.06 Attendance Security

Regular employees who report for work for which they are scheduled shall receive payment for the full day or shift at the applicable rate, except when the employee:

- (a) leaves work without authorization;
- (b) is sent home for disciplinary reasons.

ARTICLE 15 - SHIFT AND ROTATION

15.01 Shift Designation

The day shift shall be the first shift worked in each day, the afternoon shift shall be the second shift, and the night shift shall be known as the third shift in each calendar day.

15.02 Shift System

(a) Shifts and scheduled days off, shift lengths, shift assignments, rotations and adjustments, changing shifts and days off, volunteering for shifts, rotation between wards, and changes through local arrangements, shall be covered under the respective Component Agreements.

(b) If the Employer rosters an employee in such a manner that the employee is indebted to the Employer, such indebtedness shall not be repayable.

*15.03 Shift Differential

(a) Employees working an afternoon shift or night shift shall receive shift differential for all hours worked on the shift.

(b) Shift differential shall be

\$1.40 per hour.....effective **April 1, 2016**

for the afternoon shift

\$1.50 per hour.....effective April 1, 2016

for the night shift.

(c) Shift differential will apply to overtime hours worked on an employee's regularly scheduled work day. The shift differential which will be paid will be the shift differential for the shift within which the overtime is worked.

(d) Employees covered by Employer-Union approved flex-time or modified work week arrangements, who, by their own volition, choose to begin their shift

at a time which qualifies them for shift differential, shall not be entitled to the differential.

15.04 Hours and Overtime Rates

Subject to the provisions of this Agreement, the application of hours of work and the application of overtime and statutory holiday premium rates will be in accordance with the provisions laid out in the Component Agreements.

15.05 Changes in Schedules

Work schedules for regular employees and casual auxiliary employees shall be posted at least 14 calendar days in advance of the starting day of a new schedule.

(a) In the event that any such employee's schedule of hours of work is changed without 48 hours advance notice, and if such change is the result of another employee utilizing a benefit provided for by the provisions of the Agreement, such as sick leave, bereavement leave, or for reasons attributable to another employee, the employee will receive a premium of 85ϕ per hour for work performed in the first shift to which they changed, in addition to their regular pay.

(b) In the event that any such employee's schedule of hours of work are changed after the hours of work schedule has been posted, and if such change in schedule is the result of indiscriminate rostering by the Employer, the employee shall receive pay at overtime rates for work performed on the shift to which they changed.

ARTICLE 16 - OVERTIME

16.01 Definitions

(a) "Overtime" means work performed by a full-time employee in excess or outside of their regular scheduled hours of work.

- (b) "Straight-time rate" means the hourly rate of remuneration.
- (c) "Time and one-half" means one and one-half times the straight-time rate.
- (d) "Double-time" means twice the straight-time rate.

(e) "Double-time and one-half" means two and one-half times the straight-time rate.

16.02 Authorization and Application of Overtime

- (a) An employee is entitled to overtime compensation when:
 - (1) the overtime worked is authorized in advance by the Employer; and
 - (2) the employee does not control the duration of the overtime worked.

(b) In the case of an emergency or any other legitimate circumstance where it is not possible for an employee to obtain advanced authorization to work overtime, the employee will be entitled to receive overtime.

16.03 Overtime Entitlement

(a) A full-time employee will be entitled to compensation for authorized overtime in excess of scheduled daily, weekly, monthly, annual, or other hours of work as specified in the Component Agreements at the applicable overtime rates.

(b) An employee who works less than the normal full-time hours of work shall be entitled to compensation for authorized overtime for hours worked in excess of the normal full-time daily, weekly, monthly, annual or other hours of work as specified in the Component Agreements at the applicable overtime rates.

For hours worked up to and including the normal full-time daily, weekly, monthly, annual or other hours of work as specified in the Component Agreements the employee shall be paid at the rate of straight time.

(c) Overtime worked shall be compensated at the following rates:

(1) Time and one-half for the first two hours of overtime on a regularly scheduled work day;

(2) Double-time for hours worked in excess of (1);

(3) Double-time for all hours worked on a day of rest. The compensation of overtime in (1) and (2) is to be on a daily basis and not cumulative.

(d) For the purposes of calculating the hourly rate, an employee's bi-weekly rate shall be divided by 70.

(e) Overtime shall be calculated in 30 minute increments. An employee working five minutes or less of overtime shall not receive overtime compensation, unless such is a regular occurrence.

16.04 Recording of Overtime

Employees shall record starting and finishing times, and the circumstances, of overtime worked and sign the appropriate form determined by the Employer.

16.05 Sharing of Overtime

Overtime work shall be allocated on an equitable basis.

16.06 Overtime On Designated Paid Holidays

(a) Overtime on a designated paid holiday shall mean authorized time worked which is in excess of the employee's regularly scheduled hours of work on a designated paid holiday, or time worked on a scheduled day off, where such day off is a designated paid holiday.

(b) An employee who works authorized overtime on a designated holiday shall be compensated at the rate of double time for the overtime hours worked, except for Christmas and New Year's when the compensation shall be at the rate of double time and one-half.

16.07 Overtime On Travel Status

An employee on travel status who is required to travel on Employer's business outside their regular working hours shall be compensated at the applicable overtime rates for all hours travelled. The Employer may determine the means of such travel.

16.08 Overtime Meal Allowance

Meal allowances for those working overtime will be as defined in the Component Agreement.

16.09 No Layoff to Compensate for Overtime

Employees shall not be required to layoff during regular hours to equalize any overtime worked.

16.10 Right to Refuse Overtime

All employees have the right to refuse to work overtime, except when required to do so in emergency situations, without being subject to disciplinary actions for so refusing.

16.11 Call-out Provision

A regular employee who is called back to work outside their regular working hours shall be compensated for a minimum of three hours at overtime rates. They shall be compensated from the time they leave their home to report for duty until the time they arrive back upon proceeding directly to and from work.

16.12 Pre-Arranged Overtime

A regular employee who reports to work for pre-arranged overtime, shall be paid a minimum of three hours at the appropriate overtime rate unless the duration of the work assignment is known in advance by the employee.

16.13 Rest Interval

(a) An employee required to work overtime beyond their regularly scheduled shift shall be entitled to 7 hours and 45 minutes clear between the end of the overtime work and the start of their next regular shift. If 7 hours and 45 minutes clear are not provided, overtime rates shall apply to hours worked on the first regular shift.

(b) In the event that the next regular shift referred to in (a) above falls on a designated holiday listed in Clause 17.01, then in addition to the compensation for the designated holiday, the employee will be paid an additional one half time payment for the first two hours worked on the regular shift; single time payment for all hours worked on the regular shift in excess of the first two hours.

16.14 Transportation

Transportation will be provided to employees who are required to work other than their normal working hours when public or private transportation facilities are not available.

*16.15 Contact While Off-Duty

It is recognized that there may be occasions that demand an employee be contacted during their off-duty time by the Employer or supervisory personnel or by subordinates who report directly to the employee. In such cases, the time so worked will be credited to the hours worked for those working a flexible schedule or recognized as overtime for those employees not working a flexible schedule. In all such cases, it is understood that Clause 16.11 (Call-out) and 16.13 (Rest Interval) of the Master Agreement, Clause **6**.02 of the Community Component Agreement and Clause 5.01 of the Hospital Component Agreement do not apply.

ARTICLE 17 - PAID STATUTORY HOLIDAYS

17.01 Designated Paid Holidays

The following have been designated as paid holidays:

New Year's Day	British Columbia Day
Family Day	Labour Day
Good Friday	Thanksgiving Day
Easter Monday	Remembrance Day
Queen's Birthday	Christmas Day
Canada Day	Boxing Day

Any other holiday proclaimed as a holiday by the Federal, Provincial, or Municipal Government for the locality in which an employee is working shall also be a paid holiday.

17.02 Holidays Falling on Saturday or Sunday

For an employee whose work week is from Monday to Friday and when any of the above-noted holidays fall on a Saturday and is not proclaimed as being observed on some other day, the following Monday shall be deemed to be the holiday for the purpose of this Agreement; and when a holiday falls on a Sunday and it is not proclaimed as being observed on some other day, the following Monday (or Tuesday, where the preceding section already applies to the Monday), shall be deemed to be the holiday for the purposes of this Agreement.

17.03 Holiday Falling on a Day of Rest

When a paid holiday falls on an employee's day of rest, and the employee does not work the holiday, the Employer shall make every reasonable effort to give the employee a lieu day off with pay on the first regularly scheduled work day following the day of rest so affected. Where this is not possible, the lieu day shall be taken at a mutually agreeable time. When a paid holiday is moved to another day under the provisions of this Clause:

(a) work performed by an employee on the day from which the holiday was moved shall be considered as work performed on a day of rest; and

(b) work performed by an employee on the day to which the holiday was moved shall be considered as work performed on a holiday.

17.04 Holiday Falling on a Scheduled Work Day

(a) An employee who works the majority of their shift on a designated holiday which is a scheduled work day shall be compensated at the rate of double-time in addition to their basic salary except for Christmas and New Year's when the compensation shall be at the rate of double-time and one-half in addition to their regular salary.

(b) Employees shall have the option of taking the compensation of (a) above which is in excess of the basic salary in either cash or in time off in accordance with the provision of Article 26.

17.05 Holiday Coinciding with a Day of Vacation

Where an employee is on vacation leave and a day of paid holiday falls within that period, the paid holiday shall not count as a day of vacation.

17.06 Christmas or New Year's Day Off

The Employer agrees to make every reasonable effort to ensure that employees required to work shift shall have at least Christmas Day or the following New Year's Day off.

17.07 Shift Workers

For employees on a shift pattern where statutory holidays are already included in the total number of rest days for the year:

(a) Clauses 17.03, 17.04 and 17.05 do not apply;

(b) An employee who works the majority of their shift on a designated holiday which is a scheduled work day shall be compensated at the rate of single time in addition to their basic salary except for Christmas and New Year's when the compensation shall be at the rate of time and one-half in addition to their regular salary.

Employees shall have the option of taking the compensation in excess of the basic salary in cash or in time off in accordance with the provisions of Article 26.

17.08 Conversion of Hours and Part-Time Entitlement

Statutory holidays and lieu days - Where an employee is granted a designated paid holiday pursuant to Article 17 or a lieu day pursuant to Clauses 17.03 or 17.04, the time off granted will be seven hours per day for the designated paid holiday or for the lieu day for a full-time employee and prorated, based on the following formula, for a part-time employee:

straight time hours worked in the previous 30 calendar days divided by the straight time hours of work of a full-time employee for the same 30 calendar day period.

ARTICLE 18 - ANNUAL VACATION

*18.01 Entitlement

(a) (1) Each employee will earn annual vacation based on their years of continuous service with the Employer. A year of continuous service is defined as service performed from a given date in one month to the immediately preceding date 12 months later.

(2) Vacation Year - For the purposes of this Article a vacation year shall be the calendar year commencing January 1 and ending December 31.

First Vacation Year - The first vacation year is the calendar year in which the employee's first anniversary falls.

Employees on vacation are not subject to recall to duty.

Vacation Year	Total Hourly Entitlement	Daily Equivalent	
i cai	Entitiement	7 hr. shifts	7 1/2 hr. shifts
1	140	20	18.667
2	140	20	18.667
3	140	20	18.667
4	140	20	18.667
5	140	20	18.667
6	147	21	19.600
7	154	22	20.533
8	161	23	21.467
9	168	24	22.400
10	175	25	23.333
11	182	26	24.267
12	189	27	25.200
13	196	28	26.133
14	203	29	27.067
15	210	30	28.000
16	217	31	28.933

(b) Each employee will be entitled to receive an annual vacation as follows:

17	224	32	29.867
18	231	33	30.800
19	238	34	31.733
20	245	35	32.667

Where an employee's regularly scheduled work day is greater than seven hours per day, the annual vacation entitlement shall be converted to hours on the basis of a seven hour day and deducted accordingly.

(c) (1) An employee who commences initial employment and who completes six months' service prior to the completion of the calendar year in which such service commenced would be entitled, subject to the scheduling of vacation, to take any earned vacation period prior to January 31st of the following year and the provisions of Clause 18.01(a)(2) and (b) do not apply.

(2) Should an employee resign while being indebted to the Employer for vacation time taken, but not earned, the Employer shall take repayment from the employee's final cheque(s).

(d) An employee may carry over up to 10 **days'** vacation leave per vacation year except that such vacation carryover shall not exceed 10 **days'** at any time. The Employer may under special circumstances, i.e. extended trip, education, or compassionate grounds, permit an employee to take five extra days of vacation prior to entitlement.

(e) (1) Full-time employees working a partial year shall earn one-twelfth of the annual vacation entitlement for each month in which the employee has received at least 10 days' pay at the straight time rate.

(2) Regular part-time employees shall earn vacation credit on a pro rated basis calculated month by month, and shall be entitled to the same number of calendar days without recall to duty as regular full-time employees, as provided in Clauses 18.01(a) and 18.02.

18.02 Scheduling of Vacation

(a) Employees can take annual vacation during the entire calendar year. Unless stated otherwise in a Component Agreement, the following procedure will apply. Employees may submit their requests for vacations any time prior to October 1st of the current calendar year, the date when the formal notice for vacation requests for the following calendar year is posted. All employee requests for vacation selection shall be completed by November 30th.

All employees must have exercised their seniority rights by November 30th. An employee who does not exercise their seniority rights by November 30th shall not be entitled to exercise their rights in respect to any vacation time previously selected by an employee with less seniority.

The final date for posting the completed vacation schedule shall be December 31st of each calendar year.

(b) (1) An employee who splits their vacation shall not receive the choice of when they wish to take their subsequent portion of their vacation until all other employees in the unit or sub unit have made their choice.

(2) A single vacation period which overlaps the end of a calendar year (December 31) shall be considered as vacation for the vacation year in which the vacation commenced. The portion of vacation taken subsequent to but adjoining December 31 shall not be considered as vacation carry-over, nor as a seniority choice for the subsequent vacation year.

(3) The provisions of sub-clause (2) do not prohibit an employee from adjoining the entitlement for two vacation years, subject to the provisions of Clause (d) herein, however in such case the provisions of sub-clause (2) do not apply to the vacation entitlement for the second vacation year.

(c) Once a vacation schedule has been posted it may only be changed by mutual consent.

(d) Seniority for vacation schedules is to be calculated on the basis of continuous service with the Government of British Columbia and not on the basis of classification seniority. The service seniority principle is to be followed at the sub unit level.

(e) The Employer will make every effort to ensure that vacation periods are granted in conjunction with rest days in order to ensure a maximum number of consecutive days' absence.

(f) Where vacation scheduling arrangements other than the foregoing may be more appropriate at a local level, the Employer and the Union may vary the vacation scheduling procedures.

18.03 Salary During Vacation Period

(a) Payment for vacations will be made at an employee's basic pay except if an employee has been working in a higher-paid position than their regular position for a majority of their regularly scheduled hours in the 60 working days preceding their vacation, in which case they shall receive the higher rate.

(b) When a pay day falls during a regular employee's vacation, the employee shall be entitled to have the pay cheque forwarded to a mailing address supplied by the employee in writing. Once per calendar year upon 30 days' written notice, a regular employee receiving a pay cheque as of November 25, 1997 shall be entitled to receive prior to commencement of a vacation a payroll advance equivalent to the amount of their regular pay cheque issued during the vacation period except that no payroll advance shall be issued in December for any pay periods that fall in January or in March for any pay periods that fall in April.

18.04 Approved Leave of Absence With Pay During Vacation

When an employee is eligible for short-term illness and injury plan benefits, bereavement leave, or leave pursuant to Clause 20.04 (Leave for Court Appearances) or 20.08 (Civil Emergency) during their vacation period, there shall be no deduction from the vacation credits for such leaves. An employee who seeks to have their vacation displaced under this section must provide the Employer with the necessary substantiating documentation.

The period of vacation so displaced shall be taken at a mutually agreeable time. If not taken by March 31st of the following vacation year, it shall be paid in cash, unless carried over and taken in accordance with Clause 18.01(d).

*18.05 Vacation Leave on Retirement

An employee scheduled to retire and to receive **pension benefits under the Public Service Pension Pan Rules** shall be granted full vacation entitlement for the final calendar year of service.

18.06 Separation Allowance

An employee leaving the Government Service without having received their annual vacation for that year shall be allowed pay in lieu of earned vacation at regular rate of pay.

ARTICLE 19 - ILLNESS, INJURY, REHABILITATION AND WORKERS COMPENSATION

19.01 Short-Term Illness and Injury and Long Term Disability

Employees shall be entitled to coverage for short-term illness and injury and longterm disability in accordance with agreed upon regulations as included in Appendix 3.

19.02 Rehabilitation Committee

(a) The Rehabilitation Committee shall review cases submitted to it concerning:

(1) Regular employees who have completed their initial probationary period, who have become incapacitated through temporary disablement or permanent partial disablement, and who as a result are precluded from performing the duties of their own occupation.

(2) All employees who have become incapacitated through industrial injury or industrial illness.

(3) Employees requesting transfers on compassionate or medical grounds.

(b) The Committee shall consist of a BCPSA Occupational Health Program physician as chairperson and two members appointed by the Union and two members appointed by the Employer. Each Ministry's representative at the Committee meetings will be recognized as an Employer's representative to the Rehabilitation Committee for that Ministry.

(c) Following the review of the cases outlined in (a) above, the Committee, taking into account the best interests of the employee and the Employer, shall

make recommendations to the BC Public Service Agency and/or the appropriate Deputy Minister or their designee, in accordance with their agreed terms of reference.

(d) Where the Rehabilitation Committee is unable to decide upon recommendations for a particular case it shall be referred to the Bargaining Principals for final disposition.

19.03 Workers' Compensation

See Appendix 3

ARTICLE 20 - SPECIAL LEAVE

Definition

For the purposes of Article 20 "immediate family" means spouse, parent, child, brother, sister, father in-law, mother in-law including the father and mother of a same sex partner; or a relative who permanently resides in the employee's household or with whom the employee permanently resides.

20.01 Bereavement Leave

In the case of bereavement in the employee's immediate family, an employee, not on leave of absence without pay, shall be entitled to special leave without loss of basic pay, from date of death to and including the day of the funeral, with if necessary, a time allowance for return travelling. Further time may be granted at the Employer's discretion. Such leaves shall normally not exceed five working days. In the event of the death of a relative such as grandparent, son-in-law, daughter-in-law, brother-in-law or sister-in-law the employee shall be entitled to special leave without loss of basic pay of up to one day for the purpose of attending the funeral. Other cases may be considered on their merit. For the purposes of this Clause only, "immediate family" shall include "grandchild", step-child, step-sibling and "stepparent".

20.02 Leave for Professional Association/College Duties

(a) Elected or appointed officials shall be given leave of absence without pay in order to attend short monthly, special membership, or executive meetings of the professional association(s)/college(s). This will be determined by operational requirements, and local management receiving advance notice of not less than 48 hours. The term "short" shall be deemed to apply to periods not exceeding 90 minutes.

(b) Leave of absence without pay shall be granted to employees to attend the annual meeting and conferences of each of the national professional association(s) or their successor organization(s). Time allotment shall be the number of days of the meeting and conference(s) plus reasonable travelling time. The employee will be entitled only to the time they actually require and may be required to substantiate their claim.

(c) Leave of absence without pay shall be granted to employees to attend the annual meeting and conferences of each of the provincial professional association(s)/college(s). Time allotment shall be the number of days of the annual meeting and conference(s) plus reasonable travelling time actually required. They may be required to substantiate claims.

(d) Leave of absence without pay shall be granted to employees for attendance at meetings of which the prime function it is to carry out the duties required by the legislation governing registered nurses and registered psychiatric nurses.

(e) Leave of absence without loss of basic pay shall be granted for attendance at professional meetings established by Ministry or at Employer direction.

(f) Employees who are elected to the board of their professional association/college will be granted time off without pay including reasonable travel time to attend regular or special meetings of their board.

(g) Leave of absence without pay shall be granted for attendance at other professional meetings not exceeding one week.

(h) Consideration of leaves requested under Clause 20.02 [(b) to (e) and (g)] shall be determined by operational requirements and the Employer receiving advance notice of not less than 96 hours.

*20.03 Union Business or Public Duties

(a) Union business shall be considered good cause for leave of absence without pay and on at least five days' notice in writing, to the appropriate local

management, leave of absence without pay shall be given to any designated employee or employees for the purpose of conducting Union business, subject to mutual agreement, if the Employer considers the numbers appointed to be excessive. The Employer may waive any portion of the notice period. The employee shall retain all benefits and security however, such leave will be without salary.

(b) Employees who are members of the Union's Negotiating Committee shall be granted leave of absence without loss of basic pay, including travelling time, in order to conduct negotiations. The provisions of Article 16 - Overtime, do not apply.

(c) Employees who are members of the Union's Negotiating Committee will be granted leave of absence without pay to attend preparatory negotiating meetings.

(d) Employees attending third party hearings at the request of the Union will be granted leave of absence without pay.

(e) The Employer shall grant, on written request, leave of absence without pay:

(1) for employees to seek election in the Municipal, Provincial, or Federal or Aboriginal Community Government Election;

(2) for employees elected for a full-time position with the Union, or any body with which the Union is affiliated, for a maximum period of five years;

(3) for employees elected to a public office for a maximum period of five years.

(f) To facilitate the administration of Clauses (a), (c), (d) when leave of absence without pay is granted, it shall be with pay, and the Union agrees to reimburse the Employer for the appropriate salary and benefit costs including travel time.

20.04 Leave for Court Appearances

(a) The Employer shall grant leave without loss of basic pay to employees other than employees on leave without pay to serve as jurors or witnesses in a court action, provided such court action is not occasioned by the employee's private affairs.

(b) In cases where an employee's private affairs have occasioned a court appearance such leave to attend at court shall be without pay.

(c) An employee in receipt of their regular earnings while serving at court will remit to the Employer all monies paid to them by the court, except travelling and meal allowances not reimbursed by the Employer.

(d) Time spent at court by an employee in their official capacity shall be at their regular rate of pay.

(e) Court actions arising from employment, requiring attendance at court, shall be with pay.

(f) In the event an accused employee is jailed pending a court appearance, such leave of absence shall be without pay.

20.05 Reduction of Benefits

Employees on leave of absence with pay shall retain all benefits in their absence including, but not limited to, increment increases, educational bonus payment, paid vacation, holiday pay, all health and welfare benefits, classification and seniority, but not including shift differential, maximum security allowance, on-call payment, and/or overtime. Isolation allowance, special living allowance, and subsidized housing shall continue to apply providing the employee's home location remains unchanged. On return to active employment, leave time shall be counted as service time with the Employer.

20.06 Leave for Committee Meetings

Employees who are representatives of the Union serving on Standing Joint Committees, their subcommittees, or other Employer-Union committees called for in this or the Component Agreements, shall be granted leave without loss of basic pay when attending such joint committee meetings. The Employer shall not bear the cost of expenses incurred by the Union's representatives.

20.07 Her Majesty's Forces

(a) Where an employee is required to take annual training with Her Majesty's reserve forces, special leave shall be granted without pay; provided, however, that where it is certified that the pay and allowances granted by Her Majesty's reserve forces are less than the remuneration ordinarily received from the Employer, the employee shall be paid the difference between the reserve forces pay and their Employer remuneration.

(b) Where an employee takes the annual reserve forces training during their annual vacation leave they shall, during such annual leave, be paid their full remuneration from the Employer in addition to any pay and allowances received from the reserve forces.

(c) Where an employee makes application to attend, as a delegate, meetings of service associations related to Her Majesty's Forces or the conference of the defence associations, special leave may be granted without pay; provided, however, that where it is certified that any pay and allowances received whilst attending the aforementioned meetings are less than the remuneration ordinarily received from the Employer, the employee may be paid the difference between such pay and allowances and their Employer remuneration.

(d) Where an employee makes application to take a prescribed course of training for the purpose of qualifying for a higher rank in the reserve forces special leave for the purpose may be granted; such special leave to be without pay.

(e) If an employee who was granted leave of absence for the purpose of enlisting with the forces of the Crown was discharged from the forces and immediately came under the jurisdiction of the Department of Veterans' Affairs, their leave of absence continues until their discharge from the care of the Department of Veterans' Affairs.

(f) Employees shall be granted leave of absence from the Employer in order to serve with Her Majesty's forces or with an allied force on active duty and shall on their return to the employ of the Employer return to their former classification and shall have all time served with the Armed Forces credited as if it were time served with the Employer.

20.08 Civil Emergency

Employees summoned to control a forest fire, flood, or other civil emergency will be retained on the payroll of the Employer and will receive the regular rate of pay for their working day position for a normal working day. If an employee is required to control a civil emergency beyond their normal working day they may be paid for the extra hours by the authority involved, such as the Forest Service, at regular rates. If an employee is on vacation with pay at the time of the summons to control a forest fire, flood, or other civil emergency, the number of days fighting such an emergency shall not be credited to their vacation leave provided that any remuneration other than overtime is turned over to the Employer and their vacation days shall be rescheduled to a mutually convenient time.

20.09 Elections

For the purpose of voting in a Federal, Provincial, Municipal or Aboriginal Community Government Election, whether such election be Province-wide, a by-election, or a referendum, employees eligible to vote shall be granted necessary time off without loss of pay of three or four consecutive clear hours, as prescribed by the applicable statute, during the hours in which the polls are open in which to cast their ballot. Employees who have a period of three or four consecutive hours, as applicable, off either prior to reporting to duty or after leaving duty shall not be entitled to such leave. Employees with less than three or four clear hours outside of working hours in which to vote shall be allowed the necessary time off duty so that three or four clear hours are available.

20.10 General Leave

Notwithstanding any provision for leave in this Agreement, the Employer may grant leave of absence without pay to an employee requesting such leave for emergency or unusual circumstances or for other valid reasons. Such requests are to be submitted in writing for approval by the Employer. Approval shall not be unjustly delayed or withheld. If leave is refused, reasons for the refusal shall be given to the employee in writing.

20.11 Household Emergency

Where there is a serious household or domestic emergency an employee shall be granted necessary time off without loss of basic pay, in order to make suitable arrangements. Such leave will be limited to a maximum of one day at any one

time, and on not more than three occasions per calendar year. The Employer may grant additional leave with or without pay, as the occasion merits. The employee may be required to provide supporting evidence under such circumstances.

20.12 Family Illness

An employee is entitled, after notifying their supervisor, to a maximum of two consecutive days at any one time to care for an ill member of the immediate family. The Employer reserves the right at any time to call for a report by a physician. For the purpose of this clause, "*child*" includes a child over the age of 18 residing in the employee's household who is permanently dependent on the employee due to mental or physical impairment.

20.13 Leave for Medical and Dental Care

(a) Where it is not possible to schedule medical and/or dental appointments outside regularly scheduled working hours, reasonable time off for medical and dental appointments for employees and for employees with dependent children (in need of medical and/or dental care) shall be permitted, but where any such absence exceeds two hours, the full-time absence shall be charged to the entitlement described in Clause 20.15. "Medical and/or dental appointment" include only those services covered by the B.C. Medical Services Plan, the Public Service Dental Plan, the Extended Health Benefit Plan and assessment appointments with the Employee and Family Assistance Program.

(b) Employees in areas where adequate medical and dental facilities are not available shall be allowed to deduct from their credit described in Clause 20.15 the necessary return travelling time to receive personal or immediate family medical and dental care at the nearest medical centre. The Employer may request a certificate of a qualified medical or dental practitioner, as the case may be, stating that treatment could not be provided by facilities or services available at the employee's place of residence. An employee on leave provided by this clause shall be entitled to reimbursement of reasonable receipted expenses for accommodation and travel to a maximum of \$500 per calendar year.

(c) An employee otherwise entitled to leave pursuant to (b) above who chooses to travel on a vacation day or a day of rest or to remain at work and not accompany their spouse, dependent child or dependent parent, as provided in (b) above, may claim the reimbursement of receipted expenses under the conditions stipulated.

(d) Employees in receipt of STIIP benefits who would otherwise qualify for leave under this clause shall be eligible to claim expenses in the manner described above.

(e) Where leave pursuant to (b) above would be reduced, the Employer may approve airfare payment for the employee in lieu of the \$500 reimbursement, once per calendar year.

(f) For the purposes of this clause, "*child*" includes a child over the age of 18 residing in the employee's household who is permanently dependent on the employee due to mental or physical impairment.

*20.14 Special Leave

(a) An employee not on leave of absence without pay shall be entitled to special leave at their regular rate of pay for the following:

- (1) Attend wedding of employee's child one day
- (2) Birth or adoption of the employee's child two days
- (3) Moving household furniture and effects one day
- (4) Attend funeral as pall-bearer or mourner maximum one-half day
- (5) Attend their formal hearing to become a Canadian citizen one day
- (6) Marriage of the employee three days

(7) In the case of serious illness or hospitalization of a parent or stepparent of the employee, when no one other than the employee can provide for the needs of the parent or step-parent, and, after notifying their supervisor - **two days per calendar year effective September 1, 2017** this may be used in one-half shift increments

- (8) Court appearance for hearing of employee's child one day
- (9) Child custody hearing one day per calendar year
- (b) Two weeks' notice is required for leave under (a)(1), (3), (5) and (6).

(c) For the purpose of (a)(1), (3), (4), (5), (7), (8) and (9) leave with pay will be only for the work day on which the situation occurs.

(d) For the purpose of determining eligibility for special leave under (3) an employee will qualify if they are maintaining a self-contained household and if they are changing their place of residence which necessitates the moving of

household furniture and effects during their normal working hours, and if they have not already qualified for special leave under (3) on two occasions within the preceding 12 months.

20.15 Special Leave Limitation

(a) For leave provided in Clause 20.11 (Household Emergency), 20.12 (Family Illness), 20.13 (Leave for Medical and Dental Care) and 20.14 (Special Leave) the maximum length specified for each circumstance shall not be exceeded; however, a leave may be granted more than once for the same circumstance within a calendar year provided that the total of such leaves do not exceed 70 hours per calendar year, unless additional special leave is approved by the Employer.

(b) Regular part-time employees shall be entitled to leave with pay pursuant to Clauses 20.11, 20.12, 20.13, and 20.14 providing the total days on which leave is required does not exceed 10 working days per year. Pay for such day(s) where leave is required shall be on a pro-rata basis.

Examples:

(i) An employee on a 2:4 shift pattern where the normal shift pattern is 4:2 would be entitled to pay for each working day of special leave required up to 35 hours per year.

(ii) An employee working four hours per day in a unit whose normal shift is seven hours per day would be entitled to 4/7 (.571) of a day's pay for each working day of special leave required up to 40 hours per year.

(iii) An employee who is scheduled to work 10 seven and one-half hour days per month would be entitled to pay for each working day of special leave required up to $32.18 (10 \times 70/21.75)$ hours per year.

20.16 Service Breaks

Approved absences without pay shall not be considered breaks in service. This does not mean an employee will accumulate or accrue benefits as though they were working, but that they will maintain existing benefits already accrued. Health and welfare benefits shall continue during such absences and the employee shall pay the full Employer and employee portions for each full calendar month(s) that the employee receives no salary payment.

20.17 Donor Leave

An employee shall be granted the necessary leave of absence with pay for the purpose of donating bone marrow or an organ.

20.18 Extended Child Care Leave

Upon completion of maternity, adoption and/or parental leave, including any extension to such leaves, a regular employee will be entitled, upon written application, to a leave of absence without pay to care for the child. Subject to Clause 11.03(a), the following conditions shall apply:

(a) The employee's application shall be submitted to the Employer at least four weeks prior to the expiration of Article 21 (Maternity/Parental/Adoption) leave.

(b) The combined length of leaves under this Clause and under Article 21 shall not exceed 18 months.

(c) The employee's return to work requirements of Clauses 21.08(b) and 21.11 shall be deferred until the expiration of this leave. Notification of return to work and return to work shall be subject to Clause 21.09.

(d) Upon return to work from this leave, the employee shall be placed in their former classification.

20.19 Compassionate Care Leave

An employee who is entitled to compassionate care benefits under the *Employment Insurance Act* is entitled to a leave of absence without pay of up to eight weeks for the purpose of providing care or support to a gravely ill family member at risk of dying within 26 weeks. Notwithstanding Clause 11.03(a), there will be no interruption in the accrual of seniority or eligibility for benefits provided for under Article 25.

ARTICLE 21 - MATERNITY/PARENTAL/ADOPTION LEAVE

21.01 Maternity Leave

(a) An employee is entitled to maternity leave of up to 17 weeks without pay.

(b) An employee shall notify the Employer in writing of the expected date of the termination of her pregnancy. Such notice will be given at least 10 weeks prior to the expected date of the termination of the pregnancy.

(c) The period of maternity leave alone or in combination with the leave period of 21.03 shall commence six weeks prior to the expected date of the termination of the pregnancy. The commencement of leave may be deferred for any period approved in writing by a duly qualified medical practitioner or registered midwife.

21.02 Parental Leave

(a) Upon written request an employee shall be entitled to parental leave of up to 37 consecutive weeks without pay.

(b) Where both parents are employees of the Employer, they shall each qualify for up to 37 weeks of parental leave.

(c) Such written request pursuant to (a) above must be made at least four weeks prior to the proposed leave commencement date.

(d) Leave taken under this clause shall commence:

(1) in the case of a mother, immediately following the conclusion of leave taken pursuant to Clause 21.01 or 21.03;

(2) in the case of the other parent, immediately following the birth or placement of the adoptive child.

(3) The commencement of the leave taken pursuant to (1) or (2) above may be deferred by mutual agreement, however, the leave must begin within the 52-week period after the date of birth or placement of the adoptive child. Such agreement shall not be unreasonably withheld.

Such leave request must be supported by appropriate documentation.

21.03 Maximum Combined Entitlement

An employee's combined entitlement to leave pursuant to 21.01 and 21.02 is limited to 52 weeks.

21.04 Benefit Waiting Period Allowance

An employee who qualifies for and takes leave pursuant to 21.01 or 21.02 and is required by Employment Insurance to serve a two week waiting period for Employment Insurance Maternity/Parental benefits, shall be paid a leave allowance equivalent to two weeks at 85% of the employee's basic pay.

21.05 Maternity Leave Allowance

(a) An employee who qualifies for maternity leave pursuant to Clause 21.01, shall be paid a maternity leave allowance in accordance with the Supplemental Employment Benefit (SEB) Plan. In order to receive this allowance, the employee must provide to the Employer, proof that she has applied for and is eligible to receive employment insurance benefits pursuant to the *Employment Insurance Act*. An employee disentitled or disqualified from receiving employment insurance benefits is not eligible for maternity leave allowance.

(b) Pursuant to the Supplemental Employment Benefit (SEB) Plan, the maternity leave allowance will consist of 15 weekly payments equivalent to the difference between the employment insurance gross benefits and any other earnings received by the employee and 85% of the employee's basic pay.

21.06 Parental Leave Allowance

(a) An employee who qualifies for parental leave pursuant to Clause 21.02, shall be paid a parental leave allowance in accordance with the Supplemental Employment Benefit (SEB) Plan. In order to receive this allowance, the employee must provide to the Employer proof of application and eligibility to receive employment insurance benefits pursuant to the *Employment Insurance Act*. An employee disentitled or disqualified from receiving employment insurance benefits is not eligible for parental leave allowance.

(b) Pursuant to the Supplemental Employment Benefit (SEB) Plan, the parental leave allowance will consist of a maximum of 35 weekly payments, equivalent to the difference between the employment insurance gross benefits and any other earnings received by the employee and 75% of the employee's basic pay. Where both parents are employees of the Employer, the employees shall determine the apportionment of the 35 weeks parental leave allowance between them.

21.07 **Pre-Placement Adoption Leave**

Upon request and with appropriate documentation, an employee is entitled to pre-adoption leave without pay of up to seven weeks (245 work hours) per calendar year with an allowance of 85% of their basic pay during the leave period.

The leave may be taken intermittently and only for the purpose of:

(1) attending mandatory pre-placement visits with the prospective adoptive child;

(2) to complete the legal process required by the child's or children's country, including travel, for **an** international adoption while the employee is in that country.

Leave under this provision will end with the placement of the adoptive child(ren).

Pre-placement visits are not normally required where the adoption is a direct placement. Examples of direct placement adoptions are:

- (1) adoptions by a family member;
- (2) adoptions by the partner of a birth parent; and

(3) adoptions by foster parents if the child or children were living with the foster parents immediately before the adoption process.

21.08 Benefits Continuation

(a) For leaves taken pursuant to Clauses 21.01, 21.02, and 21.07, the Employer shall maintain coverage for medical, extended health, dental, group life and long term disability, and shall pay the Employer's share of these premiums.

(b) Notwithstanding (a) above, should an employee be deemed to have resigned in accordance with Clause 21.09 or fail to remain in the employ of the Employer for at least six months or a period equivalent to the leave taken at (a)

above, whichever is longer, after their return to work, the Employer will recover monies paid pursuant to this clause, on a pro-rata basis.

21.09 Deemed Resignation

An employee shall be deemed to have resigned on the date upon which leave pursuant to Clauses 21.01, 21.02, or 21.07 commenced unless they advised the Employer of their intent to return to work one month prior to the expiration of the leave taken pursuant to Article 21 – Maternity, Parental and Pre-Adoption Leave or Clause 20.18 or if they do not return to work after having given such advice.

21.10 Entitlements Upon Return to Work

(a) An employee who returns to work after the expiration of maternity, parental or pre-adoption leaves shall retain the seniority the employee had accumulated prior to commencing the leave and shall be credited with seniority for the period of time covered by the leave.

(b) On return from maternity, parental, or pre-adoption leaves, an employee shall be placed in the employee's former position or in a position within their classification without loss of benefits.

(c) Notwithstanding Clauses 18.01(e) and 18.01(d) vacation entitlements and vacation pay shall continue to accrue while an employee is on leave pursuant to Clause 21.01 providing:

(1) the employee returns to work for a period of not less than six months; and

(2) the employee has not received parental allowance pursuant to 21.06; and

(3) the employee was employed prior to March 28, 2001.

Notwithstanding Clause 18.01(d) vacation earned pursuant to this Clause may be carried over to the following year or be paid out, at the employee's option.

(d) Employees who are unable to complete the return to work period in (c) as a result of proceeding on maternity, parental or pre-adoption leave shall be credited with their earned vacation entitlements and vacation pay providing the employee returns to work for a period of not less than six months following the expiration of the subsequent maternity, parental or pre-adoption leave.

21.11 Maternity and/or Parental Leave and/or Pre-Adoption Leave Allowance Repayment

(a) To be entitled to the maternity, parental, benefit waiting period and/or pre-adoption leave allowances pursuant to Clauses 21.04, 21.05, 21.06 and/or 21.07, an employee must sign an agreement that they will return to work and remain in the Employer's employ for a period of at least six months or equivalent to the leaves taken, whichever is longer, after their return to work.

(b) Should the employee fail to return to work and remain in the employ of the Employer for the return to work period in (a) above, the employee shall reimburse the Employer for the maternity, parental, benefit waiting period and/or pre-adoption leave allowance received under Clauses 21.04, 21.05, 21.06, and/or 21.07 above on a pro-rata basis.

21.12 Benefits Upon Layoff

Regular employees who have completed three months of service and are receiving an allowance pursuant to Clauses 21.04, 21.05 and/or 21.06 shall continue to receive that allowance upon layoff, until the allowance has been exhausted, provided the notice of layoff is given after the commencement of the leave.

ARTICLE 22 - OCCUPATIONAL HEALTH AND SAFETY

***22.01** Copies of Regulations

The Employer agrees to ensure that **the** current *Workers' Compensation Act* **and Occupational Health and Safety Regulation (OHSR)** are available in each workplace and readily accessible to all employees.

*22.02 Safe Workplace

Consistent with the provisions of the Occupational Health and Safety Regulations:

(a) No employee shall be required or expected to lift any patient without assistance, if in the opinion of the employee, the weight of the patient is excessive.

(b) The Parties will instruct their representatives on Local **Joint** Occupational Health and Safety Committees to review the matter of employee safety while travelling to or from their workplace. The Committees will make recommendations regarding the establishment of policies and/or procedures to eliminate or minimize such risk to employees. Where elimination of such risk is not reasonably possible, the Committees shall make recommendations to either manage or avoid the risk.

*22.03 Right to Refuse Unsafe Work

(a) No employee shall carry out or cause to be carried out any work process or operate or cause to be operated any tool, appliance or equipment when that employee has reasonable cause to believe that to do so would create an undue hazard to the health or safety of any person.

(b) No employee shall be subject to disciplinary action because they have acted in compliance with (a) above and 3.12 of the OH&S regulation.

*22.04 Communicable Disease

(a) It is agreed that:

(1) there be active promotion for communicable disease control information, in particular Hepatitis B prevention and protocol, in individual workplaces.

(2) specifically, written information in addition to formal presentations utilizing guest speakers and audio visual materials, as appropriate should be provided.

(3) Local Occupational Health and Safety Committees should coordinate these activities. Materials should be obtained directly from **Immunization Services at the BC Centre for Disease Control.**

(b) Voluntary Hepatitis B immunization shall be provided by the Employer free of charge to the employees.

*22.05 Occupational Health and Safety Committees

(a) The Employer and the Union agree to establish Occupational Health and Safety Committees (OH&S) at all facilities. The composition will be determined locally through management and stewards. When such committees are formed,

they may encompass the employees of more than one bargaining unit. These committees will meet at least monthly, to make recommendations on unsafe, hazardous, or dangerous conditions with the aim of preventing and reducing risk of occupational injury and illness. A copy of all minutes of the OH&S Committees shall be sent to the Provincial Joint Occupational Health and Safety Committee as well as to the Union and the Employer.

At any worksite where a committee has not been established pursuant to the above, a less formal program shall be maintained in accordance with *Workers' Compensation Act*, Part 3, Division 4. For the purpose of assisting in the administration of this program, the Employer will recognize an employee at that worksite designated by the Union who will function as a safety representative of the employees. Records of the meetings and matters discussed shall be forwarded to the Union and the nearest local committee established in the above within the ministry administrative management area.

(b) Employees who are representatives of the Committee shall be entitled to attend meetings of the Committee and perform job site inspections and incident investigations in accordance with *Workers' Compensation Act* and **Occupational Health and Safety Regulation (OHSR)**, and shall not suffer any loss of basic pay for the time spent.

(c) In areas where worksite inspections involve considerable travel, each worksite shall submit regular safety reports to the Committee and inspections shall be carried out when feasible or where considered necessary by a majority of the Committee.

(d) Committee business and meetings shall be scheduled during normal working hours whenever practicable. Time spent by employees attending Committee meetings or business on their days of rest or outside their regularly scheduled hours of work shall not be considered time worked but such employees shall receive equivalent time off at straight time.

22.06 Disagreement Over Recommendations

When a disagreement over unsafe working conditions is referred to the OH&S Committee, the Committee shall investigate the matter of disagreement and make its recommendations to the Employer and the Union.

22.07 Injury Pay Provisions

An employee who is injured on the job during working hours, or during a meal break while on or adjacent to the Employer's premises, and is required to leave for treatment or is sent home for such injury, shall receive payment for remainder of the shift.

22.08 Transportation of Accident Victims

Transportation to the nearest physician or hospital for employees requiring medical care as a result of an on-the-job accident shall be at the expense of the Employer.

*22.09 Provincial Joint Occupational Health and Safety Committee (PJOSH)

There shall be established a **Provincial** Joint Committee composed of four representatives of the Employer (at least one from the BC Public Service Agency) and four representatives of the Union. Each party may have support staff present at the table for consultation if necessary. Employees shall be on leave of absence without loss of basic pay for time spent on this committee. The Committee's responsibilities will be:

(a) To review reports on matters referred by Occupational Health and Safety Committees or by Ministry Joint Committees and make recommendations to the bargaining principals regarding occupational health and safety matters;

(b) To monitor and assess results of the Training Program for Occupational Health and Safety Committee members; and

(c) To jointly develop a new or approve an existing training package on risk assessment of violence in the workplace.

(d) To review and recommend Violence Prevention Programs in accordance with the terms of Appendix 6 – Addressing Workplace Violence.

The Parties shall identify committee members and have a meeting date set no later than 60 days after ratification of this contract and meet thereafter quarterly. The parties may jointly determine to meet more or less frequently.

Minutes of the Committee meetings shall be kept and a copy of all minutes of the PJOSH shall be sent to the local OH&S Committees as well as the Union and the Employer.

22.10 Strain Injury Prevention

(a) The Parties agree that there is a shared interest in minimizing and/or eliminating musculo-skeletal strain injuries or illness which are work related.

(b) Local Occupational Health and Safety Committees shall, in the performance of regular worksite inspections, identify the following risk factors which may contribute to risk:

- (i) the work methods and practices
- (ii) the layout and condition of the workplace and workstation
- (iii) the characteristics of objects or equipment handled
- (iv) the environmental conditions
- (v) the physical demands of work

in a manner consistent with generic guidelines developed by the Provincial Joint Occupational Health and Safety Committee.

(c) Where new equipment will be introduced to the workplace, or during the design and planning stages of new or renovated workplace or workstations, the Employer shall seek the appropriate advice with respect to the risk factors noted in (b). Such advice will be sought from resources which will, where appropriate, include a joint occupational health and safety committee or designated safety representatives.

***22.11** Occupational First Aid Requirements and Courses

(a) The Union and the Employer agree that First Aid Regulations made pursuant to the *Workers' Compensation Act* shall be fully complied with.

(b) Where the Employer requires an employee to perform first aid duties in addition to the normal requirements of the job, the cost of obtaining and renewing the Occupational First Aid Certificate shall be borne by the Employer, and leave to take the necessary courses shall be granted with pay.

(c) Employees required to possess an Occupational First Aid Certificate and who are designated to act as the First Aid Attendant in addition to their normal job responsibilities shall receive the following allowance on the basis of the Level of certificate which they hold:

Level 3 Occupational First Aid Certificate

Per biweekly period

\$56..... effective April 1, 2016 \$58 effective April 1, 2018

Level 2 Occupational First Aid Certificate

Per biweekly period

\$44..... effective April 1, 2016 \$45..... effective April 1, 2018

The allowance shall be prorated for partial months. For the purpose of calculating the hourly rate, the bi-weekly allowance shall be divided by 70; however, no employee shall receive more than the monthly allowance for the Level of certificate which they hold.

Employees designated to act as the Occupational First Aid Attendant in addition to their normal job duties will receive their full monthly allowance while on approved leave with pay of up to 10 days or while on vacation leave with pay.

Where the Employer has an additional requirement for a First Aid Attendant on a temporary basis, then provided the employee acts as the First Aid Attendant for a minimum of 10 work days in any month, they shall receive the full monthly allowance.

(d) (1) In order to meet the requirements of (a) above, the Employer will designate in order of seniority from among those regular employees holding an appropriate Occupational First Aid Certificate to act as the First Aid Attendant in addition to the normal requirements of the job.

(2) Where no employee within the work unit possesses an Occupational First Aid Certificate, the opportunity to obtain a Certificate will be offered to regular employees within the work unit in order of service seniority, provided the employee can meet the requirements of the WCB regulations to undertake the training in order to obtain an Occupational First Aid Certificate.

(3) In the event that the procedures outlined above do not meet the requirements of (a), a Union representative at the worksite will assist the

Employer to meet their obligations by approaching regular employees in the work unit on behalf of the Employer.

(4) Where (d)(1), (2) and (3) do not meet, within a reasonable period of time, the requirements of the Employer to achieve (a) above, the Employer may:

(i) recall a qualified auxiliary employee in order of seniority from those holding the appropriate Occupational First Aid Certificate, and/or

(ii) include an Occupational First Aid Certificate as a desirable qualification on a posting pursuant to Clause 12.01.

(5) Failing (4) above, the Employer may require the most senior regular employee within the work unit who can meet the requirements of the WCB regulations to undertake Occupational First Aid training in order to obtain a Certificate.

* 22.12 Violence In The Workplace

The Employer and the Union recognize the need for a safe working environment free of violence or threats of violence. Violence is defined as the attempted or actual exercise by a person of any physical force so as to cause injury to an employee and includes any threatening statement or behaviour which gives an employee reasonable cause to believe that they are at risk of injury. The Employer will implement a prevention program which includes, but is not limited to, the following elements:

(a) The Employer will conduct regular risk assessments in accordance with the *Workers' Compensation Act* and Occupational Health and Safety Regulation (OHSR).

(b) Should the Union notify the Employer of its concern for the safety of any worksite due to the potential of violence, the Employer will conduct a timely risk assessment to determine whether there is a risk of injury to employees.

(c) Where a risk of injury is identified from a violence risk assessment, the Employer will, in consultation with the Union, establish on a timely basis,

policies, procedures and work environment arrangements to eliminate, or if that is not possible, to minimize the risk to employees. Such initiatives shall include:

(1) training of employees in the means for the recognition and reporting of the potential for violence and in the appropriate means of protecting themselves from violence;

(2) policies, procedures, documentation and work environment measures to minimize or control the risk to employees from violence including policies and procedures for protection of employees who may be required to work alone, and this information will be made available to staff;

(3) policies and procedures for the reporting and investigation of incidents and corrective action in accordance with the *Workers' Compensation Act* and Occupational Health and Safety Regulation (OHSR).

(d) Subject to statutory limitation, employees shall be informed concerning the potential for violence from a client, a person in care or custody, or another member of the public, where such a person is known to have a history of violence.

(e) Where there is a risk of verbal abuse from a client, a person in care or custody or another member of the public, appropriate measures to protect employees shall be implemented.

(f) When an employee has suffered as a result of violence, the chief steward or designate of the BCNU shall be notified as soon as is reasonably possible.

(g) Each Employer shall designate an appropriate senior representative responsible for the development and support of crisis response teams for employees impacted by workplace violence. Critical incident stress defusing shall be immediately provided to employees who have suffered a work-related, traumatic incident. Critical incident stress debriefing and appropriate support shall be made available for all employees who have suffered as a result of the violence. Appropriate resources will be made available as soon as possible following the incident. Leave required to attend such defusing, debriefing or support sessions will be without loss of pay.

(h) An employee performing visitation to clients in the community shall have the right to request support where they are concerned about a potential violent

situation. Appropriate communication equipment will be provided to nurses for visitations.

(i) Should a patient with a history of violence towards staff be placed on a unit or, should a patient develop a history of violence towards staff while on a unit, the Employer will be required to take all reasonable steps to eliminate, reduce or minimize the risk of violence.

*22.13 Investigation of Accidents

(a) Consistent with Part 3, Division 10 of the *Workers' Compensation Act*, all accidents shall be investigated jointly by at least one representative designated by the Union and one representative designated by the Employer.

(b) Reports shall be submitted on the appropriate accident investigation form and copies sent to the designates from the Union and the Employer; the Occupational Health and Safety Committee; and the Workers' Compensation Board. Nothing in this Clause restricts the right of the Employer to require the Employer representative in (a) above, if a member of the bargaining unit, to complete other reports related to the accident under investigation.

(c) In the event of a fatality, the ministry shall immediately notify the President of the BCNU, or designate, of the nature and circumstances of the accident and arrange as soon as possible for a joint investigation.

ARTICLE 23 - EDUCATION POLICY

The Employer and the Union agree that one of the means of improving the quality of nursing care and the delivery of health-care services is to provide opportunities for nurses to acquire, and put into practice, knowledge and skills essential to this end. The Employer and the Union also recognize that the responsibility for continuing education of nurses is shared by the individual nurse, the professional organization, the Employer, education institutions, and health-care agencies.

23.01 Educational Leave and Assistance

(a) Requests for educational assistance will only be considered from regular employees who have completed their probationary period.

(b) The Employer may grant educational assistance in the form of leave (with or without pay), tuition fees, expenses and/or any other monetary aid, for education programs, courses, workshops and/or seminars. At no time, however, will the amount paid in salary together with any additional funding in the form of scholarships, bursaries, grants, etc., exceed 100% of the employee's normal basic salary. This is exclusive of tuition fees and other expenses which may be paid or shared by the Employer and of monies paid to the employee to provide for additional expenses for approved research or relevant courses.

(c) The duration of education leave granted to employees to take advanced or special training beneficial to both the Employer and the employee, may be for varying periods of up to one year. Under special circumstances, this leave may be renewed but only when the applicant shows evidence of satisfactory progress in this course during the initial leave period.

(d) (1) Applications for educational leave for periods of four months or longer must be submitted to the Employer at least two months prior to the beginning of the requested leave period.

(2) Applications for educational leave for periods of less than four months should be submitted to the Employer with as much lead time as practical.

(e) An eligible employee with three or more years of continuous service with the Employer may be granted sabbatical leave with full or partial pay or without pay, for a period of time approved by the Employer. During the sabbatical leave it would be the obligation of the nurse to attend continuing education programs, a research project, visit other facilities, interview leaders in the health field, donate work to a community agency, or any other legitimate task which would benefit the nurse's development.

If an employee has received financial assistance from the Employer, it would be the obligation of the employee, if requested by the Employer, to pass on this information and new knowledge to their co-workers through programs of in-service education.

(f) (1) An employee shall be granted leave without loss of basic pay to take courses at the request of the Employer. The Employer shall bear the full cost of the courses and related expenses.

(2) An employee shall be granted leave with pay to take courses at the direction of the Employer. The Employer shall bear the full cost of the courses and related expenses.

(g) Costs related to the basic preparation and training required for registration as a Registered Nurse or Registered Psychiatric Nurse shall be the responsibility of the individual employee.

(h) When requests for education leave are considered, an endeavour will be made to provide equalized opportunities to select courses for eligible employees in smaller communities within the Province and may include leave of absence without loss of pay, travel expenses and costs associated with smaller population centres.

 (i) (1) An employee granted educational leave with full or partial pay shall be required to sign a statement stating that, on the completion of the leave they will remain in the service of the Employer for a period equivalent to three times the length of their educational leave multiplied by the percentage of basic salary.

(2) Should they leave the service of the Employer before this period expires they shall refund to the Employer the total cost of their training, including allowance and expenses on a pro rata basis.

(3) An employee granted educational leave without pay shall be required to sign a statement to the effect that on completion of the training they will remain in the service of the Employer for a period equivalent to the leave granted, or refund any financial assistance granted under this section on a pro rata basis.

(j) Employees required to write an examination leading toward a degree, certificate, or diploma relevant to nursing shall be granted the full day(s) of the examination(s) plus necessary travelling time without loss of basic pay.

23.02 Joint Education Committee and Jurisdiction

(a) A joint Education Committee shall be constituted consisting of four representatives, two appointed by the Union, two appointed by the Employer and a Chair.

- (b) The Chair shall be appointed by the four representatives of the parties. If the representatives are unable to agree on an appointment within 30 days following the signing of this Agreement, either party may request the Director of the Collective Agreement Arbitration Bureau of the Labour Relations Board to appoint a Chair.
- (c) The Committee shall meet as required.
- (d) The Chair shall have the power to decide whether or not any matter under Clauses 23.01 or 23.04 presented by a party is properly for consideration by the Committee.
- (e) The role of the Committee shall be to review and adjudicate disputes with respect to requests for educational assistance in the form of leave (with or without pay), tuition fees, or other monetary aid.

In reviewing and/or adjudicating disputes with regard to this Article, consideration shall be given to the following:

- (i) operational requirements;
- (ii) budget constraints;
- (iii) the best interests of the employee;
- (iv) the best interests of the Public Service.
- (f) The Chair shall be paid one-half by the Union and one-half by the Employer. Each party shall pay the fees and expenses of its representatives.
- (g) If the Chair should resign or be unable to act, a new Chair shall be appointed in the manner provided in (b) above.

*23.03 Ancillary Bonuses

A nurse who acquires and maintains **dual** registration **as a Registered Nurse and Registered Psychiatric Nurse** under the *Health Professions Act* shall receive an allowance of \$23 per bi-weekly period.

23.04 Academic Bonuses

The following bonuses will be paid for the acquisition of the following academic qualifications:

(a) A regular employee who has received a Baccalaureate degree in Nursing from a recognized university in Canada or the United States will receive an additional \$46 bi-weekly.

(b) A regular employee who has received a Master's degree in Nursing from a recognized university in Canada or the United States will receive an additional \$57.49 bi-weekly.

(c) A regular employee who has received a Baccalaureate degree with major in psychology, sociology or in a course of study approved by the Employer from a recognized university in Canada or the United States, the content of which is related to nursing practice will receive an additional \$46 bi-weekly.

(d) A regular employee who has received a Master's degree in psychology, sociology or in a course of study approved by the Employer from a recognized university in Canada or the United States, the content of which is related to nursing practice will receive an additional \$57.49 bi-weekly.

(e) CNA/CHA Certification in Nursing Administration, BCHA/BCIT Certificate in Nursing Administration, BCHA/BCIT Certificate in Health Care Management, CHA Certificate in Departmental Management:

\$11.50 bi-weekly

(f) Successful completion of a post graduate certificate or diploma program of a minimum of one academic year (eight months) or its equivalent from a recognized college or similar educational facility in Canada or the United States, the content of which is related to nursing practice or to the supervision of nurses:

\$23 bi-weekly

(g) A certificate, diploma or degree substantially similar in content to (c), (d), (e) and (f) above, granted by an educational facility outside Canada or the United States, at the rate equivalent to the appropriate analogous qualification listed above.

(h) These bonuses are not cumulative; the amount shown against the highest qualification will be paid where an employee has more than one such qualification.

(i) Disputes with respect to the application of this Article shall be submitted to the Committee constituted under Clause 23.02 above, which shall determine whether or not the disputed bonus is payable. Such disputes may be referred to the Committee for determination only after Step 2 of the grievance procedure has been exhausted.

ARTICLE 24 - NO EMPLOYEE LIABILITY

24.01 Actions of Patients

(a) When, in the proper performance of their duties, employees are supervising patients participating in authorized activities or programmes, such employees shall not be held responsible for acts committed by such patients.

(b) When employees transport patients in their own or Employer vehicles in the proper performance of their duties, such employees shall not be held responsible for acts committed by such patients.

24.02 Personal Property Damage

(a) Upon submission of reasonable proof, the Employer will repair or replace employee's chattels damaged while on duty by the actions of a patient, provided such personal property be articles of a type suitable for use or wear while on duty.

(b) Upon submission of reasonable proof of damage by any patient to an employee's property the Employer will repair or indemnify.

ARTICLE 25 - HEALTH AND WELFARE

25.01 Basic Medical Insurance

All regular employees, whether full-time or part-time, may choose to be covered by the Public Service Medical Plan, for which the British Columbia Medical Plan is the licensed carrier. Benefits and premium rates shall be in accordance with the existing policy of the plan. The Employer will pay 100% of the regular premium.

*25.02 Extended Health Care Plan

The Employer shall pay the monthly premium for regular employees entitled to coverage under a mutually acceptable Extended Health Care Plan. Employees shall be able to obtain details of the Extended Health Care Plan benefits, **by accessing the Benefits Guide at** <u>http://www2.gov.bc.ca/assets/gov/careers/all-employees/pay-and-benefits/benefits-for-bargaining-unitemployees/bargaining_benefits_guide.pdf</u>, or contacting MyHR directly at http://www2.gov.bc.ca/gov/content/careers-MyHR.

25.03 Dental Plan

(a) The Employer shall pay the premium for employees entitled to coverage under a mutually acceptable dental plan which provides:

Part A - 100% coverage Part B - 65% coverage Part C - 55% coverage

(b) Orthodontic services are subject to a lifetime maximum payment of \$3,500 per patient.

*25.04 Group Life

(a) Employees shall, as a condition of employment, enroll in the group life plan of (b) below and shall complete the appropriate payroll deduction authorization form.

(b) For those regular employees hired on or after May 1, 1990 and any auxiliary who, on or after May 1, 1990, qualifies for Clause 25.04, the Employer shall provide a mutually acceptable group life plan with benefits equivalent to three times an employee's annual salary, with a minimum of \$80,000.

The Employer shall pay 100% of the premium as set out above and the employee shall pay the premium for any insurance over the base minimum.

(c) For those employees enrolled in the Group Life Plan under Clause 25.04 prior to May 1, 1990, the Employer shall provide a mutually acceptable group life plan with benefits equivalent to twice an employee's annual salary, with a \$40,000 minimum. The Employer shall pay 100% for the premium on the \$40,000 base and the employee shall pay the premium for any insurance over \$40,000.

In lieu of this provision, any such employee may opt, in writing, to be covered by the provision of (b) above. Should such an option be exercised, that employee shall not be entitled to the provision of Clause 27.17 of this Agreement and the exercising of the option is irreversible.

(d) The group life plan of (b) and (c) above shall include the following provisions for accidental dismemberment:

- (1) loss of both hands or feet the principal sum
- (2) loss of sight of both eyes the principal sum
- (3) loss of one hand and one foot the principal sum
- (4) loss of one hand or one foot and sight of one eye the principal sum
- (5) loss of one hand or one foot one-half the principal sum
- (6) loss of sight of one eye one-half the principal sum

(e) The Employer will implement an Advanced Payment Program for the terminally ill under the circumstances described in Information Appendix **B**.

25.05 Air Travel Insurance

(a) In the event of death or disability incurred while travelling by aircraft on business of the Government, regular and auxiliary employees shall be covered by the terms and conditions of the Government blanket insurance policy.

(b) The amounts specified in the policy shall be paid to employees in the case of disability; and in the case of death, to the employee's beneficiary as designated under the Group Life Plan, if any, or in the absence of such beneficiary, to the employee's estate.

(c) Coverage shall commence from the place of employment or residence, whichever may last occur, and end upon returning to the regular place of employment or residence, whichever may occur first. Employees are not covered while piloting an aircraft in the course of their duties unless employed or paid as a pilot, or unless otherwise authorized.

25.06 Medical Examination

Where the Employer requires an employee to submit to a medical examination or medical interview, it shall be at the Employer's expense and on the Employer's time, other than a medical examination under Appendix 3 Section 1.04.

25.07 Employment Insurance

Employment insurance coverage will be provided during the life of this Agreement for employees who would, if employed by a private employer, be eligible for coverage under the provisions of the *Employment Insurance Act*.

25.08 Employee and Family Assistance Program

The Employee and Family Assistance Program shall be the Program which is negotiated for the majority of unionized employees in the Public Service of British Columbia.

ARTICLE 26 - ACCUMULATION OF TIME

26.01 Compensation for Statutory Holidays, Overtime and Standby

(a) Wherever in this Agreement or in the Component Agreements there are provisions made for compensation for statutory holidays, overtime, or standby, the employee may elect (subject to the other provisions of this Article) to receive that portion of the compensation which is in excess of basic salary, as cash or as time off.

(b) When an employee elects to take such compensation as time off, the employee may accumulate such time so that on any given date the employee may have an accumulation of not exceeding ten days.

(c) An employee may elect at any time upon reasonable notice to convert their accumulation of time or a portion thereof to cash.

(d) An employee may elect to take their accumulated time or a portion thereof at a mutually agreed time.

(e) Notwithstanding (b) above the Bargaining Principals may agree that the ten day maximum accumulation may be exceeded for a specified time limited period and for a specified group of employees. Any such agreement must be in writing.

***26.02** Compensation for Surplus Time

An employee shall be entitled to take the earned surplus time, not required to be scheduled, referred to in Clause 3.05 of the Hospital Component Agreement (i.e., 4:2 10 days and 5:2 5 days [effective January 1, 2018, 11 days]) as cash or time off at their option. The scheduling of such time off shall be by mutual agreement at the local level subject to operational requirements. Such surplus taken in cash shall not be counted as part of the annual working hours. Such earned time off shall form part of the accumulation provided in Clause 26.01 above.

26.03 Requests for Time Off Procedures

(a) Where deemed appropriate by the parties at the local level, each worksite shall develop a procedure detailing when employees can apply for time off under this Article. Proposals to develop such a procedure shall be reasonably

considered by the Employer. Such procedures shall be mutually agreed upon by the Employer and the Local Union. Mutual agreement shall not be unreasonably withheld.

(b) The procedure in (a) above will set reasonable timeframes for when employees can apply for time off and the Employer will grant requests in a timely manner.

(c) All requests must be submitted on the approved request form.

(d) Requests for time off will be granted on a first come first serve basis. However, should more than one employee on a unit submit a request for the same days off, and the request forms are submitted on the same day, the most senior employee will be given preference.

(e) There shall be no restriction on requests for time off as long as the operational requirements of the unit are met.

ARTICLE 27 - PAYMENT OF SALARIES AND ALLOWANCES

*27.01 Salaries

(a) The salaries shall be in accordance with those rates negotiated by the parties and recorded in Appendix **5 Wage Schedules** of this Agreement.

(b) Former employees who were employed on the effective date of a salary or allowance increase shall receive full retroactivity upon written request to their payroll office.

(c) Increase rates of pay for classifications listed in Appendix 5 Wage Schedules on the following dates:

Date	Percentage Increase
April 5, 2015	1.0
February 7, 2016*	0.0
April 3, 2016	0.5
February 5, 2017*	1.0
April 2, 2017	0.5
February 4, 2018*	1.0

April 1, 2018	0.5
February 3, 2019*	1.0

Note: *Dates for Economic Stability Dividend – reference MOU #20 on ESD and Appendix 5 WAGE SCHEDULES.

*27.02 Pay Period

(a) (1) Employees shall be paid bi-weekly every second Friday.

(2) Except when an employee is terminating their services or being terminated, if an employee's pay cheque or direct deposit is for an amount in excess of their entitlement, they shall nevertheless receive the pay, and the adjustments will be made on a subsequent pay day. The employee will not be expected to suffer a delay in receiving pay while adjustments are being made.

(3) If the employee's pay cheque or direct deposit is not available on the pay day, the Employer shall arrange for the employee to be provided, on the same day, with a pay advance in the amount required to make up the employee's full entitlement to their basic pay.

(4) If the employee's pay cheque or direct deposit is short of basic pay, the Employer shall, upon the employee's request, arrange for the employee to be provided with an advance in the amount required to make up the employee's full entitlement to their basic pay. Such advance shall be provided not later than eight days after the pay period in which the basic pay was due.

(b) The Employer shall provide for the direct deposit (electronic funds transfer) of the employee's pay in a participating financial institution of the employee's choice on or before the appropriate pay day. Employee participation shall be compulsory.

(c) Auxiliary employees shall receive their pay cheque or direct deposit no later than the pay day at the end of the bi-weekly pay period immediately following the bi-weekly pay period in which the pay was earned.

(c) All premiums and allowances payable shall be paid no later than the pay day at the end of the second bi-weekly pay period after the pay period in which the premium was earned.

(e) A comprehensive statement detailing all payments, **premiums**, allowances and deductions shall be provided for each pay period.

*27.03 Increment Dates

(a) The increment date for a full-time employee receiving an appointment (initial employment, reclassification, promotion, demotion as the case may be) will be the first day of the pay period 18 months after the employee's date of appointment. Effective September 3, 2017, the increment date will be the first day of the pay period 12 months after the employee's date of appointment.

(b) The increment date for a part-time employee receiving an appointment (initial employment, reclassification, promotion, demotion as the case may be) will be the first day of the pay period after the completion of 2614.5 hours after their date of appointment or date they received their previous increment. Effective September 3, 2017, the increment date will be the first day of the pay period after the completion of 1743 hours after their date of appointment or date they received their previous increment. Regular working hours are defined as non-overtime hours.

(c) For the purpose of 27.03(a) and (b), up to 875 regular working hours paid as an Interim Permit Nurse will count towards the completion of **2614.5** (1743 **hours effective September 3, 2017**) regular working hours required for an increment at the Nurse 4 level, provided the 875 hours are also worked with the Employer.

(d) A leave of absence without pay, other than union, maternity, adoption, parental or education leave, for more than 30 days which occurs prior to an employee's anniversary date will defer the increment and the employee's anniversary date will be adjusted by a time period equivalent to the period of leave of absence. This date will become the employee's new anniversary date for increment purposes.

*27.04 Increments - Eligibility

Increments for employees eligible pursuant to Clause 27.03 shall be one full step in the appropriate salary range specified in Appendix **5**.

*27.05 In-hiring Rates of Pay

(a) The in-hiring rates of pay for new employees without previous experience other than student time shall always be at the minimum rates in the pay plan provided in Appendix 5.

(b) Where a new employee to the Public Service is employed for a regular or an auxiliary position, salary recognition of one increment **step** for every **eighteen months** experience shall be granted for relevant nursing experience as determined by the Employer, provided not more than two years have elapsed since such experience was obtained.

Effective **September 3, 2017,** salary recognition of one **annual** increment for every **one year's** experience shall be granted for relevant nursing experience as determined by the Employer, provided not more than two years have elapsed since such experience was obtained.

Any time spent in an educational program mutually acceptable to the Employer and the Union shall not be counted as experience but shall not constitute a break in service.

(c) Notwithstanding (a) and (b) above:

Nurses previously employed by a general hospital, Regional Health Authority, the Provincial Health Authority or by a society funded directly or indirectly by the Provincial Government who within 90 days of termination of employment receive an appointment to a regular position in this bargaining unit, and whose previously held position is deemed by the Employer to be commensurate with the new appointment, shall be placed on the step of the salary scale nearest to, but not greater than, their previous salary; otherwise, their commencing salary shall be in accordance with the provisions of (a) and (b) above.

27.06 Salary on Promotion or Reclassification

When an employee is promoted or reclassified to a higher paying position, the employee will receive the rate corresponding to the step in the new level that is nearest to 8% above the employee's former salary or the minimum of the new salary level, whichever is greater, but not more than the top step of the new level.

27.07 Salary on Demotion

(a) An employee shall not have their salary reduced by reason of a change in the classification of their position that is caused other than by the employee, including a reclassification or placement pursuant to Article 13. The employee shall remain at their current salary until the salary in the new classification attains that level, after which the employee will be entitled to regular salary increases, if any.

(b) When an employee is demoted to a lower classification for reasons other than (a) above, the reduction in salary will be the closest step to 8%, but where the differential between the employee's salary before demotion and the maximum salary of the lower position is greater than 8%, the salary on demotion shall be the maximum salary of the lower position.

27.08 Temporary Substitution

(a) Employees who are designated to accept the principal responsibilities of a higher paying position shall receive substitution pay corresponding to the step of the level of the position being substituted to that is nearest to 8% above the employee's current salary or the minimum of the level being substituted to, whichever is greater, but not more than the top of the salary range of the position being substituted to.

(b) Should the Employer decide to fill a temporarily vacant, higher paying position, designation shall be made according to classification seniority within the unit or sub unit always provided the designated employee is able or is reasonably anticipated to be able to perform the duties of the higher paying position in a satisfactory and efficient manner.

(c) (1) Where an employee has temporarily assumed the responsibilities of a higher paying position as described above, either by designation or at the request or with the consent of the Employer for a shift or longer, the employee shall be considered as designated as a temporary substitute and shall be paid as such from and including the first such shift of assumption of responsibilities.

(2) An employee shall receive the substitution rate of pay for a designated paid holiday, and for leave under Clauses 20.01 (Bereavement

Leave) and 20.14 (Special Leave) if they have worked the majority of the 60 work days preceding the holiday or leave at the substitution rate of pay.

(d) Subject to (e) below, substitution pay is not payable when an employee's current position normally requires periodic substitution in the higher position as defined in the classification grade descriptors.

(e) Where the classification grade descriptors requires periodic substitution:

(1) substitution pay shall not be payable for periods of substitution of 16 consecutive shifts or less in the higher position;

(2) substitution in excess of the 16 consecutive shifts shall be payable from the commencement of the first shift of substitution;

(3) substitution is not payable for any period of substitution during vacation relief in the higher position.

(f) An employee who is temporarily detailed to carry out the duties incidental to a position which is lower paid than the position which they normally hold shall be paid their normal salary.

(g) Employees substituting in excluded positions shall retain the advantages and be bound by the terms and conditions of employment as provided in this Agreement; subject, however, that they will be required to carry out all the functions and duties of the excluded positions.

(h) The Employer will send to the Union a list (and subsequent amendments) of persons with the authority to designate employees to substitute.

27.09 Temporary Appointments

When it is determined that a temporary appointment is to be used to fill a position then such a temporary appointment will be in accordance with directives established under the *Public Service Act*.

*27.10 Vehicle Allowance

(a) Vehicle allowances for all distances travelled on government business shall be paid to employees required to use their own vehicles in the performance

of their duties. The allowance shall cover the distance to and from the employee's place of residence up to a total maximum of 32 kilometers, only when the employee is required to have their vehicle at work for use in the performance of their duties.

The vehicle allowance shall be:

Date	Rate per km
April 1, 2016	53¢
April 1, 2018	54¢

(b) Where an employee's vehicle is damaged by a person in the care or custody of the Employer, or as a direct result of the employee being employed by the Employer, the Employer shall reimburse the employee the lesser of actual vehicle damage repair costs, or the cost of any deductible portion of insurance coverage on that vehicle up to a maximum of \$600.

* 27.11 Meal Allowance

Employees on travel status away from their headquarters shall be entitled to meal allowance for the time spent away from headquarters.

,,,,,,, _	
Meal	Allowance
Breakfast	\$12.00
Lunch	\$13.80
Dinner	\$23.25

Effective April 1, 2016

Meal	Allowance
Breakfast	\$12.00
Lunch	\$14.00
Dinner	\$24.00

27.12 Return to Higher-Paid Classification

When an employee returns to a higher previously held classification within one year of vacating a position at that classification level they shall return to the step in the higher grade which they enjoyed at the time of vacating their former position.

This provision does not apply to employees who had left the service of the Employer.

***27.13** Accommodation, Board and Lodging Allowance

(a) Accommodation, board and lodging allowance for employees required to work away from their headquarters shall be paid in accordance with regulations pursuant to the Treasury Board Order respecting Board and Lodging and Relocation Expenses.

(b) Employees on travel status who are required to obtain overnight accommodation shall be reimbursed upon production of receipts for one fiveminute telephone call home, to or within British Columbia, for each night away.

(c) Employees on travel status who stay in non-commercial lodging shall be entitled to claim \$30.75 (effective April 1, 2016) and \$31.65 (effective April 1, 2018) per day except where the lodging is supplied by the Employer. An employee submitting a lodging allowance claim shall not be entitled to reimbursement for commercial lodging costs for the same period.

27.14 Standby Provisions

Where regular employees are required to standby to be called for duty under conditions which restrict their normal off-duty activities, they shall be compensated in the proportion of one shift for each 24 hours standing by. An employee designated for standby shall be available during the period of standby at a known telephone number. No standby payment shall be made if an employee is unable to be contacted or to report for duty when required. The provisions of this Clause do not apply to part-time employees who are not assigned a regular working schedule. It is not the policy of the Employer to have employees on standby duties during Statutory Holidays.

*27.15 Retirement Allowance

Upon retirement from service, an employee who has completed 20 years of service, and who under the provisions of the *Public Sector Pension Plans Act* is entitled to receive a pension benefit on such retirement, is entitled to **be paid** an amount equal to their salary for one month, and for each full year of service exceeding 20 years, but not exceeding 30 years, is entitled to an additional amount equal to one-fifth of their monthly salary. The employee may opt to take the allowance as equivalent paid leave of absence to be taken immediately prior to retirement.

For purposes of this Article, one month's salary is:

bi-weekly salary x 26.0893 12

*27.16 Relocation Expenses

Employees who have to move from one geographic location to another after winning a competition, or at the Employer's request, shall be entitled to relocation expenses in accordance with Memorandum of Understanding **#14**. Employees shall not be entitled to relocation expenses where their new worksite is closer to their current residence.

- (a) Where an employee receives relocation expenses as a result of winning a competition, and subsequently resigns within the two-year period immediately following the relocation, they will be required to reimburse the Employer expenses paid on a pro rata basis.
- (b) Provisions of (a) above do not apply to employees who resign in order to care for a dependent child or who resign or are deemed to have resigned pursuant to Clause 12.08 or Article 13.

27.17 Death Benefit

(a) When an employee dies, the period for which salary shall be paid during the month in which death occurred shall be deemed to be the full month.

(b) Payment shall be made to the spouse or the estate for vacation leave accrued by the deceased employee to date of death.

(c) Where an employee dies while in the Public Service, the following amounts shall be paid to the dependent or dependents to be determined by the Commission, of the employee:

(1) If the employee has completed one continuous year in the Public Service, one month's salary.

(2) If the employee has completed two continuous years in the Public Service, two months' salary.

(3) If the employee has completed three continuous years in the Public Service, three months' salary.

(4) If the employee has completed four continuous years in the Public Service, four months' salary.

(5) If the employee has completed five continuous years in the Public Service, five months' salary.

(6) If the employee has completed six or more continuous years in the Public Service, six months' salary.

For purposes of this Article, one month's salary is:

bi-weekly salary x 26.0893 12

(d) This provision applies only to those employees employed prior to May 1, 1990 and who are enrolled in the group life plan of Clause 25.04(c).

27.18 Formula for Hourly, Daily and Partial Month Calculations

The formula for paying bi-weekly or hourly rates, allowances or premiums is as follows:

Annual Rate 26.0893	=	bi-weekly rate
<u>Monthly Rate x 12</u> 26.0893	=	bi-weekly rate
Bi-weekly rate 70	=	hourly rate

The daily rate shall be determined by multiplying the number of regular scheduled hours in the employee's day shift by the hourly rate.

For the purposes of converting a bi-weekly rate to a monthly rate, the formula will be as follows:

When an Article has a reference to payments at the "end of the month following the month" in which an event occurs, payment will be "at the end of the second pay period following the pay period" in which the event occurs.

Similarly, a reference to payments on specified dates will mean payment on the closest pay period pay day to the specified date.

27.19 Child Care Expenses

(a) Where an employee is requested or required by the Employer to attend:

(1) Employer endorsed education, training and career development activities, or

(2) Employer sponsored activities

which are not included in the normal duties of the employee's job, and are outside their headquarters or geographic location, such that the employee incurs additional child care expenses, the employee shall be reimbursed for the additional child care expenses up to \$60 per day upon production of a receipt.

(b) Where an employee, who is not on leave of absence, attends a course approved by the Employer outside the employee's normal scheduled work day such that the employee incurs additional child care expenses, the employee shall be reimbursed for the additional child care expense up to \$30 per day upon production of a receipt. This reimbursement shall not exceed 15 days per calendar year.

(c) Reimbursement in (a) or (b) shall only apply where no one else at the employee's home can provide the child care.

(d) The receipt shall be a signed statement including the date(s), the hourly rate charged, the hours of care provided and shall identify the caregiver/agency.

27.20 Travel Expense Reimbursement

The Employer shall provide for the direct deposit (electronic funds transfer) of travel expense reimbursement in a participating chartered bank, trust company or credit union of the employee's choice. Employee participation shall be compulsory except where access to a financial institution with capability of accepting direct deposit is not available.

ARTICLE 28 - CLASSIFICATION AND RECLASSIFICATION

It is understood and agreed that the terms of section 12(d) of the *Public Service Labour Relations Act* and section 25(1)(f) of the *Public Service Act* override and take precedence over the terms of any and all clauses of this Article.

28.01 Classification Grade Descriptors

The Employer agrees to supply the Union Bargaining Principals with the grade descriptors of those classifications in the bargaining unit specified within subsection 4(a) of the *Public Service Labour Relations Act*.

***28.02** Job Evaluation Plan

(a) The Employer agrees that no job evaluation plan pertaining to positions covered by this Agreement will be introduced without the mutual agreement of the parties.

(b) To facilitate the orderly introduction or change in job evaluation plans, the Joint Classification Committee shall consist of an equal number of representatives from the Employer and the Union.

(c) The Committee shall formulate the job evaluation plans used as a means to establish classification systems within the Nurses' Bargaining Unit and shall make a recommendation to the Bargaining Principals for ratification.

(d) The Committee may direct the formation and establish the terms of reference to subcommittees to undertake the mechanics of any study approved by this committee.

(e) Introduction and establishment of mutually agreed upon job evaluation plans shall be subject to mutual agreement as to timing, in conjunction with Clause 28.03.

28.03 New Classifications or Proposed Changes in Existing Classifications

(a) Where a new classification covered by this Agreement is introduced, or where the BC Public Service Agency (BCPSA) proposes changes to existing classification standards, the BCPSA shall inform the Union in writing of such proposals. The rate of pay and the placement of a new or substantially altered classification in a component shall be negotiated between the Employer and the Union.

(b) If the parties are unable to agree on the rate of pay and/or the placement in a component for the new or substantially altered classification, within 10 days of their first meetings, or such other period as agreed by the parties, the Employer may implement the new or substantially altered classification and attach a salary.

(c) The Union may then refer the matter within 30 days to a special arbitrator agreed by the parties who shall determine the new rate of pay.

(d) The new rate of pay and/or placement in a component shall become effective on the date agreed to by the parties or the date set by the arbitrator but, in any event not earlier than the date of implementation.

***28.04** Classification Appeal Procedure

An employee shall have the right to appeal, through the Union, the classification of the position they occupy. Such an appeal shall be in accordance with the provisions of this Clause and shall not be considered a grievance under Article 8.

(a) <u>Step 1</u>

(i) If an employee believes that the position they occupy is improperly classified, they may discuss their duties and responsibilities with their **excluded manager**, immediate supervisor **and union steward if requested by the employee**.

(ii) The **excluded manager or** designate shall, upon request and within 40 days after the request, provide the employee with a written job

description describing the latter's duties and responsibilities. Such job description shall be consistent with the employee's assigned duties.

(iii) The employee their union steward and the excluded manager will review the job description and identify in writing any areas where the job description is not consistent with the assigned duties. The agreed upon duties will be incorporated into the job description and signed by the excluded manager.

(b) <u>Step 2</u>

If the employee still believes that their position is improperly classified, the employee may initiate a formal appeal by completing a Classification Appeal Form and forwarding the completed form to the excluded manager, the Union and the BC Public Service Agency. The BC Public Service Agency shall review the appeal and respond to the Union with a full explanation of its decision within 60 days of the submission.

(c) <u>Step 3</u>

The Union will respond to the BC Public Service Agency with a full explanation within 90 days of receipt of the written response. The parties shall attempt to resolve any outstanding issues through consultation.

(d) <u>Step 4</u>

If the classification level remains in dispute, the Union and the BC Public Service Agency may agree to an alternative process in order to resolve the classification appeal.

(e) <u>Step 5</u>

If the above procedure does not lead to a satisfactory resolution, within 60 days, the employee shall have the right to appeal the results through the Union to the Classification Referee in article 28.05, by providing written notification to the BC Public Service Agency.

(f) The time limits in this Clause may be extended by mutual agreement.

*28.05 Classification Referee

The parties shall agree to a list of classification referee(s) who shall hear classification appeals. The referee shall be assigned to hearings, depending upon availability, on a rotating basis from the list of referees. The order of rotation may be varied by mutual agreement of the parties.

(a) The Union's submission to the referee, with a copy to the BCPSA, shall include:

- (1) Name of employee (s)
- (2) Position title and PMS number
- (3) Ministry
- (4) Location

(5) A copy of the "Classification Appeal Form" submitted by the employee in accordance with the classification appeal procedure.

- (6) A copy of the BCPSA's response to the employee's appeal.
- (7) The classification requested by the Union.

(8) An explanation of the duties the employee is required to perform which would conform to the classification sought.

(9) The relevant classification grade descriptors.

(b) Within 30 days of receipt of the Union's submission to the referee the BCPSA may make a submission to the referee with a copy to the Union providing additional information in support of its decision.

(c) In arriving at a decision the referee is restricted to the application of the classification system in effect for members of the bargaining unit and shall confine their judgments to classifying or evaluating positions being appealed against the classification grade descriptors.

(d) Recommendations and salary awards are not within the purview of the classification referee.

(e) Effective date of any reclassification shall be the first day of the pay period following the date of receipt by the **BC Public Service Agency** of the employee's "Classification Appeal Form" **pursuant to Clause 28.04 (b).**

(f) The classification referee's recommendations, consistent with these terms of reference, shall be implemented by the Employer.

ARTICLE 29 - AUXILIARY EMPLOYEES

29.01 Terms of Employment

(a) An auxiliary employee shall receive a letter of appointment immediately following recruitment, clearly stating their employment status as defined in Clause 1.02 Definitions, and their expected duration of employment.

(b) (1) "On-Call Employees" at the time of recruitment shall mutually agree upon the days and hours of availability. Continuation on the "On-Call" list will be dependent on the employee fulfilling the terms of employment set out in the letter of appointment.

(2) Calls for work shall be made on the basis of seniority within the agreed upon hours of availability.

(3) When calls are made by the Employer for auxiliary employees to report for work in excess of that expressed in the letter of appointment, the acceptance of such work shall be at the employees' discretion.

(c) (1) Auxiliary employees who report for work at the call of the Employer pursuant to (b) above shall be paid for all hours worked with a minimum of two hours pay at their regular rate unless the employee is unfit to perform their duties or has failed to comply with the Industrial Health and Safety Regulations of WorkSafe BC.

(2) Where an employee commences work they shall receive three and one-half hours pay at their basic pay unless:

(a) their work is suspended for reasons completely beyond the control of the Employer; or

(b) the duration of the work assignment is known in advance by the employee.

In which instances the provisions of Clause (c)(1) shall apply.

(3) An auxiliary employee who is called back to work in a circumstance such that they would be entitled to overtime compensation for the time worked, shall be compensated for a minimum of three hours at overtime rates. They shall be compensated from the time they leave home to report for duty until the time they arrive back upon proceeding directly to and from work.

*29.02 Seniority on Applying for Regular Positions

(a) Employees who have completed 210 working hours at straight time rates as outlined in Clause 29.02(b) will be recognized as in-service applicants when applying for appointment to "regular" positions.

(b) Employees who have worked 210 hours at straight time rates immediately prior to application for a "regular" position, or casual employees who have worked 210 hours prior to being laid off, will have their length of service as an auxiliary employee recognized in accordance with Section 8 of the *Public Service Act*.

(c) Employees who have completed 210 working hours at straight time rates as outlined in Clause 29.02(b) and who have to move from one geographic location to another after winning a competition, or at the Employer's request, shall be entitled to relocation expenses in accordance with Memorandum of Understanding **#14**.

29.03 Seniority

- (a) (1) For the purpose of layoff and recall and other seniority related provisions, and classification seniority in an auxiliary seniority block, an auxiliary employee who has worked in excess of 30 days shall accumulate service and classification seniority within an auxiliary seniority block, on the basis of:
 - (i) all hours worked at the straight time rate;

(ii) paid holidays or days off in lieu in accordance with Clause 29.08 of this Agreement;

(iii) vacation in accordance with Clause 29.09 of this Agreement.

(iv) compensatory time off provided the employee has worked 1827 hours in 33 pay periods;

(v) missed work opportunities during Union leaves pursuant to Clause 20.03(a) to (d), except that during the first 33 pay periods, such credit shall be limited to 105 hours.

(2) The total hours above shall be converted to seven hour shifts to establish seniority.

(3) Upon completing 30 working days (7 hour shifts), an auxiliary employee's seniority shall include the accumulated 30 working days.

(b) Subject to Clause 29.04, service and classification seniority of an auxiliary employee shall move with them where such a movement to a different auxiliary seniority block, including across jurisdictional units, is approved by the Employer.

(c) Auxiliary employees who are on a claim recognized by WorkSafe BC which arises out of a work-related injury while employed by the Employer, shall earn seniority for all hours the employee would have worked had they not been injured and been able to stay on the job.

(d) For the purposes of Clause 29.03 only, "auxiliary seniority blocks" shall be identical as are the seniority blocks identified in Clause 13.03(e) and (f).

29.04 Loss of Seniority

An employee will lose any entitlement to service and classification seniority when:

- (a) they are terminated for just cause;
- (b) they voluntarily terminate or abandon their position;
- (c) as a casual employee, they are on layoff for more than six months;

(d) they are a casual employee and are unavailable or decline two offers of re-employment in which the duration and nature of work is reasonably similar to that which they carried out prior to layoff;

(e) they become a regular employee.

(f) they are an on-call employee and do not fulfill the terms of employment set out in the letter of appointment.

However for the purposes of Articles 12 and 13, with the exception of Article 13.10, an employee who changes their status from regular to auxiliary, or from auxiliary to regular, will maintain their previous seniority providing the break in service has not exceeded 30 days. This provision applies only to employees who change their status after January 1, 2011.

29.05 Layoff and Recall

Layoff of casual auxiliary employees shall be in reverse order of seniority. Employees on layoff shall be recalled in order of seniority. Seniority accumulated as a "regular" employee, including training course time under Clause 11.06 of this Agreement, will be recognized providing the break in service has not exceeded 30 days.

***29.06** Application of Agreement

(a) The term "employee" used in other articles of this Agreement includes "casual employees" except in Articles 11, 13, 17, 18, 19, 20, 21, 23, and 25 and any other article or clause which specifies "regular" employees, unless specified otherwise in this Article.

(b) The term "employee" used in other articles of this Agreement includes "oncall" employees except in the following numbered articles and clauses: Articles 11, 13, 17, 18, 19, 20, 21, 23, 25, 26; Clauses 14.05, 15.02, 15.05, 27.03, **and** 27.06, and any other clauses which specify "regular" employees unless otherwise specified in this Article.

(c) An auxiliary employee who is a member of the Union's negotiating committee shall be eligible for leave in accordance with clauses 20.03(b), (c), and (f).

(d) Time spent at court by an auxiliary employee in their official capacity shall be at their regular rate of pay.

(e) Court actions arising from employment which require an auxiliary employee's attendance at court, shall be with pay.

(f) Any auxiliary employee who is eligible to vote in a Federal, Provincial, Municipal or Aboriginal Community Government Election or a referendum shall have three or four consecutive clear hours, as prescribed by the applicable statute, during the hours in which the polls are open in which to cast their ballot.

(g) Where leave from work is required, employees shall be entitled to the provisions of Clause 20.01 (Bereavement Leave).

(h) Maternity and parental leave for auxiliary employees with less than 1827 hours worked in 33 pay periods shall be in accordance with the *Employment Standards Act*.

*29.07 Health and Welfare

(a) In lieu of health and welfare benefits, auxiliary employees shall receive compensation of:

Date	Rate per hour	Max biweekly
April 1, 2016	72¢	\$50.40
April 1, 2018	74¢	\$51.80

(b) Clause (a) above will not apply when an auxiliary is receiving benefits under the provisions of Clause 29.11 (weekly indemnity).

(c) (i) Auxiliary employees who have worked 1827 hours in 33 pay periods or after working three consecutive years without loss of seniority and maintaining 1200 hours worked at the straight time rate within the previous 26 pay periods will be eligible for coverage under Clauses 25.01, 25.02, 25.03, 25.04 and 25.08. Such auxiliary employees eligible for benefits under this Clause will not receive the payment under Clause 29.07(a).

(ii) An auxiliary employee moving to a new auxiliary position in the same ministry, without any break in service will retain if already qualified, health and welfare benefits under (i) above but if not yet qualified will retain accumulated time in that ministry towards the required 1827 hours worked.

(iii) An auxiliary employee moving from one ministry to an auxiliary position in another ministry, without any break in service, will retain if already qualified, health and welfare benefits under (i) above; but if not yet qualified, will lose any hours accumulated toward 1827 hours worked.

(iv) For purposes of this Article, the term "no break in Service" means that the "employee" moves from the present position to the new position on the next available working day, taking into account shift changes, weekends, statutory holidays or any other approved scheduled time off.

(d) An auxiliary employee will cease to be entitled to coverage under (c)(i) above when they lose their seniority in accordance with Clause 29.04(a), (b), (c), (d) or (f).

(e) Auxiliary employees qualified under (c) above shall be entitled to maintain coverage under such plans for a maximum period of six consecutive months immediately following the month in which the layoff occurs by paying the premium themselves.

(f) (i) An auxiliary employee on layoff, who is recalled to an auxiliary position by the same ministry, without loss of seniority pursuant to (d) above, will retain, if already qualified, health and welfare benefits under (c)(i) above; but if not yet qualified they will retain accumulated time in that ministry towards the required 1827 hours worked.

(ii) An auxiliary employee on layoff, who moves to, or is recalled by another ministry to an auxiliary position, without loss of seniority pursuant to (d) above, will retain, if already qualified, health and welfare benefits under (c)(i) above; but if not yet qualified, they will not be entitled to carry any hours from the previous ministry towards the required 1827 hours worked, into the new ministry.

29.08 Paid Holidays

- (a) Designated paid holidays are as specified in Clause 17.01.
- (b) Auxiliary employees shall be compensated for the paid holiday who have:

(1) worked, or received pay at straight time rates for the day before and the day after a paid holiday; or

(2) worked, or received pay at straight time rates for 15 of the previous 30 days; or

(3) worked, or received pay for at least 105 hours at the straight time rate in the previous 30 days.

This Clause shall not apply to employees who have been terminated and not on layoff status.

(c) An auxiliary employee who is qualified under (b) to receive compensation for the paid holiday shall receive a prorated day's pay based on the following formula:

straight times hours paid in the previous 30 calendar days divided by the straight time hours of work of a full-time employee for the same 30 calendar day period.

(d) An auxiliary employee who is qualified in (b) to receive compensation for the holiday and who is required to work on that day shall be compensated at the same rate as regular employees in the same situation, as outlined in Article 17.

29.09 Annual Vacation

(a) Auxiliary employees who have worked 1827 hours in 33 pay periods will be entitled to receive the same vacation entitlement benefits as regular employees. When an auxiliary employee schedules their vacation, their selection(s) of vacation period(s) shall be subject to operational requirements and shall not restrict the choices of regular employees.

(b) Except as indicated in (a) above, an auxiliary employee shall be entitled to receive vacation pay at the rate of six per cent of their regular earnings. Auxiliary employees shall receive their earned vacation bi-weekly.

Such an employee shall be entitled to three weeks each calendar year free from call of the Employer for the purpose of vacation leave. An employee will provide a minimum of two **weeks'** notice in writing for this purpose, and the mutually agreed period free from call shall not restrict operational requirements or interfere with the vacation schedules of regular employees.

(c) A casual employee may elect to use their vacation entitlement subject to the provisions of this Agreement and the component agreements. Casual employees shall receive any unused portion of annual vacation upon termination of employment or on the last pay day of the year in which the vacation was earned.

(d) An "on-call" employee shall receive vacation payment in cash only.

(e) The provisions of this section are not affected by an employee having once been a "regular" employee.

(f) (i) An auxiliary employee moving to a new auxiliary position in the same ministry, without any break in service will retain if already qualified, annual vacation entitlement benefits under (a) above; but if not yet qualified will retain accumulated time in that ministry towards the required 1827 hours worked.

(ii) An auxiliary employee moving from one ministry to an auxiliary position in another ministry, without any break in service, will if already qualified, retain annual vacation entitlement benefits under (a) above; but if not yet qualified, will lose any hours accumulated toward 1827 hours worked.

(iii) An auxiliary employee on layoff, who is recalled to an auxiliary position by the same ministry, without loss of seniority, will retain, if already qualified, annual vacation entitlement benefits under (a) above; but if not yet qualified they will retain accumulated time in that ministry towards the required 1827 hours worked.

(iv) An auxiliary employee on layoff, who moves to, or is recalled by another ministry to an auxiliary position, without loss of seniority, will retain, if already qualified annual vacation entitlement benefits under (a) above; but if not yet qualified, they will not be entitled to carry any hours from the previous ministry towards the required 1827 hours worked, into the new ministry.

(v) For purposes of this Article, the term "no break in Service" means that the "employee" moves from the present position to the new position on the next available working day, taking into account shift changes, weekends, statutory holidays or any other approved scheduled time off.

29.10 Eligibility Requirements for Benefits

Auxiliary employees will qualify for Short Term Illness and Injury Plan (STIIP), Clauses 20.03, 20.04, 20.06, 20.09, 20.12, 20.13, 20.14, 20.15 and Article 21 as follows:

(a) An employee shall be entitled to benefits under this Clause after completion of 1827 hours worked in 33 pay periods.

(b) An auxiliary employee will cease to be entitled to coverage when they:

(1) fail to maintain 1200 hours worked at the straight time rate within the previous 26 pay periods, except as provided under Article 21.

(2) lose seniority in accordance with Clause 29.04(a), (b), (c), (d) or (f).

(c) Benefits will not be paid on layoff.

(d) Auxiliary employees on layoff or subject to recall will not be eligible for benefits until their return to work and subject to meeting the eligibility requirements.

("Return to work" is understood to mean the employee completed at least onehalf of a scheduled work day.)

(e) Where there is no established work schedule, the calculation of hours for the purpose of STIIP benefits shall be based on the average number of hours worked during the three month period immediately preceding absence due to illness.

29.11 Weekly Indemnity

(a) Auxiliary employees are eligible for weekly indemnity benefits upon accumulation of 400 hours. Once established, eligibility for weekly indemnity is retained unless the auxiliary employee loses auxiliary seniority. Weekly indemnity benefits are payable for each period of illness up to a maximum of 15 weeks at 60% of the auxiliary employee's normal average earnings. Normal average earnings are calculated by averaging the total of the straight time compensation and the compensation paid in accordance with Clause 29.07(a) in the six most recent bi-weekly pay periods in which earnings occurred. (b) The benefit waiting period in each case of illness will be 14 calendar days. This means that benefits will be paid from the 15th day of illness.

(c) Subject to (c) above, full benefits will be reinstated:

(1) in the case of a new illness, after the auxiliary employee returns to active employment following the most recent absence due to illness and accumulates 150 more hours;

(2) in the case of a recurrence of a previous illness, after the auxiliary employee returns to active employment following the most recent absence due to that illness and accumulates 400 more hours.

(d) The payment of benefits to a person who is laid off or separated prior to termination of their illness shall be continued after the layoff or separation until the total number of weeks for which benefits have been paid in respect of that illness is 15 weeks or the duration of the illness, whichever occurs first, except that benefits will cease on the effective date of a scheduled layoff or separation, if the illness occurs two months (or less) before that layoff or separation, provided that notice of the layoff or separation was given prior to the occurrence of the illness.

(e) The benefits described in this Clause shall not be available to an employee whose illness, injury or personal circumstances may be described by any one of the following conditions:

(1) who is not under the care of a licensed physician;

(2) whose illness is occupational and is covered by Workers' Compensation;

(3) whose illness is intentionally self-inflicted;

(4) whose illness results from service in the Armed Forces;

(5) whose illness results from riots, wars or participation in disorderly conduct;

(6) who is ill during a period of paid vacation;

(7) whose illness is sustained while they are committing a criminal offence;

(8) who is engaged in an employment for wage or profit;

(9) who is ill during a strike or lockout at the place they were employed if that illness commenced during the strike or lockout;

(10) who is serving a prison sentence;

(11) who would not be entitled to benefits payable pursuant to Part II of the *Employment Insurance Act* because they are not in Canada;

(12) who is absent from work because of plastic surgery performed solely for cosmetic purposes except where the need for surgery is attributable to an illness or injury.

(f) The parties agree that the complete premium reduction from the Employment Insurance Commission accruing through the improved sick leave plan and the weekly indemnity plan will be returned to the Employer. This is in exchange for the implementation of the above mentioned plans.

*29.12 Increments

Employees will receive an increase to the next step within their salary scale after they complete 2614.5 non-overtime working hours (effective September 3, 2017 - 1743 hours) from the date on which they qualified to receive their last increment; or from the date they commenced employment, whichever is the later.

29.13 Changing Status

(a) An auxiliary employee who is appointed as a regular employee to a classification on the same grid level, provided their break in service has not exceeded 30 days, shall retain the same increment step they had earned as an auxiliary employee, and time worked towards receiving an increment under Clause 29.12 will be used in determining the employee's increment date under Clause 27.03 of this Agreement.

(b) A regular employee who is offered and accepts an auxiliary appointment or resigns and is re-employed as an auxiliary employee within 30 days of their effective date of resignation, shall:

(1) retain the same increment step as that enjoyed as a regular employee provided their appointment is to a classification on the same grid level

(2) be eligible for future increments pursuant to Clause 29.12, provided their appointment is to a classification on the same grid level

(3) be deemed to have resigned their regular appointment.

(c) Where such employee described in (b) above has completed 1827 hours worked in 33 pay periods at the date of their auxiliary appointment, the employee will be considered to have met the requirements of Clauses 29.10(a) and 29.07(c). Such employee shall also retain the vacation year attained as a regular employee.

(d) Where such employee described in (b) above had not completed 1827 hours worked in 33 pay periods at the date of their auxiliary appointment, they shall be credited with non-overtime hours worked towards becoming eligible for the benefits in Clause 29.07.

*29.14 Conversion of Auxiliary Employees

(a) Auxiliary employees who have worked 1827 hours in 33 pay periods and who are employed for work which is of a continuous full-time or part-time nature, shall be converted to regular status effective the beginning of the month in which they attain the required hours.

(b) Notwithstanding (a) above an auxiliary employee shall have the right to decline conversion to regular status. In such circumstances, the employee shall notify **their excluded manager** in writing and the provisions of this Clause shall not apply. From the date of that notification an employee may requalify for conversion by working an additional 1827 hours in 33 pay periods and satisfying the provisions of (a) above.

(c) For the purpose of (a) above and Clauses 29.06 (Application of Agreement), 29.07(c)(i) (Health and Welfare), 29.09 (Annual Vacation), and 29.10 (Eligibility Requirements for Benefits) hours worked shall include:

(1) hours worked at the straight-time rate;

(2) hours compensated in accordance with Clause 29.08 (Paid Holidays);

(3) hours that a seniority rated auxiliary employee cannot work because they are on a recognized WCB claim arising from their employment with the Employer to a maximum of 210 hours of missed work opportunity within eight weeks from the beginning of the claim;

(4) annual vacation pursuant to Clause 29.09;

(5) compensatory time off provided the employee has worked 1827 hours in 33 pay periods;

(6) missed work opportunities during Union leaves pursuant to Clause 20.03(a) to (d), except that during the first 33 pay periods of employment such credit shall be limited to 105 hours;

(7) extra days off pursuant to Hospital Services Component Clause 5.03(a)(1) and (2).

(8) Notwithstanding (3) above, an auxiliary employee eligible for conversion to regular status shall not be converted until the employee has returned to active employment for 140 hours. The effective date of such conversion shall be the first of the month following the date on which eligibility for conversion occurs.

(d) For the purpose of (a) above and Clauses 29.06, 29.07, 29.09, and 29.10, hours beyond the 210 hours in (c)(3) above, that an auxiliary employee cannot work because they are on a WCB claim arising from their employment with the Employer, are not added to the 1827 or 1200 hours nor are the days charged against the 33 or 26 pay periods.

ARTICLE 30 - GENERAL CONDITIONS

30.01 Parking

The Employer and the Union agree that there will be no change in the parking regulations and notices except by mutual agreement between the parties. A joint

Employer-Union parking committee shall be established to study the matter of employee parking and make recommendations to the parties.

30.02 Termination of Employment

(a) Regular employees who have successfully completed their probationary periods, other than employees affected by layoff under the provisions of Article 13, whose employment is terminated by the Employer, shall be given 28 calendar days' notice or 20 days' pay in lieu of notice except employees whose employment is terminated for just cause.

(b) A regular employee who wishes to terminate their employment shall give the Employer 28 calendar days' notice prior to the effective date of termination.

(c) Unless by mutual agreement to the contrary, the time stated in Clause 30.02(a) and (b) shall be time scheduled to work and shall not include accrued vacation time.

30.03 Standing Joint Committee

There shall be a high level Standing Joint Committee comprised of equal numbers from the Union and the Employer to discuss policies, communications etc. and to study issues arising from changing conditions and concepts in the delivery of health care services including the monitoring of hours of work schedules developed pursuant to the applicable Component Agreement. Such hours of work schedules may include Master rotation schedules which may be developed by mutual agreement at the local level.

Copies of any such Master rotation schedules shall be sent to the Union and the BC Public Service Agency for review.

*30.04 Indemnity

(a) <u>Civil Actions</u>

Except where the joint Union-Employer Committee considers that there has been flagrant or willful negligence on the part of an employee:

(1) The Employer agrees not to seek indemnity against an employee whose actions result in a judgment against the Employer.

(2) The Employer agrees to pay any judgment against an employee arising out of the performance of their duties.

(3) The Employer agrees to pay any legal cost incurred in the proceedings including those of the employee.

(b) <u>Criminal Actions</u>

Where an employee is charged with an offence resulting directly from the proper performance of their duties, the employee shall be reimbursed by the Employer for reasonable legal fees, unless the employee is subsequently found guilty of the charge.

(c) At the option of the Employer, the Employer may provide for legal services in the defence of any legal proceedings involving the Employer, so long as no conflict of interest arises between the Employer and the employee, or pay the legal fees of counsel chosen by the employee.

(d) In order that the above provisions shall be binding upon the Employer, the employee shall notify the Employer immediately, in writing, of any incident or course of events which may lead to legal action against them, and the intention or knowledge of such possible legal action is evidenced by any of the following circumstances:

(1) when the employee is first approached by any persons or organizations notifying them of intended legal action against them;

(2) when the employee themselves requires or retains legal counsel in regard to the incident or course of events;

(3) when any investigation or other proceeding which might lead to legal action against the employee;

(4) when information first becomes known to the employee in the light of which it is a reasonable assumption that the employee would conclude that they might be the object of legal action;

(5) when the employee receives notice of any legal proceeding of any nature or kind.

(e) Where an employee is required to defend their professional actions arising out of the proper performance of their duties with the Employer in proceedings

taken pursuant to the *Health Professions Act* or pursuant to an Act(s) which succeed **it** the Employer will provide either legal counsel, or, at the Employer's option, reimbursement of reasonable legal fees incurred in such defense.

30.05 Payroll Deductions

Where provisions of this Agreement call for payment or a contribution by the employee it shall be by payroll deduction.

An employee shall be entitled to have deductions from their salary assigned for the purchase of Canada Savings Bonds or bonds issued by the Provincial Government.

30.06 Political Activity

(a) Employees may seek election to municipal and school board offices and Aboriginal Community Government provided that there is no conflict of interest and the office does not affect the performance of the person's duties in the Public Service.

(b) There are no restrictions other than the oath of office on employees engaging in political activities on their own time as campaign workers. If an employee is nominated as a candidate for election the employee shall be granted leave without pay to engage in the election campaign. If elected, the employee shall be granted leave of absence without pay for a maximum of five years. If not elected, the employee shall be allowed to return to their former position.

***30.07** Personnel Files

An employee, or their Union designate with the written authority of the employee, shall be entitled to review the employee's personnel file. The employee or designate, as the case may be, shall give the employer adequate notice prior to having access to such file. Files will be sent electronically to an employee through the government e-mail system, or at the employee's request, to their union designate.

30.08 Employment of Relatives

There shall be no restriction against relatives working in the same hospital, clinic or facility subject to Article 6 of this Agreement, and subject to there being no supervisor-subordinate relationship between the direct relatives and that no special

consideration with respect to the application of the provisions of this Agreement is invited.

30.09 Copies of Agreement

(a) The Union and the Employer desire every employee to be familiar with the provisions of this agreement, and their rights and obligations under it. For this reason, the Union and the Employer will make the Agreement available electronically to all employees. A limited number of copies will be printed for distribution to employees that do not have access to computers at work. The cost of such printing and distribution shall be borne equally by the parties.

(b) Where required, the Union shall distribute the collective agreements to its members and the Employer shall reimburse the Union for 50 percent of the distribution costs.

30.10 Technical Information

The Employer agrees to provide the Union, on request, such information, as and if it is available relating to employees in the bargaining unit as may be required by the Union for collective bargaining purposes which may include an employee's mailing address including postal code, employment site, work and home telephone numbers (unless unlisted), job classification, and employment status of regular full-time, regular part-time, or auxiliary.

30.11 Excessive Work Load

The Employer acknowledges that, except in the case of emergency, the work load of an employee covered by this Agreement will not be unduly increased as a result of positions vacant due to illness, vacation, or recruitment lag.

30.12 Previous Policies and Letters of Understanding

This Agreement voids all previous Agreements, Letters of Understanding, memos, policies, or provisions of the Manual of Personnel Administration which in any way alter the terms and conditions contained herein.

30.13 Statement of Duties and Responsibilities

The Employer's designate shall, upon request and within a reasonable period after the request, provide the employee with a written statement of the latter's duties and responsibilities or, in the alternative, a job description.

30.14 Clothing and Uniforms

(a) <u>Clothing</u>

(i) The type of clothing worn by an employee (uniforms or street clothing) shall be as determined by mutual local agreement compatible with the type of work involved. If a uniform is necessary, it shall be of a type and style agreed upon locally. The Employer shall provide the appropriate uniform to employees required to wear a uniform.

(ii) Where conditions are such that either a uniform or street clothes is appropriate, the option of the type of wearing apparel shall be at the discretion of the employee.

(b) <u>Protective Clothing</u>

Wearing apparel required to protect the employee's clothing, to provide aseptic techniques, or non-standard apparel for isolation, or similar work areas, shall be determined by mutual local agreement and shall be provided, maintained and retained by the Employer.

(c) <u>Pins</u>

Except where it is a hazard to patients, employees shall be allowed to wear their graduation pin, Union lapel pin and Association registration pin.

ARTICLE 31 - TERM OF AGREEMENT

***31.01** Expiration of Agreement

This Agreement covers the period from and including April 1, **2014** to and including midnight, March 31, **2019**. All terms and conditions of this Agreement shall remain in full force and effect after March 31, **2019** until the Union gives notice of strike and a strike occurs, or until the Employer gives notice of lockout and lockout occurs, or until a new or amended Agreement comes into force, whichever is earliest, and as may be provided by statute.

***31.02** Notice to Bargain

(a) This Agreement may be opened for collective bargaining by either party giving written notice to the other party on or after January 1, **2019**, but in any event not later than midnight January 31, **2019**.

(b) Where no notice is given by either party prior to January 31, **2019**, both parties shall be deemed to have been given notice under this section on January 31, **2019**, and thereupon Clause 31.03 of this Article applies.

31.03 Commencement of Bargaining

Where a party to this Agreement has given notice under Clause 31.02 of this Article the parties shall, within 14 days after the notice was given, commence collective bargaining.

31.04 Changes in Agreement

Any change deemed necessary in this Agreement may be made by mutual agreement at any time during the life of this Agreement.

***31.05** Effective Date of Agreement

The provisions of this Agreement, except as otherwise specified, shall come into force and effect on the date of signing. The date of signing of the **Sixteenth** Agreement is **September 3, 2017**.

SIGNED ON BEHALF OF THE UNION Date:

Laura Anderson Labour Relations Officer/Negotiator BC Nurses' Union

Lola Backeland BCNU Bargaining Committee

Christine Brisebois BCNU Bargaining Committee

Arvita Cotter BCNU Bargaining Committee

Karen Edwards BCNU Bargaining Committee

Lawrence Egan BCNU Bargaining Committee

Lisa Isaac BCNU Bargaining Committee SIGNED ON BEHALF OF THE EMPLOYER Date:

Patricia Brown Senior Labour Relations Specialist BC Public Service Agency

Sarah Campbell Senior Labour Relations Specialist BC Public Service Agency

Andrea Cruickshank Employee Relations Specialist BC Public Service Agency

Tom Jensen Director Maples Adolescent Treatment Centre, MCFD

Andrew Cronkhite Director Burnaby Youth Secure Custody Centre, MCFD

Martin Bartel Director of Operations Service Delivery – East Fraser, MCFD

Agnes Tao Assistant Director, YFPS Ministry of Children and Family Development

Claudia Calderon Compensation Advisor BC Public Service Agency

Deborah Myles Office Manager BC Public Service Agency

*APPENDIX 1 COMPONENT STRUCTURE & JURISDICTIONAL AREAS

There shall be two components in this bargaining unit as follows:

(a) Community Services Nurses Component Structure and Jurisdictional Areas:

All nurses who are employed in a program which provides other than 24 hour per day in-patient care services.

At the time of this writing, these facilities are:

- **1.** Youth Forensic Psychiatric Services by geographic location
- Ministry of Public Safety and Solicitor General Road Safety BC
- **3.** Ministry of Health
- 4. Ministry of Children and Family Development Child and Youth Mental Health by geographic location
- (b) Hospital Services Nurses Component Structure and Jurisdictional Areas

All nurses who are employed in a program which provides 24 hour per day inpatient care within a facility, including facilities at the time of this writing, being:

- 1. Maples Adolescent Treatment Centre
- 2. Youth Forensic Psychiatric Services In-Patient Assessment Unit
- **3.** Burnaby Youth Custody Services
- 4. Prince George Youth Custody Services

*APPENDIX 2 EXCLUSIONS

Pursuant to Clause 2.02 of this Agreement, incumbents of the following positions shall be excluded from this Agreement including the Component Agreement. Changes to the number and class of incumbents of positions may be agreed to by the parties from time to time.

HOSPITAL SERVICES NURSES COMPONENT

As at February 22, 2017 there are no Appendix 2 Exclusions

COMMUNITY SERVICES NURSES' COMPONENT

Ministry of Health

- Director, Healthy Living
- Director Health Human Resources (Nursing and Allied Services
- Manager, System Redesign End of Life
- Manager, Maternal and Women's Health
- Manager, Nursing Directorate
- Provincial Coordinator, Nurse Family Partnership
- Manager, Policy & Planning, BCISC

Ministry for Children and Family Development

- Manager, Public Health Nursing
- Area Manager, Youth Forensic Psychiatric Services

*APPENDIX 3 SHORT TERM ILLNESS AND INJURY PLAN AND LONG TERM DISABILITY PLAN

Part I – Short Term Illness and Injury Plan

1.01 Eligibility

(a) Regular employees shall be covered by the Short Term Illness and Injury Plan upon completion of six months of active service with the Employer.

(b) Regular employees with less than six months of service who are unable to work because of illness or injury are entitled to six days coverage at 75% pay in any one calendar year.

(c) Regular employees with three months but less than six months of service will be entitled to 15 weeks (75 work days) of coverage, consisting of the above six days, or what remains of the six days entitlement, at 75% pay, and the remainder of the 15 weeks at two-thirds of pay, not to exceed a maximum weekly benefit of \$413 or the EIC maximum weekly sickness benefit, whichever is higher.

(d) (1) Notwithstanding (a), (b) and (c) above, where a regular employee is on a claim recognized by WorkSafe BC while the employee was on the Employer's business, they shall be entitled, to leave with pay up to 130 days for any one claim in lieu of benefits as outlined in Section 1.02.

(2) Employer and employee contributions and deductions for pension contributions and Employment Insurance during the period of absence will comply with statutory requirements.

(3) During the leave period, the employee will receive net take-home pay equal to wage loss benefits (inclusive of any earnings over and above basic pay) as calculated WorkSafe BC, less any voluntary deductions and those employee deductions referenced in (2) above.

(4) If net take-home pay as calculated in (3) above is less than the employee would receive if they had continued to work, the Employer will top up so there is no difference in net take-home pay.

(5) The compensation payable by WorkSafe BC shall be remitted to the Employer.

(e) Pay for a regular part-time employee under this plan shall be based on their part-time percentage of full-time employment at date of present appointment.

1.02 Short Term Plan Benefit

(a) In the event an employee is unable to work because of illness or injury they will be entitled to a benefit of 75% of pay for a period not to exceed six months from date of absence, (Short Term Plan Period).

(b) The 75% benefit may be supplemented at the rate of 25% of actual duration of absence due to illness or injury by the use of the following in descending order:

- (1) Accumulated sick leave credit under the old sick leave plan;
- (2) Compensatory Time Off (CTO);

(3) Banked Earned Time Off (ETO), excepting where scheduled in a shift schedule;

(4) Vacation entitlement.

1.03 Recurring Disabilities

(a) Employees who return to work after being absent because of illness or injury, and within 15 consecutive scheduled days of work again become unable to work because of the same illness or injury are considered to still be within the original Short Term Plan period as defined in Section 1.02(a).

(b) Employees who return to work after being absent because of illness or injury and within 15 consecutive scheduled work days again become unable to work because of a new illness or injury unrelated to the illness or injury that caused the previous absence shall be entitled to a further six months of benefits under this plan.

(c) Employees who return to work after being absent because of illness or injury, and after working 15 or more consecutive scheduled days of work, again become unable to work because of the same illness or injury will be entitled to a further six month period of benefits under this plan, except as provided in (d) below, where the Short Term Plan period shall continue to be as defined in Section 1.02(a).

(d) Where an employee is returning to work after a period of illness or injury and where the Rehabilitation Committee has approved such return on a trial basis for assessment and/or rehabilitation purposes, the Short Term Plan period shall continue to be as defined in Section 1.02(a). Such trial period must be approved during the period the employee is receiving short term benefits, however, the end of the trial period can go beyond the Short Term Plan benefit period.

(e) Employees who return to work after a period of illness or injury and who do not work the same number of hours that were scheduled prior to the illness or injury shall receive pro-rated benefits under this plan, however, not beyond six calendar months from the initial date of absence as defined in Section 1.02(a), if absence is due to the same illness or injury.

1.04 Doctor's Certificate of Inability to Work

The Employer may require an employee who is unable to work because of illness or injury to provide a statement from:

(a) a medical practitioner qualified to practice in the province of B.C.; or

(b) where necessary, from a medical practitioner licensed to practice in the province of Alberta or the Yukon; or

(c) the consulting physician to whom the employee is referred by the medical practitioner in (a) or (b) above, providing medical evidence of the employee's inability to work in any of the following circumstances:

(1) where it appears that a pattern of consistent or frequent absence from work is developing;

(2) where the employee has been absent for six consecutive scheduled days of work;

(3) where at least 30 days have elapsed since the last statement was obtained and the employee has been in receipt of plan benefits throughout that period.

With the exception of the STO2 and doctor's certificates referenced above, where the Employer requires a medical assessment from the employee's physician specifying the employee's employment limitations and/or capabilities, the employee will be reimbursed, upon production of receipt, for 50% of the cost of the medical assessment.

Benefits will cease to be paid when an employee fails to provide satisfactory evidence of medical disability during the benefit period.

1.05 Integration With Other Disability Income

Short term benefits will be reduced by all other disability income benefits to which the absent employee is entitled except disability income which was being received prior to the illness or injury resulting in the employee being absent from work and which is unrelated to the illness or injury causing the current absence and the ¹/₄ day accumulation that is being used to supplement the plan, pursuant to Section 1.02(b). Other disability income benefits will include:

(a) any amount the absent employee receives from any group insurance, wage continuation or pension plan of the Employer;

(b) any amount of disability income provided by any compulsory act or law, except Employment Insurance sickness benefits and WCB benefits payable in accordance with Section 1.01(d);

(c) any periodic benefit payment from the Canada or Quebec Pension Plan or other social security plan of any country.

Notwithstanding the above, in the case of ICBC Weekly Indemnity payments or, in the case of personal insurance coverage integration will apply to the extent that the combination of Plan benefits and ICBC Weekly Indemnity payments, or personal insurance disability income benefits exceed either:

(1) 100% of pay; or

(2) the applicable benefit percentage of the individual's average total monthly income in the 12 month period immediately preceding commencement of the disability, whichever is the greater. Where this provision is to apply, the employee will be required to provide satisfactory evidence of their total monthly income.

Notwithstanding the above, where an employee makes a successful wage loss claim against a third party for an injury for which the employee received or would

receive STIIP benefits, the Employer will be entitled to recover or decrease Plan benefits by an amount equal to the amount that Plan benefits in combination with the wage loss claim paid exceed 100% of pay.

This section does not apply to a war disability pension paid under an Act of the Governments of Canada or other Commonwealth countries.

1.06 Benefits Not Paid During Certain Periods

Benefits will not be paid when an employee is:

- (a) receiving designated paid holiday pay;
- (b) engaged in an occupation for wage or profit;

(c) on strike or is locked out unless the strike or lockout occurred after the illness or injury resulting in the employee being absent from work;

- (d) serving a prison sentence;
- (e) on suspension without pay;
- (f) on paid absence in the period immediately preceding retirement;
- (g) on any leave of absence without pay.

Notwithstanding (g) above, where an illness or injury occurs during a period of approved:

- (1) educational leave;
- (2) general leave of absence not exceeding 30 days;
- (3) maternity leave, parental leave, or adoption leave

which prevents the employee from returning to work on the scheduled date of return, the Short Term Plan will be effective from the date of disability due to illness or injury and benefits will be paid for the balance of the six month period remaining from the scheduled date of return to work. (h) not actively engaged in a treatment program where the employee's physician determines it to be appropriate to be involved in such a program. An employee shall be afforded the opportunity to demonstrate there were reasonable grounds for not being engaged in a treatment program.

1.07 Employee to Inform Employer

The employee shall inform the Employer as soon as possible of their inability to report to work because of illness or injury. The employee shall inform the Employer of the date of return to duty, in advance of that date, in order that relief scheduled for that employee can be notified.

1.08 Entitlement

For the purpose of calculating six days per calendar year, one day shall be considered to be one day regardless of the regularly scheduled work day. Calculation for part-time employees and partial days will be on a pro-rated basis.

1.09 EIC Premium

The parties agree that the complete premium reduction from the Employment Insurance Commission accruing through the improved illness and injury plan will be returned to the Employer.

1.10 Benefits Upon Layoff or Separation

(a) Subject to (b) and (c) below, regular employees who have completed three months of service and who are receiving benefits pursuant to Section 1.01(c), 1.01(d), or 1.02 shall continue to receive such benefits upon layoff or separation until the termination of the illness or until the maximum benefit entitlement has been granted, whichever comes first, if the notice of layoff or separation is given after the commencement of the illness for which the benefits are being paid.

(b) In the event that layoff or separation notice was given prior to the commencement of the illness, benefits will cease on the effective date of the layoff or separation only if the illness commenced within two months of the effective date of the layoff or separation.

(c) Benefits will continue to be paid in accordance with (a) above for which notice of layoff or separation was given prior to the commencement of the illness

and if the illness commenced more than two months before the effective date of the layoff or separation.

The maximum six month period identified in Appendix 3, Part 1 shall be a maximum seven month period for auxiliary employees who qualify for benefits pursuant to Master Agreement 29.10.

Part II - Long Term Disability Plan

2.01 Eligibility

(a) (1) Regular full-time employees shall be covered by the Long Term Disability Plan upon completion of six months active employment with the Employer. To be covered by the Plan, a regular part-time employee must be working in a position that requires at least half-time work on a regularly scheduled basis, and must have completed six months active service in such a position.

Employees must submit their LTD Plan application within four weeks following the end of the STIIP period. An employee who fails to submit their application for LTD benefits within four weeks of the end of the STIIP period will be presumed to have abandoned their claim for LTD benefits. An employee shall be afforded the opportunity to rebut such presumption to the Plan Administrator and demonstrate that there were reasonable grounds for not having applied for LTD benefits within the prescribed time period.

(2) Where an employee is converted from auxiliary to regular status, plan coverage shall commence the earlier of (a)(1) above, or upon the completion of six months of full-time, unbroken employment from the date the employee qualified for Short Term Illness and Injury Plan benefits under Clause 29.10.

(b) An employee who is not actively at work because of illness or injury on the work day coincident with, or immediately preceding, the date they would otherwise have become eligible for coverage under the Plan will not be eligible for coverage until the date the employee returns to active employment.

(c) Coverage in the plan is a condition of employment.

2.02 Long Term Disability Benefit

In the event an employee, while covered under this plan, becomes totally disabled as a result of an accident or a sickness, then, after the employee has been totally disabled for six months, including periods approved in Sections 1.03(a) and (c), they shall be eligible to receive a monthly benefit as follows:

(a) While the employee has a sick bank balance to be used on a day-for-day basis, full monthly earnings will continue until the sick bank is exhausted, and Section 2.06 will not apply.

(b) When an employee has no sick bank, or after it is exhausted, the employee shall receive a monthly benefit equal to the sum of:

- (1) 70% of the first \$2300 of monthly earnings; and
- (2) 50% of the monthly earnings above \$2300.

For the purposes of the above, earnings shall mean basic monthly earnings as at the date of disability as determined by the Employer.

The basic monthly earnings as at the date of disability shall be the salary in effect for the last month of the Short Term Plan period, or equivalent six month period, taking into consideration any retroactive adjustments. The date of disability for determining the commencement of the first twenty-five months of disability shall be the day following the last month of the Short Term Plan period, or an equivalent six month period.

(c) The Long Term Disability benefit payment will be made as long as an employee remains totally disabled in accordance with Section 2.03, and will cease on the date the employee recovers, or at the end of the month in which the employee reaches age 65, or resigns or dies, whichever occurs first.

(d) An employee in receipt of long term disability benefits will be considered an employee for purposes of pension benefits and will continue to be covered by group life, extended health, dental and medical plans. Employees will not be covered by any other portion of a collective agreement but will retain the right of access to a Rehabilitation Committee established thereunder and will retain seniority rights should they return to employment within nine months following cessation of

benefits. A temporary assignment or auxiliary appointment will not disqualify an employee from the nine month access period.

(e) When an employee is in receipt of the benefit described in (b) above, contributions required for benefit plans in (d) above and contributions for pension will be waived by the Employer.

(f) An employee engaged in rehabilitative employment with the Employer and who is receiving partial Long Term Disability benefit payments will have contributions required for benefit plans in (d) above and contributions for pension waived by the Employer, except that pension contributions shall be deducted from any salary received from the Employer to cover the period of rehabilitative employment.

2.03 Total Disability

(a) Total disability, as used in this Plan, means the complete inability, because of an accident or sickness of a covered employee to perform all the duties of their own occupation for the first 25 months of disability except where accommodation has been made which enables an employee to work:

- (1) in their own occupation, or
- (2) in a job other than their own occupation.

Where accommodation has been made which enables an employee to return to work they will not be considered totally disabled and the rate of pay shall be the rate for the job.

If the rate of pay for this job is less than the rate of pay of the employee at the date of disability, the employee's salary will be protected in accordance with Clause 27.07 at the employee's basic rate at the date of disability.

After the first 25 months of total disability, where accommodation has been made that enables an employee to return to a job other than their own occupation, the employee will not be considered totally disabled and their basic rate shall be the basic rate for the job or 75% of the basic rate of their own occupation, whichever is greater.

After the first 25 months of total disability, employees able by reason of education, training or experience to perform the duties of a gainful occupation for which the rate of pay is not less than 75 per cent of the current rate of pay of their regular occupation at the date of disability will not be considered totally disabled and will therefore not be eligible for benefits under this Long Term Disability Plan.

(b) Total disabilities resulting from mental or nervous disorders are covered by the Plan in the same manner as total disabilities resulting from accidents or other sicknesses, except that an employee who is totally disabled as a result of a mental or nervous disorder and who has received 25 months of Long Term Disability Plan benefit payments must be confined to a hospital or mental institution, or where they are at home, under the direct care and supervision of a medical doctor, in order to continue to be eligible for benefit payments.

During a period of total disability an employee must be under the regular and personal care of a legally qualified doctor of medicine.

(c) (1) If an employee becomes totally disabled and during this period of total disability engages in rehabilitative employment, where they are unable to perform the principal duties of their previous classification, the employee may earn in combination with benefits from this Plan up to 100% of their earnings at the date of disability. In the event that income from rehabilitative employee's earnings at date of disability, the benefit from this Plan will be further reduced by the excess amount.

(2) If an employee is able to perform the principal duties of the position they are placed into on rehabilitative employment, the employee may earn, in combination with benefits from this Plan, up to 100% of their earnings at the date of disability or the position's current rate of pay, whichever is greater.

"Rehabilitative employment" shall mean any occupation or employment for wage or profit or any course or training that entitles the disabled employee to an allowance, provided such rehabilitative employment has the approval of the employee's doctor and the Employer.

The rehabilitative employment of a disabled employee will continue until such time as the employee's earnings from rehabilitative employment exceed 100% of the employee's earnings at the date of disability but in no event for more than 25 months from the date benefit payments commence.

If earnings are received by an employee during a period of total disability and if such earnings are derived from employment which has not been approved of as rehabilitative employment by their doctor and the Employer, then the regular monthly benefit from the Plan will be reduced by 100% of such earnings.

(3) In the event that an employee has been classified as totally disabled for all occupations and engages in approved rehabilitative employment, the provisions of (1) and (2) above apply except that the rehabilitative employment may continue for 25 months from the date rehabilitative employment commenced.

(4) In the case where rehabilitative employment has been approved while an employee is receiving a benefit under the provisions of Section 2.02(a), the provisions of Section 2.03(c)(1) shall not apply until the employee is receiving a benefit under Section 2.02(b).

2.04 Exclusions from Coverage

The Long Term Disability Plan does not cover total disabilities resulting from:

(a) war, insurrection, rebellion, or service in the Armed Forces of any country after the commencement of this plan;

(b) voluntary participation in a riot or civil commotion except while an employee is in the course of performing the duties of their regular occupation;

(c) intentionally self-inflicted injuries or illness;

2.05 **Pre-existing Conditions**

An employee shall not be entitled to Long Term Disability benefits from this Plan if their total disability resulted from an accident, sickness or mental or nervous disorder with respect to which medical treatment, services or supplies were received in the 90 day period prior to the date of hire unless they have completed 12 consecutive months of service after the date of hire during which time they have not been absent from work due to the aforementioned accident, sickness or mental or nervous disorder with respect to which medical treatment, services or supplies were received. This Clause does not apply to present employees who have been continuously employed since April 1,1987.

2.06 Integration With Other Disability Income

In the event a totally disabled employee is entitled to any other income as a result of the same accident, sickness, mental or nervous disorder that caused them to be eligible to receive benefits from this Plan, the benefits from this Plan will be reduced by 100% of such other disability income.

Other disability income shall include, but not necessarily be limited to:

(a) any amount payable under the *Workers' Compensation Act* or Law or any other legislation of similar purpose; and

(b) any amount the disabled employee receives from any group insurance, wage continuation or pension plan of the Employer that provides disability or retirement income; and

(c) any amount of disability income provided by any compulsory act or law; and

(d) any periodic primary benefit payment from the Canada or Quebec Pension Plans or other similar social security plan of any country to which the disabled employee is entitled or to which they would be entitled if their application for such a benefit were approved; and

(e) any amount of disability income provided by any group or association disability plan to which the disabled employee might belong or subscribe.

The amount by which the disability benefit from this Plan is reduced by other disability income will normally be the amount to which the disabled employee is entitled upon becoming first eligible for such other disability income. Future increases in such other disability income resulting from increases in the Canadian Consumer Price Index or similar indexing arrangements will not further reduce the benefit from this Plan.

Notwithstanding the above, in the case of ICBC Weekly Indemnity payments or, in the case of personal insurance coverage, integration will apply to the extent that the

combination of Plan benefits and ICBC Weekly Indemnity payments or, personal insurance disability income benefits exceed either:

(1) 100% of basic pay; or

(2) the applicable benefit percentage of the individual average total monthly income in the 12 month period immediately preceding commencement of the disability, whichever is the greater. Where this provision is to apply the employee will be required to provide satisfactory evidence of their total monthly income.

Notwithstanding the above, where an employee makes a successful wage loss claim against a third party for an injury for which the employee received or would receive LTD benefits, the Employer will be entitled to recover or decrease Plan benefits by an amount equal to the amount that Plan benefits in combination with the wage loss claim paid exceed 100% of pay, subject to the following:

(1) The amount of plan benefit recovered or decreased will be reduced limited to the legal fees attributed to the Employer's share of total claim recovery.

(2) The existence of an action commenced by or on behalf of an employee does not preclude the Employer from joining the employee's action or commencing an action on its own behalf respecting the benefits paid.

(3) Where the Employer or the employee intends to commence or join such an action, they shall advise the other in writing of that intention.

This Section does not apply to a war disability pension paid under an Act of the Governments of Canada or other Commonwealth countries.

2.07 Successive Disabilities

(a) If, following a period of total disability with respect to which benefits are paid from this Plan, an employee returns to work on a full-time basis for a continuous period of six months or more, any subsequent total disability suffered by that employee, whether related to the preceding disability or not, shall be considered a new disability and the disabled employee shall be entitled to benefit payments in accordance with the provisions of this Plan.

(b) In the event the period during which such an employee has returned to work is less than six months and the employee again suffers a total disability and

that is related to the preceding disability, the subsequent disability shall be deemed a continuation of the preceding disability, and the disabled employee shall be entitled to benefit payments in accordance with the provisions of this Plan as though they had not returned to work.

(c) Should such an employee suffer a subsequent disability that is unrelated to the previous disability and, provided the period during which the employee returned to work is longer than one month, the subsequent disability shall be considered a new disability and the employee shall be entitled to benefit payments in accordance with the provisions of this Plan. If the period during which the employee returned to work is one month or less, the subsequent disability shall be deemed a continuation of the preceding disability and the disabled employee shall be entitled to benefit payments in accordance with the provisions of this Plan.

(d) Limitation of benefits for successive disabilities in (b) and (c) above must be determined within one year from the date of absence due to successive disability.

2.08 Cessation of Benefits

An employee shall cease to be eligible for benefits of this Plan at the earliest of the following dates:

(a) at the end of the month in which the employee reaches their 65th birthday (60th birthday for correctional centre employees);

- (b) on the date of commencement of paid absence prior to retirement;
- (c) on the date of termination of employment with the Employer.

Benefits will not be paid when an employee is serving a prison sentence.

Cessation of active employment as a regular employee shall be considered termination of employment except when an employee is on authorized leave of absence with or without pay.

2.09 Leave of Absence

Employees on leave of absence without pay may opt to retain coverage under the plan and shall pay the full premium, except when on approved Maternity Leave. Coverage will be permitted for a period of 18 months of absence without pay except that if the leave is for educational purposes the maximum period will be extended to two years. If an employee on leave of absence without pay or with partial pay, who has elected coverage under this Plan, becomes disabled, benefits under this Plan will be based upon monthly earnings immediately prior to the current leave of absence.

2.10 Benefits Upon Plan Termination

In the event this Long Term Disability Plan is terminated, the benefit payments shall continue to be paid in accordance with the provisions of this Plan to disabled employees who became disabled while covered by this Plan prior to its termination.

2.11 Contributions

The cost of this Plan will be borne by the Employer.

2.12 Waiver of Contributions

Employee contributions to this Plan shall be waived with respect to disabled employees during the time such an employee is in receipt of disability benefit payments from this Plan.

2.13 Claims

(a) Long Term Disability claims will be adjudicated and paid by a claims-paying agent to be appointed by the Employer. In the event a covered employee disputes the decision of the claims-paying agent regarding a claim for benefits under this Plan, the employee may arrange to have their claim reviewed by a Claims Review Committee composed of three medical doctors; one designated by the claimant, one by the Employer, and a third agreed to by the first two. Written notice of a disputed claim or an appeal under this Plan shall be sent to the Plan Administrator.

(b) (1) Written notice of an appeal must be submitted to the Plan Administrator within 60 days from the date the claims-paying agent rejected the claim. Due to extenuating circumstances, the time frame may be extended by the Plan Administrator. (2) Where the claims-paying agent denies benefits due to insufficient medical evidence being provided, an employee will have 60 days in which to provide satisfactory medical evidence to support their claim.

In such circumstances the 60 day appeal period in (1) above will not commence until the claims paying agent renders its decision based on the medical evidence provided.

Where the employee fails to provide further satisfactory medical evidence within the 60 day period, the claim will be deemed to have been denied and the appeal period in (1) above shall commence.

(c) The expenses incurred by a Claims Review Committee will be paid by the Plan.

(d) Where an employee has disputed the decision of the claims-paying agent and is awaiting the outcome of a review or an appeal, the employee will be considered to be on leave of absence without pay during the portion of the waiting period when they are not receiving pay or benefit allowance. During the waiting period an employee will continue to be covered by group life, extended health, dental and medical plans.

(e) LTD benefits received will be reduced by the same amount of benefits received for the same period under the *Employment and Assistance Act* and/or the *Employment and Assistance for Persons with Disabilities Act* (referenced in this section as the "Acts"), except where the benefits received for that period under the *Employment and Assistance Act* and/or the *Employment and Assistance Act* and/or the *Employment and Assistance for Persons with Disabilities Act* (referenced in this section as the "Acts") are repaid to government. Where the employee has been deemed eligible for benefits under these Acts, which benefits exceed the LTD benefits level, LTD benefits will not be subject to reduction for that additional amount.

2.14 Physical Examination

The Employer, at its own expense, shall have the right and be given the opportunity to have a medical doctor appointed by the Employer examine, as often as it may reasonably require, any employee whose injury, sickness, mental or nervous disorder is the basis of claim upon this Plan.

2.15 Canadian Currency

All monies payable to or from this plan shall be payable in Canada in Canadian currency.

2.16 Administration

The Employer will be the administrator of the Plan. All questions arising as to the interpretation of this Plan shall be subject to the grievance and arbitration procedures in Articles 8 and 9 of the Master Agreement.

2.17 Implementation by Regulation

The provisions of this Plan shall become part of a memorandum of agreement between the Parties and will be implemented by regulation.

2.18 Benefit Level

Persons receiving benefits shall receive the same increases to their benefit level as do the employees covered by the terms and conditions of this Collective Agreement receive in wage increases.

Part III - Rehabilitation Committee

(1) It is the intent of both Parties to encourage and facilitate the early return to gainful employment of employees who have been ill or injured. To this end, a Rehabilitation Committee will be established as follows:

(a) The Committee shall consist of five members, two appointed by the Employer, two appointed by the Union and a mutually agreed upon Chairperson. A Secretary shall be appointed to assist in the administration of the Committee.

(b) The Committee shall review cases of regular employees who have completed their initial probationary period and are no longer capable of performing the duties of their own occupation due to illness or injury. Such employees shall make application for rehabilitation pursuant to Appendix 3, Part III - Rehabilitation.

(d) The Committee shall also review cases of all employees who have become incapacitated through industrial injury or illness. Following the review of such cases the Committee, taking into account the best interests of the employee and

the Employer, shall make recommendations to the Deputy Minister of the BC Public Service Agency.

(d) The Committee shall also review cases of regular employees who have completed their initial probationary period who request a transfer on compassionate grounds. Following the review of such cases, the Committee, taking into account the best interests of the employee and the Employer, shall make recommendations to the Deputy Minister of the BC Public Service Agency.

(e) Where the Committee is unable to decide upon recommendations for a particular case, the matter shall be referred to the Bargaining Principals for final disposition.

(f) The Rehabilitation Committee shall meet not less than once a month during working hours, and leave without loss of pay shall be granted to Committee members. Minutes of all meetings shall be taken by the Secretary and copies shall be provided to the Employer and the Union.

(g) Members of the Committee are committed to maintain confidentiality of medical and other information received in their capacity as Committee members.

(2) In the event that a regular employee becomes incapacitated through accident or sickness and they are unable to perform all the duties of their own occupation, the following shall apply:

(a) For the purpose of this Section, incapacity shall mean where the employee is unable to perform all the duties of their own occupation as defined in Section 2.03(a) of the Long Term Disability Plan.

(b) Where the employee meets the definition in (a) above, the Employer shall provide the employee with an application for alternative suitable employment (PSERC7). An employee who fails to:

(1) sign the application form;

(2) make themselves reasonably available and co-operate with a reasonable rehabilitation/return to work process consistent with Rehabilitation Committee Principles;

(3) actively engage in a treatment program where the employee's physician determines it to be appropriate to be involved in such a program shall have benefits suspended.

Prior to having benefits suspended, an employee shall be afforded an opportunity to demonstrate that there were reasonable grounds for failing to meet the above obligations.

(c) The application shall be completed and returned to the Ministry who shall within 10 work days forward the application to the Secretary. The Committee members shall be provided with copies of the application.

(d) The Rehabilitation Committee will, based on the information, coordinate the necessary medical and/or vocational assessments and determine the following:

(1) if the application is properly before the Committee;

(2) based on the assessment, determine whether the employee is immediately capable of performing modified, alternative or rehabilitative employment;

(3) if no to (2) above the Committee may, based on the assessments, implement the necessary training to place the employee in alternative or rehabilitative employment;

(4) In considering modified, alternative or rehabilitative employment, the Committee may provide advice and make recommendations to the Ministry to return the incapacitated employee to work considering the following accommodations:

- (i) modification of the duties of the employee's job;
- (ii) flexibility in scheduling hours of work within existing hours of operation;
- (iii) provision of technical or mechanical aids.

(5) where the employee is considered capable of performing alternative employment or once the rehabilitative employment is considered to be successful, and the employee is therefore able to perform the duties of a gainful occupation, they shall be subject to Article 13 of the Master Agreement excluding displacement options pursuant to Clause 13.09. (e) (1) An employee in receipt of STIIP benefits, whose prognosis for return to work exceeds eight weeks, may be referred to the Rehabilitation Committee if the Occupational Health and Rehabilitation determines it is medically appropriate to do so.

(2) In those cases where a return to their own occupation is unlikely, employees may be referred, by either Party to the Rehabilitation Committee while on STIIP. In such cases, Part III (c), and (d) will apply.

(f) Where an employee has a physical occupational illness or injury, the Ministry will, where feasible, accommodate the employee's incapacity so as to avoid a time loss illness or injury. Where a time loss illness or injury occurs, the compensation payable shall be in accordance with the applicable terms of Appendix 3.

(g) Where the Ministry has concerns with a recommendation made in accordance with (d)(4) above, the concern will be reviewed with the Rehabilitation Committee.

*APPENDIX 4 MINISTRIES, BOARDS AND AGENCIES

This Agreement applies to all employees within the Nurses bargaining unit who are employed in all ministries of the Government of the Province of British Columbia.

It also applies to all employees within the Nurses bargaining unit employed in other boards, agencies or commissions which **may be** designated by Order-in-Council pursuant to Section 3 of the *Public Service Act* or whose enabling legislation provides for the appointment of employees pursuant to the *Public Service Act*.

*APPENDIX 5 SIXTEENTH NURSES MASTER WAGE SCHEDULES

1. The classification, grid level and corresponding Grade Descriptor are as follows:

Classification	Grid Level	Grade Descriptors				
4	4	Direct Care Giver				
		Program Officer				
5	5	First Line Supervisor				
7	7	Case Manager				
		Nurse Clinician				
		Supervisor (small scope)				
		Program Coordinator				
<u>8</u>	8	Assistant Team Lead				
9	9	Supervisor (large scope)				
		Supervisor (sole charge)				
		Senior Program Coordinator				
		Senior Nursing Authority in a				
		Facility				

*APPENDIX 5 EFFECTIVE APRIL 5, 2015 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

2. Effective **April 5, 2015** there will be a **1% increase** to all grids and steps on the salary schedule.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	55,547.77	4,628.98	2,129.14	30.4163
		~		·	
	1	56,251.40	4,687.62	2,156.11	30.8016
	2	60,577.79	5,048.15	2,321.94	33.1706
4	3	62,757.03	5,229.75	2,405.47	34.3639
	4	67,068.29	5,589.02	2,570.72	36.7246
	5	69,095.16	5,757.93	2,648.41	37.8344
	6	73,832.20	6,152.68	2,829.98	40.4283
	1	59,568.92	4,964.08	2,283.27	32.6181
	2	64,151.50	5,345.96	2,458.92	35.1274
5	3	66,458.32	5,538.19	2,547.34	36.3906
	4	71,024.21	5,918.68	2,722.35	38.8907
	5	73,170.57	6,097.55	2,804.62	40.0660
	6	78,186.76	6,515.56	2,996.89	42.8127
	1	63,082.88	5,256.91	2,417.96	34.5423
	2	67,935.23	5,661.27	2,603.95	37.1993
7	3	70,377.97	5,864.83	2,697.58	38.5369
	4	75,213.63	6,267.80	2,882.93	41.1847
	5	77,487.05	6,457.25	2,970.07	42.4296
	6	82,798.83	6,899.90	3,173.67	45.3381
	1	64,943.05	5,411.92	2,489.26	35.5609
	2	69,938.89	5,828.24	2,680.75	38.2964
8	3	72,453.90	6,037.83	2,777.15	39.6736
	4	77,432.26	6,452.69	2,967.97	42.3996
	5	79,771.95	6,647.66	3,057.65	43.6807
	6	85,241.05	7,103.42	3,267.28	46.6754

	1	66,803.48	5,566.96	2,560.57	36.5796
	2	71,942.03	5,995.17	2,757.53	39.3933
9	3	74,529.56	6,210.80	2,856.71	40.8101
	4	79,650.37	6,637.53	3,052.99	43.6141
	5	82,057.63	6,838.14	3,145.26	44.9323
	6	87,683.27	7,306.94	3,360.89	48.0127

*APPENDIX 5 EFFECTIVE FEBRUARY 7, 2016 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

3. Effective February 7, 2016 there will be a 0% increase to all grids and steps of the salary schedule and an Economic Stability Dividend equal to 0.45%.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	55,797.71	4,649.81	2,138.72	30.5531
	1	56,504.47	4,708.71	2,165.81	30.9401
	2	60,850.42	5,070.87	2,332.39	33.3199
4	3	63,039.31	5,253.28	2,416.29	34.5184
	4	67,370.14	5,614.18	2,582.29	36.8899
	5	69,406.15	5,783.85	2,660.33	38.0047
	6	74,164.31	6,180.36	2,842.71	40.6101
	1	59,836.85	4,986.40	2,293.54	32.7649
	2	64,440.31	5,370.03	2,469.99	35.2856
5	3	66,757.30	5,563.11	2,558.80	36.5543
	4	71,343.80	5,945.32	2,734.60	39.0657
	5	73,499.82	6,124.99	2,817.24	40.2463
	6	78,538.71	6,544.89	3,010.38	43.0054
	1	63,366.74	5,280.56	2,428.84	34.6977
	2	68,241.00	5,686.75	2,615.67	37.3667
7	3	70,694.70	5,891.23	2,709.72	38.7103
	4	75,552.00	6,296.00	2,895.90	41.3700
	5	77,835.86	6,486.32	2,983.44	42.6206
	6	83,171.38	6,930.95	3,187.95	45.5421
	4				

	1	65,235.25	5,436.27	2,500.46	35.7209
	2	70,253.53	5,854.46	2,692.81	38.4687
8	3	72,780.02	6,065.00	2,789.65	39.8521
	4	77,780.81	6,481.73	2,981.33	42.5904
	5	80,130.94	6,677.58	3,071.41	43.8773
	6	85,624.56	7,135.38	3,281.98	46.8854

	1	67,104.03	5,592.00	2,572.09	36.7441
	2	72,265.80	6,022.15	2,769.94	39.5706
9	3	74,865.07	6,238.76	2,869.57	40.9939
	4	80,008.84	6,667.40	3,066.73	43.8104
	5	82,426.80	6,868.90	3,159.41	45.1344
	6	88,077.74	7,339.81	3,376.01	48.2287

*APPENDIX 5 EFFECTIVE APRIL 3, 2016 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

4. Effective **April 3, 2016** there will be a **0.5%** increase to all grids and steps on the salary schedule.

Grid Level Step Annual Monthly **Bi-weekly** Hourly **Interim Permit** 56,076.60 4,673.05 2,149.41 30.7059 1 56,787.01 4,732.25 2,176.64 31.0949 1 2 33.4864 61,154.62 2,344.05 5,096.22 4 3 63.354.47 5,279.54 2,428.37 34.6910 4 67,706.95 5,642.25 2,595.20 37.0743 5 69,753.14 5,812.76 2,673.63 38.1947 6 74,535.04 6,211.25 2,856.92 40.8131 60,136.10 5.011.34 2,305.01 1 32.9287 2 64,762.51 5,396.88 2,482.34 35.4620 5 3 67,090.98 5,590.92 2,571.59 36.7370 4 71,700.44 5,975.04 2,748.27 39.2610 5 73,867.42 6,155.62 2,831.33 40.4476 6 78,931.35 6,577.61 3,025.43 43.2204 63,683.46 5,306.96 34.8711 1 2,440.98 2 5,715.19 68,582.25 2,628.75 37.5536 3 7 71.048.21 5,920.68 2,723.27 38.9039 2,910.38 4 75,929.78 6,327.48 41.5769 5 78,225.11 6,518.76 2,998.36 42.8337 3,203.89 45.7699 6 83,587.25 6,965.60

	1	65,561.37	5,463.45	2,512.96	35.8994
	2	70,604.69	5,883.72	2,706.27	38.6610
8	3	73,143.96	6,095.33	2,803.60	40.0514
	4	78,169.80	6,514.15	2,996.24	42.8034
	5	80,531.67	6,710.97	3,086.77	44.0967
	6	86,052.69	7,171.06	3,298.39	47.1199

	1	67,439.54	5,619.96	2,584.95	36.9279
	2	72,627.13	6,052.26	2,783.79	39.7684
9	3	75,239.45	6,269.95	2,883.92	41.1989
	4	80,408.79	6,700.73	3,082.06	44.0294
	5	82,839.01	6,903.25	3,175.21	45.3601
	6	88,518.13	7,376.51	3,392.89	48.4699

*APPENDIX 5 EFFECTIVE FEBRUARY 5, 2017 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

5. Effective February 5, 2017, there will be a 1% increase to all grids and steps on the salary schedule and an Economic Stability Dividend equal to 0.35%.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	56,833.71	4,736.14	2,178.43	31.1204
		,			
	1	57,553.52	4,796.13	2,206.02	31.5146
	2	61,980.09	5,165.01	2,375.69	33.9384
4	3	64,209.68	5,350.81	2,461.15	35.1593
	4	68,621.12	5,718.43	2,630.24	37.5749
	5	70,694.70	5,891.23	2,709.72	38.7103
	6	75,541.31	6,295.11	2,895.49	41.3641
		•	. .		
	1	60,948.00	5,079.00	2,336.13	33.3733
	2	65,636.77	5,469.73	2,515.85	35.9407
5	3	67,996.80	5,666.40	2,606.31	37.2330
	4	72,668.35	6,055.70	2,785.37	39.7910
	5	74,864.55	6,238.71	2,869.55	40.9936
	6	79,996.84	6,666.40	3,066.27	43.8039
		•	. .		
	1	64,543.10	5,378.59	2,473.93	35.3419
	2	69,508.16	5,792.35	2,664.24	38.0606
7	3	72,007.25	6,000.60	2,760.03	39.4290
	4	76,954.83	6,412.90	2,949.67	42.1381
	5	79,281.21	6,606.77	3,038.84	43.4120
	6	84,715.61	7,059.63	3,247.14	46.3877
					•
	1	66,446.32	5,537.19	2,546.88	36.3840
	2	71,557.73	5,963.14	2,742.80	39.1829
8	3	74,131.44	6,177.62	2,841.45	40.5921
	4	79,225.12	6,602.09	3,036.69	43.3813
	5	81,618.81	6,801.57	3,128.44	44.6920
	6	87,214.44	7,267.87	3,342.92	47.7560

	1	68,350.05	5,695.84	2,619.85	37.4264
	2	73,607.57	6,133.96	2,821.37	40.3053
9	3	76,255.11	6,354.59	2,922.85	41.7550
	4	81,494.36	6,791.20	3,123.67	44.6239
	5	83,957.45	6,996.45	3,218.08	45.9726
	6	89,713.02	7,476.09	3,438.69	49.1241

*APPENDIX 5 EFFECTIVE APRIL 2, 2017 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

6. Effective **April 2, 2017** there will be a **0.5%** increase to all grids and steps on the salary schedule.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit			4,759.82	2,189.32	31.2760
Internin Permit	1	57,117.83	4,/59.02	2,189.52	51.2700
	1	57,841.28	4,820.11	2 217 05	31.6721
		/		2,217.05	
	2	62,290.03	5,190.84	2,387.57	34.1081
4	3	64,530.84	5,377.57	2,473.46	35.3351
	4	68,964.19	5,747.02	2,643.39	37.7627
	5	71,048.21	5,920.68	2,723.27	38.9039
	6	75,919.08	6,326.59	2,909.97	41.5710
	1	1	1		
	1	61,252.72	5,104.39	2,347.81	33.5401
	2	65,964.97	5,497.08	2,528.43	36.1204
5	3	68,336.75	5,694.73	2,619.34	37.4191
	4	73,031.78	6,085.98	2,799.30	39.9900
	5	75,238.93	6,269.91	2,883.90	41.1986
	6	80,396.79	6,699.73	3,081.60	44.0229
	_		-)		
	1	64,865.83	5,405.49	2,486.30	35.5186
	2	69,855.67	5,821.31	2,677.56	38.2509
7	3	72,367.28	6,030.61	2,773.83	39.6261
	4	77,339.64	6,444.97	2,964.42	42.3489
	5	79,677.50	6,639.79	3,054.03	43.6290
	6	85,139.30	7,094.94	3,263.38	46.6197
	·	· ·	· · · · ·	· · ·	
	1	66,778.43	5,564.87	2,559.61	36.5659
	2	71,915.42	5,992.95	2,756.51	39.3787
8	3	74,502.17	6,208.51	2,855.66	40.7951
	4	79,621.15	6,635.10	3,051.87	43.5981
	5	82,026.85	6,835.57	3,144.08	44.9154
	6	87,650.39	7,304.20	3,359.63	47.9947

	1	68,691.82	5,724.32	2,632.95	37.6136
	2	73,975.69	6,164.64	2,835.48	40.5069
9	3	76,636.28	6,386.36	2,937.46	41.9637
	4	81,901.88	6,825.16	3,139.29	44.8470
	5	84,377.23	7,031.44	3,234.17	46.2024
	6	90,161.49	7,513.46	3,455.88	49.3697

*APPENDIX 5 EFFECTIVE SEPTEMBER 3, 2017 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

7. Effective **September 3, 2017,** the Salary Schedule will be based on **nine steps** with **12-month** intervals.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	57,117.83	4,759.82	2,189.32	31.2760
	1	57,841.28	4,820.11	2,217.05	31.6721
	2	59,865.81	4,988.82	2,294.65	32.7807
	3	61,961.04	5,163.42	2,374.96	33.9280
	4	64,129.59	5,344.13	2,458.08	35.1154
4	5	66,374.05	5,531.17	2,544.11	36.3444
	6	68,697.04	5,724.75	2,633.15	37.6164
	7	71,101.43	5,925.12	2,725.31	38.9330
	8	73,590.09	6,132.51	2,820.70	40.2957
	9	75,919.08	6,326.59	2,909.97	41.5710
			-		
	1	61,252.72	5,104.39	2,347.81	33.5401
	2	63,396.48	5,283.04	2,429.98	34.7140
	3	65,615.37	5,467.95	2,515.03	35.9290
	4	67,912.01	5,659.33	2,603.06	37.1866
5	5	70,289.01	5,857.42	2,694.17	38.4881
	6	72,749.23	6,062.44	2,788.47	39.8353
	7	75,295.55	6,274.63	2,886.07	41.2296
	8	77,930.83	6,494.24	2,987.08	42.6726
	9	80,396.79	6,699.73	3,081.60	44.0229
		-	-	-	
	1	64,865.83	5,405.49	2,486.30	35.5186
	2	67,136.12	5,594.68	2,573.32	36.7617
	3	69,485.98	5,790.50	2,663.39	38.0484
	4	71,918.03	5,993.17	2,756.61	39.3801
7	5	74,435.12	6,202.93	2,853.09	40.7584
	6	77,040.40	6,420.03	2,952.95	42.1850
	7	79,736.73	6,644.73	3,056.30	43.6614
	8	82,527.50	6,877.29	3,163.27	45.1896
	9	85,139.30	7,094.94	3,263.38	46.6197

	1	66,778.43	5,564.87	2,559.61	36.5659
	2	69,115.77	5,759.65	2,649.20	37.8457
	3	71,534.77	5,961.23	2,741.92	39.1703
	4	74,038.56	6,169.88	2,837.89	40.5413
8	5	76,630.01	6,385.83	2,937.22	41.9603
	6	79,311.99	6,609.33	3,040.02	43.4289
	7	82,087.90	6,840.66	3,146.42	44.9489
	8	84,960.85	7,080.07	3,256.54	46.5220
	9	87,650.39	7,304.20	3,359.63	47.9947
	1	68,691.82	5,724.32	2,632.95	37.6136
	2	71,095.95	5,924.66	2,725.10	38.9300
	3	73,584.35	6,132.03	2,820.48	40.2926
	4	76,159.88	6,346.66	2,919.20	41.7029
9	5	78,825.43	6,568.79	3,021.37	43.1624
	6	81,584.37	6,798.70	3,127.12	44.6731
	7	84,439.85	7,036.65	3,236.57	46.2367
	8	87,395.24	7,282.94	3,349.85	47.8550
	9	90,161.49	7,513.46	3,455.88	49.3697

*APPENDIX 5 EFFECTIVE FEBRUARY 4, 2018 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

8. Effective **February 4, 2018** there will be a **1%** increase to all grids and steps on the salary schedule **and an Economic Stability Dividend of 0.4%**.

The Salary Schedule will be based on nine steps with 12-month intervals.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly	
Interim Permit	1	57,917.46	4,826.46	2,219.97	31.7139	
	1	58,651.09	4,887.59	2,248.09	32.1156	
	2	60,704.06	5,058.67	2,326.78	33.2397	
	3	62,828.51	5,235.71	2,408.21	34.4030	
	4	65,027.32	5,418.94	2,492.49	35.6070	
4	5	67,303.35	5,608.61	2,579.73	36.8533	
	6	69,658.69	5,804.89	2.670.01	38.1430	
	7	72,096.74	6,008.06	2,763.46	39.4780	
	8	74,620.35	6,218.36	2,860.19	40.8599	
	9	76,981.96	6,415.16	2,950.71	42.1530	
	1	62.110.27	5,175.86	2,380.68	34.0097	
	2	64.284.04	5,357.00	2,464.00	35.2000	
	3	66.533.98	5,544.50	2,550.24	36.4320	
	4	68.862.71	5.738.56	2,639.50	37.7071	
5	5	71.273.10	5.939.43	2.731.89	39.0270	
	6	73.767.76	6.147.31	2.827.51	40.3930	
	7	76.349.55	6,362.46	2,926.47	41.8067	
	8	79.021.88	6,585.16	3,028.90	43.2700	
	9	81,522.28	6,793.52	3,124.74	44.6391	
	1	65.774.00	5,481.17	2,521.11	36.0159	
	2	68,076.11	5,673.01	2.609.35	37.2764	
	3	70,458.85	5,871.57	2,700.68	38.5811	
7	4	72,924.81	6,077.07	2.795.20	39.9314	
	5	75,477.13	6,289.76	2,893.03	41.3290	
	6	78,118.93	6.509.91	2.994.29	42.7756	
	7	80,853.09	6.737.76	3.099.09	44.2727	
	8	83.683.00	6,973.58	3.207.56	45.8223	
	9	86,331.32	7,194.28	3,309.07	47.2724	

	1	67,713.21	5,642.77	2.595.44	37.0777
	2	70,083.43	5.840.29	2.686.29	38.3756
	3	72,536.34	6.044.70	2.780.31	39.7187
	4	75,075.09	6.256.26	2.877.62	41.1089
8	5	77,702.81	6,475.23	2.978.34	42.5477
	6	80,422.35	6.701.86	3,082.58	44.0369
	7	83,237.13	6,936.43	3.190.47	45.5781
	8	86,150.26	7.179.19	3.302.13	47.1733
	9	88,877.37	7.406.45	3.406.66	48.6666
	1	69.653.47	5,804.46	2,669.81	38.1401
	2	72.091.26	6.007.61	2.763.25	39.4750
	3	74.614.62	6,217.89	2.859.97	40.8567
	4	77.226.15	6,435.51	2.960.07	42.2867
9	5	79.929.01	6.660.75	3.063.67	43.7667
	6	82.726.56	6.893.88	3.170.90	45.2986
	7	85.621.95	7.135.16	3.281.88	46.8840
	8	88.618.83	7.384.90	3.396.75	48.5250
	9	91.423.69	7.618.64	3,504.26	50.0609

*APPENDIX 5 EFFECTIVE APRIL 1, 2018 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

9. Effective **April 1, 2018** there will be a **0.5%** increase to all grids and steps on the salary schedule.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	58,207.05	4,850.59	2,231.07	31.8724
	1	58,944.34	4,912.03	2,259.33	32.2761
	2	61,007.48	5,083.96	2,338.41	33.4059
	3	63,142.63	5,261.89	2,420.25	34.5750
	4	65,352.39	5,446.03	2,504.95	35.7850
4	5	67,639.90	5,636.66	2,592.63	37.0376
	6	70,006.98	5,833.92	2,683.36	38.3337
	7	72,457.29	6,038.11	2,777.28	39.6754
	8	74,993.43	6,249.45	2,874.49	41.0641
	9	77,366.78	6,447.23	2,965.46	42.3637
	1	62,420.74	5,201.73	2,392.58	34.1797
	2	64,605.46	5,383.79	2,476.32	35.3760
	3	66,866.62	5,572.22	2,562.99	36.6141
	4	69,207.09	5,767.26	2,652.70	37.8957
5	5	71,629.48	5,969.12	2.745.55	39.2221
	6	74,136.66	6,178.06	2,841.65	40.5950
	7	76,731.24	6,394.27	2,941.10	42.0157
	8	79,416.87	6.618.07	3,044.04	43.4863
	9	81,929.79	6,827.48	3,140.36	44.8623
	· · · · · ·	1		11	
	1	66,102.98	5,508.58	2,533.72	36.1960
	2	68,416.58	5,701.38	2,622.40	37.4629
	3	70,811.06	5,900.92	2,714.18	38.7740
	4	73,289.54	6,107.46	2,809.18	40.1311
7	5	75,854.64	6,321.22	2,907.50	41.5357
	6	78,509.49	6,542.46	3,009.26	42.9894
	7	81,257.47	6,771.46	3,114.59	44.4941
	8	84,101.47	7,008.46	3.223.60	46.0514
	9	86,763.10	7,230.26	3,325.62	47.5089

	1	68,051.85	5,670.99	2,608.42	37.2631		
	2	70,433.80	5,869.48	2,699.72	38.5674		
8	3	72,898.98	6,074.92	2,794.21	39.9173		
	4	75,450.52	6,287.54	2,892.01	41.3144		
	5	78,091.28	6,507.61	2,993.23	42.7604		
	6	80,824.39	6,735.37	3,097.99	44.2570		
	7	83,653.25	6,971.10	3,206.42	45.8060		
	8	86,580.99	7,215.08	3.318.64	47.4091		
	9	89,321.68	7,443.47	3,423.69	48.9099		
	1	70,001.77	5,833.48	2,683.16	38.3309		
	2	72,451.81	6,037.65	2,777.07	39.6724		
	3	74,987.69	6,248.97	2,874.27	41.0610		
	4	77,612.28	6.467.69	2.974.87	42.4981		
9	5	80,328.69	6,694.06	3.078.99	43.9856		
	6	83,140.08	6,928.34	3.186.75	45.5250		
	7	86.050.08	7,170.84	3.298.29	47.1184		
	8	89,061.83	7,421.82	3,413.73	48.7676		
	9	91,880.77	7,656.73	3,521.78	50.3111		

*APPENDIX 5 EFFECTIVE FEBRUARY 3, 2019 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

10. Effective **February 3, 2019** there will be a **1%** increase to all grids and steps on the salary schedule **and an Economic Stability Dividend of 0.75%**.

The Salary Schedule will be based on nine steps with 12-month intervals.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	59,225.58	4,935.47	2,270.11	32.4301
	1	59,975.91	4,997.99	2,298.87	32.8410
	2	62,075.05	5,172.92	2,379.33	33.9904
	3	64,247.51	5,353.96	2,462.60	35.1800
	4	66,496.15	5,541.35	2,548.79	36.4113
4	5	68,823.57	5,735.30	2,638.00	37.6857
	6	71,232.14	5,936.01	2,730.32	39.0046
	7	73,725.23	6,143.77	2,825.88	40.3697
	8	76,305.72	6,358.81	2,924.79	41.7827
	9	78,720.81	6,560.07	3,017.36	43.1051
	1	63,513.10	5,292.76	2,434.45	34.7779
	2	65,736.17	5,478.01	2,519.66	35.9951
	3	68,036.72	5,669.73	2,607.84	37.2549
	4	70,418.15	5,868.18	2,699.12	38.5589
5	5	72,883.07	6,073.59	2,793.60	39.9086
	6	75,434.08	6,286.17	2,891.38	41.3054
	7	78,074.06	6,506.17	2,992.57	42.7510
	8	80,806.65	6,733.89	3,097.31	44.2473
	9	83,363.66	6,946.97	3,195.32	45.6474
	1	67,259.78	5,604.98	2,578.06	36.8294
	2	69,613.82	5,801.15	2,668.29	38.1184
7	3	72,050.30	6,004.19	2,761.68	39.4526
	4	74,572.09	6,214.34	2,858.34	40.8334
	5	77,182.06	6,431.84	2,958.38	42.2626
	6	79,883.35	6,656.95	3,061.92	43.7417
	7	82,679.60	6,889.97	3,169.10	45.2729
	8	85,573.16	7,131.10	3,280.01	46.8573
	9	88,281.50	7,356.79	3,383.82	48.3403

	1	69,242.83	5,770.24	2,654.07	37.9153
	2	71,666.52	5,972.21	2,746.97	39.2424
	3	74,174.75	6,181.23	2,843.11	40.6159
	4	76,770.90	6,397.58	2,942.62	42.0374
8	5	79,457.83	6,621.49	3,045.61	43.5087
	6	82,238.69	6,853.22	3,152.20	45.0314
	7	85,117.12	7,093.09	3,262.53	46.6076
	8	88,096.26	7,341.36	3,376.72	48.2389
	9	90,884.69	7,573.72	3,483.60	49.7657
	1	71,226.92	5,935.58	2,730.12	39.0017
	2	73,719.75	6,143.31	2,825.67	40.3667
	3	76,299.98	6,358.33	2,924.57	41.7796
	4	78,970.48	6,580.87	3,026.93	43.2419
9	5	81,734.39	6,811.20	3,132.87	44.7553
	6	84,595.08	7,049.59	3,242.52	46.3217
	7	87,555.95	7,296.33	3,356.01	47.9430
	8	90,620.40	7,551.70	3,473.47	49.6210
	9	93,488.66	7,790.72	3,583.41	51.1916

APPENDIX 6 ADDRESSING WORKPLACE VIOLENCE

The parties recognize that it is important to provide an environment that is properly secure for all those who receive health services or who work in health care. A safe environment is important for staff and contributes to providing the highest possible standard of care. Staff should expect to work in, and patients should expect to be treated in, an environment where the risk of violence is minimized.

Violence Prevention Programs

The Provincial Joint Occupational Health and Safety Committee (PJOSH) will review and make recommendations on violence prevention programs. In order to do so the Committee will:

- (i) Receive information on the scope and nature of violence in the workplace from the local OH&S Committees;
- (ii) Receive information on violence prevention activities from the local OH&S Committees;
- (iii) Review existing best practices violence prevention education modules within the Public Service and those utilized in the Health Authorities with the intent of developing standard education principles;
- (iv) Receive ongoing worksite risk assessments coordinated by the local OH&S Committees and reported to the PJOSH;

Based upon the information provided, PJOSH shall develop and make recommendations to minimize the danger of workplace violence to employees, which shall include appropriate training, processes and a system for the ongoing reporting and monitoring of incidents and situations involving violence or the risk of violence. Such recommendations shall be made to the local OH&S Committees as well as the bargaining principals.

*INFORMATION APPENDIX A APPOINTMENT TO THE PUBLIC SERVICE

The following is pursuant to the *Public Service Act* and is included for information purposes only and does not form part of a collective agreement.

(a) <u>Appointments on Merit</u>

Subject to section 10 of the *Public Service Act* appointments to and from within the public service must be based on the principle of merit.

(b) <u>Determination of Merit</u>

The matters to be considered in determining merit shall, having regard to the nature of the duties to be performed, include the applicant's education, skills, knowledge, experience, past work performance and years of continuous service in the public service.

INFORMATION APPENDIX B ADVANCE PAYMENT OF GROUP LIFE BENEFITS

The guidelines regarding payment of group life benefits for terminally ill employees pursuant to Clause 25.04 are as follows:

- 1. Death must be "expected" within 24 months. The employee's attending physician will be required to provide sufficient medical information, including the employee's diagnosis and prognosis, to allow the group life insurance carrier to assess the life expectancy.
- 2. Requests for advance payments must be in writing.
- 3. Authorization from the employer must be submitted with the employee's request.
- 4. The amount of the payment will be 50% of the life insurance coverage, subject to a maximum of \$50,000.
- 5. A signed release will be obtained from the insured employee prior to payment being made. A release is not required from designated revocable beneficiaries as they have no legal rights to life insurance proceeds until after the insured's death. Situations involving irrevocable beneficiaries or divorce judgments will require special releases.

*INFORMATION APPENDIX C JOB SHARING

This **Information Appendix** applies to all regular employees who have been appointed under Section 8 of the *Public Service Act* and:

- outlines the circumstances under which job sharing arrangements may occur;
- outlines the terms and conditions of job sharing; and
- provides guidelines for the review of job sharing proposals and the evaluation of current job sharing arrangements.

DEFINITIONS

"Job sharing proposal" a document, initiated by two employees, which outlines their request to become part-time employees, and recommends how the duties of a position previously performed by one full-time employee, can be divided to accommodate their request.

"Job sharing arrangement" where two part-time employees perform the duties of a position previously performed by one full-time employee.

"Partners" part-time employees participating in a job sharing arrangement.

Job Sharing Proposals

Job sharing proposals can be considered where:

- one of the partners proposing the job sharing arrangement already occupies the full-time position under consideration, OR
- two partners propose to job share a vacant position which is at a classification level that is the same or lower than the partners' current position.

A job sharing proposal must be presented to an excluded manager for consideration. Job sharing proposals must include details as outlined in the mandatory procedures section of this **document**.

Approval of the job sharing proposal is at the discretion of the excluded manager.

Eligible Partners

The recommended partner(s) outlined in the job sharing proposal must be:

- qualified for the position to be shared;
- employed under the *Public Service Act* as a regular employee;
- at the same classification level or higher than the position being shared;
- performing their current duties satisfactorily.

Appointment of Job Sharing Partners

The approval of a job sharing proposal is confirmed in writing by appointing the job sharing partners as part-time employees. Appointments are subject to the **terms of the** *Public Service Act* and collective agreement language.

The appointment letter should address whether or not the employee's hours may be increased up to full-time due to operational requirements.

Acceptance of the appointment must be in writing.

Benefits

Benefits granted job sharing partners are in accordance with those approved for parttime employees. Most benefits are prorated based on the number of hours the employee works, except for the following benefits which are paid in full to both partners: basic medical insurance (MSP), extended health care plan, dental plan and air travel insurance. Each employee is also eligible for the minimum group life insurance.

Extended Absence

Where stated in the appointment letter, the supervisor may, due to operational requirements, increase one partner's work hours up to full-time to cover the other's extended absence, (e.g. leave or resignation). This is not meant to be a permanent change in hours of work unless requested by the employee and approved by the excluded manager; nor is it meant to limit the excluded manager's responsibility to determine how operational requirements will be met on each occasion.

The supervisor will give as much notice as possible to the partner before increasing a partner's hours of work.

Termination of Job Sharing Arrangement by Employees

Upon termination of the job sharing arrangement by either partner, the remaining partner may request to fill the position on a full-time basis or may submit a new job sharing proposal.

The Ministry will endeavour to find alternative employment for the job sharing partners if either wishes to terminate the agreement; however, the onus is on the employee to seek alternative employment if they no longer wish to job share.

Termination of Job Sharing Arrangement by Employer

Ministries may terminate a job sharing arrangement with reference being given to relevant provisions of the collective agreement. Such action should be limited to bona fide operational reasons.

Filling of Vacated Job Shared Position

It is at the discretion of the excluded manager, in cases where both partners leave a job sharing arrangement, to decide on how the position will be filled, e.g. approve a subsequent job sharing proposal, fill the position on a full-time basis.

The vacancy created by one partner leaving may be filled by approving the remaining partner's request for full-time employment; by approving a new job sharing proposal; or by posting the part-time position.

Responsibilities

Deputy Ministers are authorized to:

- determine whether job sharing arrangements are feasible;
- consider and approve or reject job sharing proposals;
- delegate in writing the above responsibility to other individuals within the Ministry.

Accountabilities

Deputy Ministers are accountable for ensuring that: a mechanism is in place to review and respond to job sharing proposals.

Mandatory Procedures

Job sharing proposals must include:

- a written statement signed by both partners requesting part-time employment in order to job share as outlined in the proposal;
- information on the qualifications and experience of the proposed partner(s);
- a copy of the proposed partner(s) most recent performance appraisal(s);
- a description of how job duties and responsibilities may be shared;
- details on what arrangements the partners will make to share necessary information with each other, with clients, with colleagues and with the supervisor;
- a proposal of how workload priorities will be determined by the partners on an ongoing basis;
- preferred start date;
- preferred work schedule.

The appointment letter should outline the terms and conditions of employment, and state the agreed to terms of the specific job sharing arrangement. If the Ministry intends on increasing either partner's hours of work, due to operational requirements, it must be so indicated in their appointment letter.

*INFORMATION APPENDIX D DEFERRED SALARY LEAVE PROGRAM

FOREWORD

The objective of a Deferred Salary Leave Program is to provide employees an opportunity to self-finance a planned period of absence to pursue any activity or interest. Please refer to MyHR for details on the current process and resources:

http://www2.gov.bc.ca/gov/content/careers-myhr/all-employees/leave-time-off/deferred-salary-leave-program.

This **program** applies to all regular employees appointed pursuant to sections 8 and 12 of the *Public Service Act (PSA)* with the exception of salaried physicians and to Order in Council appointments made under Section 15 of the *PSA* as specified in Category A and B of prerogative Order in Council 806/89.

DEFINITIONS

"deferred salary leave"

is defined as a period of authorized leave without pay, for a period of time not shorter than six consecutive months or longer than one year in duration, where an employee has requested such leave in advance and at the time of the request makes an arrangement with the **m**inistry to deposit a predetermined portion of salary into a trust fund which will provide an income for the employee during the leave period.

"accrued interest"

is the amount of interest earned on the monies deposited with the financial institution on behalf of the participant.

"deferral period"

is the length of time between 12 and 72 continuous months that an employee participates in the program prior to commencing the leave period.

"deferred salary"

is the portion of current salary authorized by the employee to be deducted and retained by the financial institution on behalf of the employee.

"financial institution"

is any Canadian chartered bank, trust company or credit union, authorized to carry on business in the Province of British Columbia.

"leave period"

is the period of time between 6 and 12 whole consecutive months when the participant is on a leave of absence without pay and drawing their deferred funds from the financial institution.

PURPOSE

The Deferred Salary Leave Program is designed to provide employees with an opportunity to self-finance a planned period of absence for any reason, such as continuation of educational studies, travel, or other interests.

The Program allows the financing of a leave of absence by deferring an amount of not less than 10 percent and not more than 33 1/3 percent of gross salary before tax. The program will comply with Federal Income Tax Regulations. The

Standards of Conduct for public service employees, must continue to be observed during the leave period.

ELIGIBILITY

The Program is open to all regular employees appointed pursuant to Sections 8 and 12 of the *Public Service Act*, except salaried physicians and Order In Council appointments under Category C and D. In addition, a minimum of 24 months' continuous service must be completed before an employee can begin deferring salary.

APPROVAL PROCESS

Approval of participation in the Deferred Salary Leave Program will rest with the **ministry**. Criteria for approval will depend on the operational requirements of the work unit and will be subject to the employee's eligibility for participation.

SALARY DEFERRAL PERIOD

The salary deferral period may not be less than one year and may not exceed six consecutive years. The deferred portion of salary will be deposited **in a trust account in the employee's name** and will accrue interest.

LEAVE PERIOD

The minimum leave period will be six consecutive full calendar months and the maximum will not exceed twelve consecutive full calendar months. The leave should commence immediately following the end of the deferral period. Federal Income Tax Regulations require employees to return to work for a period equal to the duration of the leave. Therefore, deferred salary leave cannot serve as an early retirement benefit.

RETURN TO EMPLOYMENT

Employees will return to their employment for a period of time **no less than** the duration of the leave. Employees will return to their former position or an equivalent position in the **ministry**. This will be decided prior to granting the employee participation in the program and will be included in the terms and conditions of approval of their leave.

SALARY DEFERRAL DEDUCTIONS

Participants authorize the Employer to deduct their contributions and deposit these amounts in the employee's trust account. Each "contribution" an employee makes towards the leave is deposited into a trust account in the employee's name with the service provider, who administers the program on behalf of the BC Public Service. Enrolled employees will receive quarterly statements from the service provider, and will be responsible for reconciling these statements against their payroll records.

The employee can invest the funds in a Guaranteed Investment Certificate or in a savings account. The employee can also split the investment of funds between the two options. The funds are placed in a trust account in the employee's name.

ADMINISTRATION FEES

It will be the responsibility of the participant to pay any **fees** levied by the **service provider.** The accumulated principal will be guaranteed for payment to the employee during their leave period or upon withdrawal from the program.

TAX ISSUES

The Employer is not responsible for providing tax advice. Employees will be expected to seek advice with respect to tax concerns from Revenue Canada or professional taxation consultants.

PREPARATION FOR LEAVE

At least three months prior to the commencement of leave, the employee will confirm their leave start date to the ministry, and complete required documentation to initiate the payment of deferred funds.

EXTENDING DEFERRAL PERIOD OR POSTPONING LEAVE OF ABSENCE

Changes to personal and/or family circumstances may occur during participation in this program. An employee may postpone the leave of absence or the employee's supervisor may ask an employee to postpone the leave **provided that the employee remains compliant with the rules of the program.** In either circumstance, three months' written notice will be required.

The leave period must commence no later than six years after the first contribution and must be completed no later than seven years after the commencement of the deferral period. Revenue Canada Regulations regarding payment of deferred funds must be followed.

CHANGES TO APPROVED APPLICATION

Employee-initiated changes to the length of the deferral period or the leave of absence must be approved by the **ministry**.

CHANGE IN CONTRIBUTION AMOUNT

An employee may change the percentage amounts being deferred through payroll deductions once per calendar year.

Reductions in deferral amounts due to financial hardship may occur at any time. **The contribution amount cannot be reduced to less than 10 percent of gross salary.**

CHANGE IN JOB PRIOR TO LEAVE

When applying for a position that could result in a promotion, transfer or secondment, it will be the employee's responsibility to notify the new employing ministry of their participation in the program. It is the employing ministry's responsibility to advise the employee at the time an offer is made as to whether the employee will be permitted to continue in the program.

WITHDRAWAL FROM PROGRAM

Participating employees may withdraw from the program only under the following circumstances:

- financial hardship;
- move to another position where continuation in the program is not approved by the employing ministry;
- total and permanent disability as defined in the Long Term Disability Plan;
- termination of employment;

- if for any reason, the employee is unable to comply with program rules, and
- death of the employee.

Accumulated deferred funds will be paid out as a lump sum in the same calendar year as the cancellation and are subject to income tax.

BENEFIT ENTITLEMENTS AND CONTRIBUTIONS

Public Service Pension Plan

During the deferral period, contributions to the Public Service **Pension** Plan will be based on 100 percent of full salary.

Contributions cannot be made during the leave period. Upon return to work, the employee may purchase pensionable service based on BC Pension Corporation and Canada Revenue Agency rules in place at that time.

All Other Benefits

During the deferral period, benefit entitlements and contributions will be based on gross salary earned.

During the leave period, the same conditions as a leave of absence without pay will apply with respect to benefit entitlements and contributions.

SALARY ADJUSTMENTS

Individual salary adjustments will be deferred when an employee is on a deferred salary leave as follows:

- from six to ten months' absence deferred six months; and
- from ten to twelve months' absence deferred one year.

RESPONSIBILITIES

Ministries are authorized to:

- approve applications;
- approve assignment of returning employees to own or equivalent position;
- authorize changes to the length of deferral period or the leave of absence;

• approve participant withdrawal from program in financial hardship situations.

Provincial Treasury is authorized to:

- issue a Request for Proposal to financial institutions for Deferred Salary Leave Program services;
- negotiate fees for the services to be charged by the financial institution;
- establish a legal agreement with the financial institution for the provision of the services and the safeguarding of program assets.

ACCOUNTABILITIES

Deputy Ministers are accountable for ensuring that:

- a system is in place to communicate all of the conditions and terms of the program to interested employees;
- all administrative requirements associated with the program are followed; and
- the *Public Service Act* Directive, Standards of Conduct, requirements are followed.

The Office of the Comptroller General is accountable for ensuring that:

• Revenue Canada requirements are met.

The Participant is accountable for ensuring that:

- financial statements are reconciled and any discrepancies are reported in writing to the **service provider immediately.**
- benefits selected during leave period are arranged for and maintained;
- all terms and conditions are completed and complied with; and
- contributions are made directly to the **service provider** while in receipt of wage loss benefits, if the participant wishes to continue in the program.

The BC Public Service Agency is accountable for ensuring that:

- ministries are informed of current policy and administrative procedures;
- policy-related issues are addressed; and
- records, statistics and reports on the Deferred Salary Leave Program are maintained.

***INFORMATION APPENDIX E DISCRIMINATION AND HARASSMENT IN THE WORKPLACE**

This policy statement applies to all employees appointed under the *Public Service Act* and applies to incidents that occur at or away from the workplace during or outside working hours if a connection exists to the employment relationship.

The human resource policy can be found at: http://www2.gov.bc.ca/gov/content/careers-myhr/managerssupervisors/employee-labour-relations/conditions-agreements/policy

*INFORMATION APPENDIX F RE: 15th MASTER AGREEMENT MEMORANDUM OF UNDERSTANDING #21 ASSISTANT LEADER

The Parties agree this information is provided for the term of the 16th Master Agreement as an Information Appendix.

For the term of the 15th Master Agreement, the Parties agree to establish an additional basic rate of pay, outside of Appendix 6, for the sole purpose of compensating Nurses in the position of Assistant Team Leader (ATL) in Child and Youth Mental Health Programs, Ministry of Children and Family Development.

This rate of pay will be halfway between current existing rates of pay for Nurse 7 and Nurse 9. The rates will be adjusted to reflect the general wage increase applicable to the bargaining unit members. The wage schedules for the term of the Agreement are reflected in the Attachment to this Memorandum.

For Administrative purposes a new classification code "Nurse (ATL)" will be created.

ATTACHMENT TO MEMORANDUM OF UNDERSTANDING #21

Assistant Team Leader Wage Schedule Ministry of Children and Family Development

Effective October 15, 2014, there will be a 1% increase to all grids and steps on the salary schedule.

The salary schedule is based on six steps with 18-month intervals and reflects increases outlined in Appendix 6.

Step	Annual	Monthly	Bi-weekly	Hourly
1	64,300.00	5,358.33	2,464.61	35.2087
2	69,246.46	5,770.53	2,654.21	37.9173
3	71,736.55	5,978.05	2,749.65	39.2808
4	76,665.63	6,388.81	2,938.58	41.9797
5	78,982.33	6,581.86	3,027.38	43.2483
6	84,397.03	7,033.08	3,234.93	46.2133

*INFORMATION APPENDIX G RE: 15th MASTER AGREEMENT – APPENDIX 6 EFFECTIVE OCTOBER 15, 2014 (ANNUAL, MONTHLY, BI-WEEKLY AND HOURLY RATES)

The Parties agree this information is provided for the term of the 16th Master Agreement as an Information Appendix.

Effective October 15, 2014, there will be a 1% increase to all grids and steps on the salary schedule.

Grid Level	Step	Annual	Monthly	Bi-weekly	Hourly
Interim Permit	1	54,997.86	4,583.16	2,108.06	30.1152
	1	55,694.30	4,641.19	2,134.76	30.4965
	2	59,978.04	4,998.17	2,298.95	32.8422
4	3	62,135.60	5,177.97	2,381.65	34.0236
4	4	66,404.33	5,533.70	2,545.27	36.3610
	5	68,411.16	5,700.93	2,622.19	37.4599
	6	73,101.23	6,091.77	2,801.96	40.0280
	1	58,979.11	4,914.92	2,260.66	32.2952
	2	63,516.34	5,293.03	2,434.57	34.7797
5	3	65,800.39	5,483.37	2,522.12	36.0303
5	4	70,321.03	5,860.09	2,695.40	38.5056
	5	72,446.17	6,037.18	2,776.85	39.6694
	6	77,412.65	6,451.05	2,967.22	42.3888
	1	62,458.39	5,204.86	2,394.02	34.2003
		,	· ·		
	2	67,262.56	5,605.22	2,578.17	36.8310
7	3	69,681.24	5,806.77	2,670.87	38.1554
	4	74,469.07	6,205.75	2,854.39	40.7770
	5	76,719.64	6,393.30	2,940.66	42.0093
	6	81,979.13	6,831.60	3,142.25	44.8893

The Salary Schedule is based on six steps with 18-month intervals.

9	1	66,142.14	5,511.84	2,535.22	36.2175
	2	71,229.85	5,935.82	2,730.23	39.0033
	3	73,791.87	6,149.32	2,828.43	40.4062
	4	78,861.65	6,571.81	3,022.76	43.1822
	5	81,245.29	6,770.44	3,114.12	44.4875
	6	86,814.92	7,234.58	3,327.61	47.5373

*MEMORANDUM OF UNDERSTANDING #1 NEW FACILITIES

It is agreed that all vacancies for professional nurses to be established in new facilities shall be posted internally and current employees covered by this Agreement will be given preference in filling such vacancies, wherever possible.

MEMORANDUM OF UNDERSTANDING #2 RESIGNATION PROMPTING

It is understood and agreed that the Employer will not make any suggestions to the employee that they resign in order to facilitate ease of termination but, rather, that the proper procedures for the dismissal of an employee as contained in this Agreement are pursued if appropriate.

Notwithstanding the foregoing:

- if the suggestion to resign is of the employee's own volition, or
- if the Employer representative states, in the presence of a Union steward, that the Employer is prepared to accept a resignation,

then it shall be deemed as not contradictory to the intention of this Memorandum.

*MEMORANDUM OF UNDERSTANDING #3 SAFEGUARDING VULNERABLE PEOPLE

The parties recognize that within the Public Service there are employees whose work assignment brings them in contact with vulnerable individuals. It is in the public interest that such employees do not have a history of behaviour which is incompatible with such assignments.

The parties recognize that an employee's privacy and reputation must be recognized and protected. Accordingly, information gathered to establish suitability for work assignments which include contact with vulnerable individuals is to be treated with confidentiality to the fullest extent compatible with meeting the Government's responsibility of safeguarding vulnerable people.

Vulnerable people includes **adults with physical/mental disabilities, and includes** children.

*MEMORANDUM OF UNDERSTANDING #4 PRE-EMPLOYMENT ELIGIBILITY AND CHECK-OFF ADMINISTRATION

The following is the Authorization of Dues Deduction form approved by the Parties, as provided for in Article 4(b):

TO Ministry of:	
in writing, IAUTHORIZE the Govern	g registration as RPN RN RN both and until this authority is revoked by me (print name) HEREBY ment of the Province of British Columbia to deduct from my wages and payable a Nurse' Union, the amount of the regular dues payable to the Union by a
I further authorize that the to the Union.	e Government of the Province of British Columbia provide the following information
Mailing Address	
Postal Code	
Home telephone	Work telephone
Nursing Registration N	umberor Social Insurance No
Job classification	
Employment Site / locat	ion address:
Home Email (optional)	
Employment Status	Regular Full-Time 🛛 Casual Auxiliary 🛛
	Regular Part-Time 🛛 On-Call Auxiliary 🛛
Signature	Date
	SES'LINION

BRITISH COLUMBIA NURSES' UNION 4060 Regent Street Burnaby BC V5C 6P5

*MEMORANDUM OF UNDERSTANDING #5 TRANSFERS WITHOUT POSTING

Re: Master Agreement Clause 12.09(b)

The parties agree to establish a consultative joint committee comprised of two representatives from the Employer and **two** representatives from the union. The joint committee shall meet every 90 days or at the call of either party.

The terms of reference for the committee shall be:

- (1) to serve as a reviewing forum for employee initiated lateral transfers;
- (2) to establish an administrative process for the review of such requests;
- (3) to consider such requests in the context of:
 - (a) operational requirements
 - (b) employee preferences; and
 - (c) interests of the Public Service;

(4) to facilitate consistent and equitable treatment of such requests which may include circumstances such as spousal transfer, family support systems, educational opportunities;

(5) to, where deemed by the committee to be appropriate, make recommendations to the appropriate Deputy Minister, or designate, in respect of any individual request.

Applicants for employee initiated lateral transfers shall be provided with written reasons if their request is denied.

*MEMORANDUM OF UNDERSTANDING #6 RE: HUMAN RIGHTS CODE

The parties subscribe to the principles of the *Human Rights Code* of British Columbia.

The parties continue to promote within the framework provided by the law, an

effective work environment that is free from discrimination where all employees are treated with respect and dignity.

The parties will continue to review methods of extending knowledge of the *Human Rights Code* to employees.

As such:

- On the date of signing of the 12th Master and Component Agreement, a joint Labour/Management ad hoc Committee will be formed for the purposes of reviewing the current policy and procedures of the various worksites covered by this agreement.
- The Committee shall consist of two representatives from the **Union** and two representatives of the BC Public Service Agency. The Committee shall also add other representatives from specific worksites on a temporary basis as needed.
- The Committee's mandate will be to review the discrimination policies, procedures and practices of the individual worksites to ensure that they are in compliance with current jurisprudence and practices. In particular, the committee will review the assumptions that result in determining whether a certain job or work unit requires a specific gender and determine if such restrictions are appropriate.
- The Committee shall also develop an information pamphlet that describes the duty to accommodate and the obligations that this creates for the Employer, Unions and individuals.
- The Committee will meet within sixty (60) days of the date of signing the **16**th Nurses Master and Component Agreements and shall meet regularly until all relevant worksites have been reviewed and the information pamphlet has been completed. Once the mandate is completed the Committee will stand down.

*MEMORANDUM OF UNDERSTANDING #7 REVISIONS OF BENEFITS

Effective the date of signing and notwithstanding any other provision of the collective agreement, the Parties agree that the following provisions shall be revised and implemented on the same basis as revised and implemented for the majority of unionized employees in the Public Service.

- 1. Clause 15.03 Shift Differential
- 2. Clause 15.05(a) Changes in Schedules (premium only)
- 3. Clause 20.03(b) Union Business or Public Duties
- 4. Clause 21 Maternity/Parental/Adoption Leave
- 5. Clause 22.12(c) Occupational First Aid Requirements and Courses
- 6. Clause 25.01 Basic Medical Insurance
- 7. Clause 25.02 Extended Health Care Plan
- 8. Clause 25.03 Dental Plan
- 9. Clause 25.04(b) Group Life
- 10. Clause 25.05 Air Travel Insurance
- 11. Clause 27.10 Vehicle Allowance
- 12. Clause 27.11 Meal Allowance
- 13. Clause 27.13 Accommodation, Board and Lodging Allowance
- 14. Clause 27.15 Retirement Allowance
- **15.** Clause 27.17 Relocation Expenses
- 16. Clause 27.22 Child Care Expenses
- 17. Clause 29.07(a) Health and Welfare
- 18. Clause 29.11 Weekly Indemnity
- 19. Memorandum of Understanding 14 Board, Lodging and Relocations
- 20. Appendix 3 Short Term Illness and Injury Plan and Long Term Disability Plan
- 21. Clause 6.02 Community Component Overtime Meal Break and Allowance
- 22. Clause 5.01 Hospital Component Overtime Meal Allowance

Note: Bolded items were amended in the **16th** Master Agreement.

MEMORANDUM OF UNDERSTANDING #8 EMPLOYMENT EQUITY

Pursuant to the Provincial Government's Employment Equity Policy, the Employer and the Union agree to establish a Joint Standing Committee mandated to develop a framework for implementation and maintenance of employment equity measures as they apply to nurses covered under this collective agreement:

(a) The Joint Committee will be comprised of four representatives appointed by the Employer and four representatives appointed by the Union.

(b) Employees representing the Union on this Joint Committee shall be on leave of absence without loss of basic pay for time on this Committee.

(c) The Joint Committee may establish sub-committees and consult with other government groups as may be necessary in carrying out its mandate.

(d) The parties agree that sufficient priority will be given to the work of the Joint Committee to ensure timeliness.

(e) As per Clause 31.04 of this Agreement amendments to the existing collective agreement related to employment equity may be made by mutual agreement of the parties.

*MEMORANDUM OF UNDERSTANDING #9 NURSING ADVOCACY COMMITTEE – PROFESSIONAL RESPONSIBILITY

A. <u>Committee Structure</u>

At the request of either party, the parties will form a joint committee of three representatives from the Employer and three from the Union.

The Union representatives attending at committee meetings will be granted Union leave without loss of basic pay.

Meetings of the committee shall be held at the call of either party within 21 calendar days of such call.

Members of the committee shall have access to any Ministry or facility policy and procedure manuals as required to undertake its mandate.

Part B of this Memorandum is not subject to the grievance or arbitration procedures of Articles 8 and 9 of the Master Agreement.

B. <u>Committee Mandate and Responsibilities</u>

The mandate of the committee is to ensure that direct caregiver input is an integral element of the Employer's policy development and response to the changing demands in health care delivery.

The committee will be charged with specific responsibility for the following areas:

Professional Responsibility

In the interest of safe patient/resident/client care and safe nursing practice, the parties agree to a problem solving process to address employee concerns relative to patient/resident/client care including:

- nursing practice conditions
- safety of patients
- workload

The committee will oversee the reporting infrastructure outlined below, which is a vehicle for employees to raise concerns in respect of the above.

(a) <u>Hospital Component</u>

(i) An employee with a concern related to safe patient/resident/client care or safe nursing practice shall discuss the matter with their immediate supervisor. Every effort will be made to resolve the concern at this stage.

(ii) Within 14 calendar days of the discussion pursuant to (i) above, where the matter remains unresolved, the employee shall refer it to the responsible excluded manager. The employee shall provide the manager with specific details of their concern, in writing, with a copy to their immediate supervisor.

(iii) Within 14 calendar days after receipt of the written concern pursuant to (ii) above, should the matter remain unresolved, the employee shall refer it to the local standing joint committee. If it is determined by the local committee to be worksite specific, local initiatives shall be implemented, where possible, to resolve the matter.

(iv) If the concern is not resolved or if it is determined to have broader implications, the local standing joint committee shall refer the matter to the Nursing Advocacy Committee formed under this Memorandum.

(b) <u>Community Component</u>

(i) An employee with a concern related to safe patient/resident/client care or safe nursing practice shall discuss the matter with their immediate supervisor. Every effort will be made to resolve the concern at this stage.

(ii) Within 14 calendar days of the discussion pursuant to (i) above, where the matter remains unresolved, the employee shall refer it to the responsible excluded manager. The employee shall provide the manager with specific details of their concern, in writing, with a copy to their immediate supervisor.

(iii) Within 14 calendar days after receipt of the written concern, the employee, the employee's union representative or designate, the employee's immediate supervisor and the responsible excluded manager shall discuss the concern. If it is determined to be worksite specific, local initiatives shall be implemented, where possible, to resolve the matter.

(iv) If the concern is not resolved or if it is determined to have broader implications the employee or the responsible excluded manager shall refer the matter to the Nursing Advocacy Committee formed under this Memorandum.

(c) The Nursing Advocacy Committee shall review the matter and, if required, make recommendations to the Deputy Minister or facility Chief Executive Officer, as the case may be, as to options available for resolution.

(d) Within 30 calendar days after receipt of the committee's recommendations, the Deputy Minister or facility Chief Executive Officer as the case may be, shall advise the employee in writing of their decision with copies to the Nursing

Advocacy Committee, the responsible excluded manager, the employee's immediate supervisor, and the local standing joint committee (Hospital Component).

The Nursing Advocacy Committee may make specific recommendations to the bargaining principals which relate to improving the processes outlined in this Memorandum and any other recommendations it may consider necessary to promote and maintain safe patient/resident/client care, safe nursing practice and the safety of patients and nurses.

MEMORANDUM OF UNDERSTANDING #10 AUXILIARY EMPLOYEES HIRED FOR DOWNSIZING/CLOSING FACILITY

In the event of a downsizing or closure of a hospital, clinic or other facility, individuals hired as an auxiliary employee, notwithstanding the Master Agreement, shall not have their status changed to regular provided:

(1) the employee is advised of such status and condition in their letter of appointment;

(2) the purpose is to facilitate the downsizing or closure of a hospital, clinic or other facility;

(3) such auxiliary appointment shall not exceed three calendar years.

In the event such auxiliary appointment exceeds three calendar years, then the auxiliary status will be determined in accordance with Clause 29.14 of the Master Agreement commencing at the date the appointment exceeds three years.

Should such an auxiliary employee be laid-off, within the first three years of their appointment, then:

(1) if the auxiliary employee is working in a hospital, clinic or other facility that is not being downsized or closed then the auxiliary employee will be subject to Article 29 of the Master Agreement.

(2) if the auxiliary employee is working in a hospital, clinic or other facility that is being downsized or closed, then the auxiliary employee's seniority will be lost the date of layoff.

*MEMORANDUM OF UNDERSTANDING #11 LABOUR ADJUSTMENT

Should the Employer or the Union believe there is a need for a labour adjustment program in response to a downsizing and/or closure during the term of this Agreement, the parties agree to meet and establish a labour adjustment program which recognizes that there may exist issues which are unique to the terms and conditions of employment established between the parties to this Agreement. In developing the labour adjustment strategy, the parties may consider other issues including, but not necessarily limited to:

- principle that the first obligation is to employees of the bargaining unit
- job placement enhancement initiatives, including canvassing affected employees for voluntary options
- expansion of displacement capabilities, including access to vacancy/displacement options equivalent to those available to laid off employees
- layoff notification extensions

Notwithstanding the generality of the foregoing, the parties agree that devolution at times is an ongoing poly-party process, which may have direct affect on employees, as patients are placed in different settings.

As such, the parties agree that flexibility and creativity best serve the objective of minimizing adverse impact on employees.

The parties agree there are a number of key mechanisms which can assist employees and the Employer in ensuring that transitions properly serve the interests of patients, employees and health care objectives.

If the parties agree that there is a need for a labour adjustment program, they will establish a Labour Adjustment Committee (LAC), specifically to respond to the downsizing and/or closure.

Terms of Reference

- 1. The Committee will be comprised of no more than four appointees each from the Union and Employer.
- 2. The Committee will meet no later than thirty days after the date of the announcement of a potential closure or downsizing, and thereafter at the call of either party.
- 3. A primary objective of the Committee is to ensure the dissemination of timely and accurate information including, but not limited to:
 - (a) ward closures;
 - (b) employee options;
 - (c) communication options;
 - (d) type and level of information sharing;
 - (e) applicability of ERIP/VDP initiatives;
 - (f) seniority lists as required.
- 4. The Employer will provide timely information generally as well as specifics for individual employee circumstances, such as confirmation of all regular and auxiliary seniority totals with the Employer.
- 5. The parties agree that third parties may assist in providing timely fact-based input which will assist in Committee deliberations.

Nothing in the Memorandum of Understanding detracts from or compromises either parties' rights under the **16th** Nurses Master and Component Agreements or relevant legislation.

*MEMORANDUM OF UNDERSTANDING #12 TRANSITION COMMITTEE

Re: Transition of Health Care from the British Columbia Public Service to Health Authorities and/or Other Health Care Employers

The Parties agree that the transition of health care from the British Columbia Public Service to Health Authorities and/or other health care employers will not be accomplished by means of, nor result in, a shift of services from the present employer to other employers unless said employers are required to become the successor employer to the Government and are required to employ the current employees, including auxiliary employees, engaged in providing such services.

In order to facilitate the transition of health care delivery services from the public service to Health Authorities and/or other health care employers, the Parties to this memorandum agree to convene a joint committee to ensure that the transitional process proceeds in a manner that takes into consideration, the interests of each of the Parties.

The joint committee will be comprised of members of each of the following:

BCPSA HEABC or the bargaining representative of the receiving employer the Public Service Employer(s), and the **BCNU**

The Employer commits HEABC or the bargaining representative of the receiving employer and the Public Service Employer(s) as participants on the joint committee.

The joint committee will meet as required at the call of BCPSA or the BCNU.

The terms of reference of the committee will be to address those matters which arise as a result of the changed employment relationship resulting from the reallocation of human resources from the Government of British Columbia to the health authorities and/or other health care employers. This shall include but not be limited to the following:

1. The process by which regular and auxiliary nurses are transferred to the new employers;

2. The creation of an employee tracking system to provide all Parties with accurate and timely information on those nurses who may be affected by the transfer of programs or service to the new employers;

3. The Government of British Columbia (BCPSA) will identify for the joint committee those public service programs and associated nursing positions which will remain with the Government of British Columbia.

*MEMORANDUM OF UNDERSTANDING #13 ROLE OF THE REHABILITATION COMMITTEE

RE: JOINT ADVISORY COMMITTEE

The Parties agree to establish a Joint Advisory Committee consisting of up to four representatives from the Union and up to four representatives from the Employer. The purpose of the Committee shall be to consider and make recommendations to the bargaining principals on all matters related to the effective administration of the Short Term Illness and Injury Plan and Long Term Disability Plan and the Rehabilitation Committee, and on any questions which may arise related to the interpretation or application of wording of Appendix 3.

The Joint Advisory Committee will make recommendations concerning the Rehabilitation Committee **in order** to:

- (a) improve access to the rehabilitation process for employees incapacitated for their own occupation through illness or injury;
 - improve rehabilitation programs to return employees to their own or other occupations as soon as possible;
 - identify and address systemic causes of illness and injury and consequent STIIP/LTD usage.

(b) Clearly establish responsibility for case management, with the Committee providing advice and recommendations as required. Such recommendations may include:

• improved placement options for those employees who are capable of performing alternative employment, in addition to the recommendations identified in Appendix 3, Part IV (d)(4).

Disability Case Management representatives are to be designated as Employer representatives to the Committee.

(c) Ensure sharing of all information pertinent to a case with the parties involved (union, employer, **Occupational Health and Rehabilitation**, insurance carrier).

Develop confidentiality standards specific to the process and consistent with the current legislation to protect the privacy of information shared.

(d) Review current forms used for STIIP and LTD and Rehabilitation in order to make them simple and more effective and/or eliminate duplication.

The Joint Advisory Committee shall consult with the Provincial Joint Occupational Health and Safety Committee, the Rehabilitation Committee and/or Ministry Joint Committee, as appropriate.

The Committee will meet within 60 days of ratification of the 16th Master Agreement.

*MEMORANDUM OF UNDERSTANDING #14 BOARD AND LODGING AND RELOCATION EXPENSES

Definitions

For the purpose of these regulations:

"stationary employees" are employees who occupy positions that require them to

- (a) carry out their duties on a day-to-day basis at their headquarters; and/or
- (b) travel from their headquarters for short periods of time; and/or

(c) travel from their headquarters more or less on a continuous basis, but whose assignments are of sufficiently short duration so that temporary headquarters cannot be practically assigned;

"*mobile employees*" are those who occupy positions requiring assignment to a "*temporary*" headquarters for a significant period of time for each specific project and who are required to carry out their duties on a day-to-day basis from their assigned temporary headquarters; these employees are usually required to change their temporary headquarters on a continual basis and would not be domiciled at a permanent headquarters;

"*field status employees*" who are normally required to work away from their point of assembly and who, on a day-to-day basis, do not work in an office, institution, plant, or other similar fixed location which is their normal point of assembly;

"*seasonal field employees*" are those employees who occupy positions which permit them to be normally domiciled at their permanent headquarters but who are assigned field duties on a seasonal basis, returning to their permanent headquarters when not working in the field;

"*permanent camp*" is a camp which will be established and occupied continuously for more than one year;

"*seasonal camp*" is a camp that will be established and occupied less than five months and is usually comprised of tents and, where feasible, trailers;

"*fly or sub-base camp*" is a camp that will be established and occupied on a very temporary basis, is mobile in nature, and is generally isolated with very restricted access;

"*local hire*" is a person who is hired or is domiciled within 80 kilometers of the job site by means of the shortest road route;

"*travel status*" with respect to an employee means absence of the employee from the employee's designated headquarters or geographic location on Government business with the approval of the Employer, but travel status does not apply to employees temporarily assigned to a position outside of the designated headquarters or to field status employees;

"*headquarters or geographic location*" is that area within a radius of 32 kilometers of where an employee ordinarily performs their duties. When employees are relocated, this definition may be redefined where exceptional circumstances such as unusual road conditions exist.

"*dependents*" for the purpose of definition, dependents are spouse, dependent children and anyone for whom the employee claims exemption on Federal Income Tax returns;

"*private dwelling house*" refers to the single family residence of the employee on a reasonable amount of property required to support such a house, owned by the employee and/or the spouse, and for which evidence of title can be provided. "House", "residence" and "property" refer solely to the property occupied as the principal residence of the employee at the time of relocation, including mobile homes;

"*reasonable amount of property*" where an employee elects to purchase a dwelling house on a piece of property that would not be considered a "reasonable amount" (i.e., hobby farm, etc.), the following formula shall be used to determine the value of the private dwelling house for legal fee reimbursement purposes:

- (a) value of an average serviced lot in or close to the nearest town;
- (b) assessed value of actual house on site;
- (c) total added value in (a) and (b).

Part I Board and Lodging Regulations

1.01 Consistent with the intent of Memorandum of Understanding **#7**, the Treasury Board Order respecting Board and Lodging Regulations shall be made available to employees, from their local pay office, upon an employee's request.

Part II Relocation Expenses

2.01 Policy

(a) Relocation expenses will apply:

(1) to regular employees and to auxiliary employees who qualify pursuant to Clause 29.02 who have to move from one headquarters or geographic location to another after completing their probation period and after winning an in-service competition where the position is permanently located at another headquarters or geographic location;

(2) to employees who have to move from one headquarters or geographic location to another at the Employer's request to fill a position which is permanently located at another headquarters or geographic location.

(b) Relocation expenses will not apply, but instead the applicable travelling, living and moving expenses provided under the Board and Lodging Regulations will apply to the following groups of employees who will not be considered to be on relocation:

(1) to field status, mobile and other employees whose normal duties require moves from one temporary headquarters to another or from one assignment to another.

(2) to field status, mobile and other employees who are successful applicants for posted positions, where such positions are not permanently located at one headquarters or geographic location, such as is the usual case with field crew positions.

(3) to apprentice employees when there is a pre-programmed change in their headquarters or geographic location.

(c) To employees entitled to relocation expenses, the Employer will pay travelling, living and moving expenses on relocation in accordance with the following provisions.

2.02 Travel Expenses on Relocation

(a) Initial trip to seek new accommodation:

The Employer shall grant, with no loss of basic pay, prior to relocation, at a time mutually agreeable to Employer and employee, up to five days plus reasonable travel time, to an employee being relocated and shall reimburse the employee for travel expenses for the employee and spouse in accordance with the current Treasury Board Order on Travel Expenses.

Any time beyond specified time may be charged against the employee's annual vacation credits, however, expenses will not be payable. This leave must be for the specific purpose of locating accommodation, with the intent, in as many instances as possible, that furniture and household effects may be delivered directly to the new residence.

(b) Travelling expenses moving to new location:

The Employer shall provide reimbursement of travel expenses incurred during the relocation for employees and dependents, for the actual travel time, plus accommodation and meals up to seven days at the new location, when employees are unable to move into the new accommodation.

Such expense allowances will be in accordance with the current Treasury Board Order on Travel Expenses.

Meals:	Adults - full rateChildren 12 and under - one-half rate
Motel or Hotel:	- on production of receipts.
Private lodging	- at old or new location at current rate.

(c) Where dependents of an employee relocate at a time different than the employee, the Employer shall reimburse the employee for their dependents' travel expenses, meals and accommodation incurred while travelling to the new headquarters area. In such cases where the employee remains eligible for benefits pursuant to Section 2.03, the employee will be reimbursed for their dependents' meals at the new location for a period of up to seven days.

The above allowances will be in accordance with the current Treasury Board Order on Travel Expenses.

2.03 Living Expenses Upon Relocation At New Location

After the first seven days has expired at the new location and the employee can establish to the satisfaction of the Employer that there is no suitable housing available, then:

(a) The Employer shall pay an employee not accompanied by dependents at the new location, a living allowance of \$25 per day up to a maximum of 30 days; or

(b) The Employer shall pay an employee accompanied by dependents at the new location, a living allowance of \$30 per day up to a maximum of 60 days.

(c) Where an employee is receiving the payment in (a) above and is later joined by their dependents at the new location and the employee is still eligible for payment under this Section, the payment shall be as in (b) above. However, the maximum period of payment under (a) and (b) shall not exceed 60 days.

2.04 Moving of Household Effects and Chattels

On relocation, the Employer shall arrange and pay for the following:

(a) moving of household effects and chattels up to 8,165 kilograms including any item(s) which the contract mover will accept as part of a load which includes household appliances and furniture, hobbies, boats, outboard motors and pianos;

(b) comprehensive insurance to adequately protect the employee's household effects and chattels during the move up to a maximum of \$60,000;

(c) where necessary, insured storage up to two months, upon production of receipts;

(d) the packing and unpacking of the employee's household effects and chattels;

(e) when an employee is being relocated and opts to move their own household effects and chattels, the employee shall receive one of the following allowances:

(1) \$500 for a move not exceeding a distance of 240 kilometers;

(2) \$800 for a move which exceeds a distance of 240 kilometers;

(3) \$250 where the employee is entitled to receive the amount pursuant to Section 2.07(d).

(f) Where the employee exercises an option pursuant to (e) above then the provisions of (a) and (d) above shall not apply.

2.05 Moving of Mobile Homes

(a) On relocation, an employee who owns a mobile home may opt to have their mobile home moved by the Employer in either of the following circumstances:

(1) where the employee's new headquarters area is on the list of isolated areas, providing no suitable accommodation is available, or

(2) where an employee is living in a mobile home which was moved to its present location by the Employer, and the employee's headquarters prior to the impending location is named on the list of isolated locations.

(b) Where an employee's mobile home is moved by the Employer under this Section then the Employer shall also arrange and pay for the following:

(1) moving of single wide mobile trailer or home up to the maximum width allowed on the highway with a permit including any skirting, cabanas or attachments. Where mobile homes in excess of the above are involved, the Employer will pay:

(i) the equivalent cost of moving a single wide mobile trailer or home up to the maximum width allowed on highways with a permit, or

(ii) the real estate and legal fees involved in selling the extra wide trailer up to a maximum of \$5,000;

(2) comprehensive insurance to adequately protect the employee's household effects, chattels and trailer during the move up to a maximum of \$60,000;

(3) the setting up and levelling of a mobile home or double wide, at the new location to a maximum of \$600 upon production of receipts;

(4) the packing and unpacking of the employee's household effects and chattels if required.

(c) Where an employee is living in a mobile home and is not included in (a) above, and chooses to move the mobile home to the new headquarters area, the employee shall be entitled to reimbursement for costs covered in (b) above up to a maximum of \$2,500 upon production of receipts.

(d) Where the employee opts under this Section to have a mobile home moved, there shall be no entitlement to the provisions of Section 2.04 and 2.10.

2.06 Moving of Personal Vehicles Upon Relocation

The Employer shall reimburse employees for the cost of transporting one personal vehicle and one trailer towed by the personal vehicle.

The vehicle and trailer, where applicable, may be driven in which case current vehicle allowance rates for the vehicle only will apply, or, vehicle and trailer, where applicable may be shipped by rail or boat, in which case the cost of the least expensive method will be paid.

In addition, the Employer will pay for any additional transportation charges such as ferry fares for the vehicle and trailer with or without load.

2.07 Incidental Expenses on Relocation

The Employer shall pay to the employee upon relocation only, one of the following amounts, to cover incidental expenses on relocation, and once the employee has claimed one allowance no alternate further claim may be made.

(a) when an employee purchases a private dwelling house in the new location - \$600;

(b) when the employee is moving to rental accommodation in the new location - \$300;

(c) when an employee is moving with a mobile home - \$200;

(d) When the employee is moving to room and board - \$150.

The application for incidental expenses on relocation must be made by the employee on the appropriate form within 60 days of the employee's arrival at the new location, unless there is no available suitable housing, in which case application must be made within 60 days of suitable housing becoming available.

2.08 Notice to Employee Upon Relocation

It is understood and agreed that the Employer will provide employees with reasonable notice of the relocation effective date and wherever possible, at least one month's notice shall be given. Where less than one month's notice is given, or the relocation date is altered either earlier or later than the relocation effective date given which directly results in duplication of rent costs to the employee, then the Employer agrees to reimburse the employee, upon production of receipts, for the duplicate rent payments at new location.

2.09 Requested Relocation by Employee

Where an employee requests a relocation from one headquarters or geographic location to another, all travelling and living expenses incurred in such a move are the responsibility of the employee.

2.10 Real Estate and Legal Fees

On relocation, or within one year of the effective date of relocation, an employee who purchases and/or sells their private dwelling house, will be entitled to claim for the following expenses upon production of receipts:

(a) Reimbursement of fees to a maximum of \$8,500; effective April 1, 2016
- \$8,700; and effective April 1, 2018 - \$8,900 charged by a real estate agency for the selling of the employee's private dwelling home in which they resided immediately prior to relocation;

(b) An employee who has sold their own home without the aid of a Realtor shall be entitled to claim \$2,000;

(c) Allowance for legal fees encumbered upon the employee because of the purchase of their private dwelling house in which they live after relocation will be paid in accordance with the following:

- (1) 1% of the first \$50,000 of the purchase price;
- (2) one-half of 1% of any amount of the purchase price above \$50,000;

(3) the total cost to the Employer under part (c) shall not exceed \$1,000; effective April 1, 2016 - \$1,025; and effective April 1, 2018 - \$1,050.

(d) Where an employee purchases a reasonable amount of property, secures a joint mortgage (land and private dwelling) and begins construction within six months of relocation (i.e., foundation poured), shall be entitled to reimbursement of legal fees not to exceed the amount specified in (c) above. In these circumstances, the reimbursement shall be for one transaction only;

(e) The employee may only claim legal fee reimbursement in either (c) or (d) above, not both.

Part III

Where a regular employee is required by the Employer to relocate (see Master Agreement Clause 12.08):

(a) as a result of the Employer moving its operation from one geographic location to another and which has been approved by Treasury Board;

(b) as a result of accepting a placement pursuant to Article 13, provided the employee is in receipt of layoff notice.

The employee will be entitled to the following reimbursements in addition to the provisions of Memorandum of Understanding #14, Part II, upon production of receipts:

(a) real estate commission fees not to exceed \$15,000. Where a claim is made under this section, there shall be no entitlement to Memorandum of Understanding #14 Part II, 2.10(a);

(b) except where the terms of the employee's mortgage allow the employee to transfer the mortgage to a new residence without penalty, the mortgage discharge fee not to exceed \$200 and mortgage pre-payment penalty, if any;

(c) survey certificate fee as required for the acquisition of a mortgage/purchase of a private dwelling at the new location;

(d) interim financing fees and/or interest charges incurred for the purchase of the private dwelling house in the new location for a maximum period of 60 days. The employee shall provide the necessary documentation to demonstrate that such interim financing arrangements were incurred and/or duplicate mortgage payments have been made.

Part III does not apply where the employee's private dwelling in which they resided immediately prior to relocation is not sold.

*MEMORANDUM OF UNDERSTANDING #15 RE: RECRUITMENT AND RETENTION INCENTIVE ADJUSTMENT

The Parties recognize that the public service shares in the systematic difficulty of recruiting and retaining nurses. Incentives to address these problems can assist public service employers in offering comparable career opportunities to prospective employees.

Therefore, the Parties agree to address specific recruitment and retention difficulties in the following manner:

For the term of the **16th** Nurses Master Agreement:

1. **Effective April 1, 2016** when working afternoon shift, as defined in the Master Agreement, the employee will receive a recruitment and retention supplemental payment of 30 cents per hour worked, less any ongoing increases in Clause 15.03, such that the total shift differential and supplemental payment does not exceed **\$1.70** per hour.

Effective January 1, 2018 when working afternoon shift, as defined in the Master Agreement, the employee will receive a recruitment and retention supplemental payment of 40 cents per hour worked, less any ongoing increases in Clause 15.03, such that the total shift differential and supplemental payment does not exceed \$1.90 per hour.

2. **Effective April 1, 2016** when working night shift, as defined in the Master Agreement, the employee will receive a recruitment and retention supplemental payment of \$1.65 per hour worked, less any ongoing increases in Clause 15.03, such that the total shift differential and supplemental payment does not exceed **\$3.15** per hour.

Effective January 1, 2018 when working night shift, as defined in the Master Agreement, the employee will receive a recruitment and retention supplemental payment of **\$1.75** per hour worked, less any ongoing increases in Clause 15.03, such that the total shift differential and supplemental payment does not exceed **\$3.35** per hour.

- 3. In addition to any other premiums or incentives received, effective January 1, 2018 an employee will receive a recruitment and retention supplemental payment of \$1.60 per hour worked between 2300 hours Friday and 2300 hours Sunday. Effective January 1, 2019 an employee will receive a recruitment and retention supplemental payment of \$1.70 per hour worked between 2300 hours Friday and 2300 hours Sunday.
- 4. For the purposes of applying the terms of the collective agreement the above noted supplemental payments will be applied in the same manner as the shift differential in Article 15.

*MEMORANDUM OF UNDERSTANDING #16 REVIEW OF WORKPLACE SAFETY TRAINING AND GUIDANCE

The Parties recognize that they share an interest in providing new mental health employees with workplace violence safety related training and guidance.

In this regard, representatives for the Employers and the Unions shall meet during the term of the 16^{th} Master Agreement to review existing resources which apply to workplace violence. Existing resources may include but shall not be limited to the following:

(i) educational workshops including the Worksafe BC rules and regulations, health and safety, relevant conflict resolution, problem solving and others;

(ii) available training to limit potential injuries in the workplace due to workplace violence;

(iii) the identification of current and future training needs and options related to the prevention of workplace violence;

(vi) relevant e-learning and audio/visual presentations;

Regarding a mentorship program, the Parties will select a Committee to explore the viability of a mentorship program. The task of the Committee will include, but not

be limited to:

- Recommending appropriate Policies and Procedures for mentor recruitment, such as screening, orientation and training, matching, match activities, ongoing support, recognition and match closure.
- Reviewing and recommending appropriate organizational support needs for a successful mentor program, such as identifying resources available or needed, advisory groups, and the management and sharing of information.
- Recommending a plan for evaluating the impact and outcomes for mentees, mentors, patients and the employers.

Notwithstanding any recommendations arising from the foregoing MOU, the Employers', Unions' and Employees' statutory responsibility for workplace safety will be adhered to. Consistent with these requirements, the Committee may make interim recommendations to local OH&S and/or the PJOSH. The Committee will arrange to meet within 60 days following ratification of the agreement.

*MEMORANDUM OF UNDERSTANDING #17 between The British Columbia Nurses Union and Maples Adolescent Treatment Centre

RE: EXTENDED WORK DAY/COMPRESSED WORK WEEK

Preamble:

The purpose of this Memorandum of Understanding (Memorandum) is to revise and/or clarify certain terms and conditions of the Collective Agreement, so as to provide for the introduction or continuance of an extended work day/compressed work week.

Any change deemed necessary may be made by mutual agreement between the parties during the life of this Memorandum.

Either party may terminate this Memorandum after serving twenty-eight (28) calendar days' written notice to the other party of its intention to terminate the extended work day/compressed work week.

It is understood and agreed that:

(A) With the exception of the specific revisions set forth in this Memorandum, all other terms and conditions of the Master and Component Collective Agreement will apply.

(B) As a general principle and unless otherwise revised in this Memorandum, the Employer will not incur any additional costs which would exceed the costs required to provide and maintain the regular work day/work week as set forth in the Collective Agreement.

(C) As a general principle and unless otherwise revised in this Memorandum, the employees will neither gain nor lose any benefit(s) presently contained within the Collective Agreement.

(D) For the purposes of this Memorandum and where revised, "days" have been converted into working hours, so that one (1) day shall equal seven point five (7.5) paid hours. For example, three (3) days Special leave is converted to $3 \times 7.5=22.5$ working hours.

The parties to this MOU agree that the following provisions will apply:

Further to Article 14 (Hours of Work) of the Master Agreement, it is understood and agreed that the hours of work as set out hereunder are specifically revised to conform to the requirements of the extended work day/compressed work week.

Further to Article 3.01 (Shifts, Shift Schedules and Shift Rotations) of the Hospital Component,

(a) All employees covered by this Memorandum shall work the schedule as set forth below.

(b)It is agreed that the interruption of the shift schedule will be kept to a minimum.

Further to Article 3.05(b) (Rotations and Adjustments) of the Hospital Component, the following Shift schedules will be in effect:

- 1. 5 days on, 2 days off, 4 evenings on, 3 off, at 7.78 hours per shift. This schedule shall be Monday to Friday days and Monday to Thursday evenings.
- 2. 4 nights on, 3 nights off, at 8.75 hours per shift. This schedule shall be Monday to Thursday nights.
- 3. 3 nights on, 4 off, at 11.25 hours per shift. This schedule shall be Friday to Sunday.
- 4. 2 days on, 5 days off, 11.50 hours per shift. This schedule shall be Saturday and Sunday.

Further to Article 14.02 of the Master Agreement and Article 2.01 of the Component Agreement:

• Monday to Friday Day Shift: 7.78hrs interrupted only by the meal break as provided in Article 14.03 of the Master Agreement and Article 2.02 of the Component Agreement;

- Monday to Thursday Afternoon Shift: 7.78hrs interrupted only by the meal break Monday to Thursday break as provided in Article 14.03 of the Master Agreement and Article 2.02 of the Component Agreement;
- Night Shift: 8.75hrs interrupted only by the meal break;
- Saturday and Sunday shifts: 11.50 hrs interrupted only by the meal break as provided in Article 14.03 of the Master Agreement and Article 2.02 of the Component Agreement;
- A Full time Shift Supervisor will work Friday to Sunday at 11.25 hours per day interrupted only by the meal break as provided in Article 14.03 of the Master Agreement and Article 2.02 of the Component Agreement. This position shall be rotated among the Shift Supervisors.

Further to Article 2.02 (Meal Time) of the Component Agreement:

- 7.78 hour shifts are entitled to one 30 min meal break
- 8.75 hour shifts are entitled to one 30 min meal break
- 11.25 hour shifts are entitled to a 45 min meal break
- 11.50 hour shifts are entitled to a 30 min meal break

Further to Article 14.04 (Rest Periods) of the Master Agreement:

- (i) for the shifts 7.78 and 8.75 hours, two rest periods of 15 minutes each will be granted during each full work day or shift.
- (ii) for the shifts 11.25 and 11.50 hours, three rest periods of 15 minutes each will be granted during each full work day or shift.
- (iii) One rest period of 15 minutes will be granted during the work day of three and one-half to six hours duration.

Further to Article 3.02 (Shift Break) of the Component Agreement:

• For those working a 7.78 hour shift, if shifts are scheduled so that there is not a rest period of 15 hours and 28 minutes between the end of one work period and the beginning of the next, overtime rates will apply to the hours worked on the succeeding work period within the 24-hour period.

- For those working a 8.75 hour shift, if shifts are scheduled so that there is not a rest period of 14 hours and 30 minutes between the end of one work period and the beginning of the next, overtime rates will apply to the hours worked on the succeeding work period within the 24-hour period.
- For those working a 11.25 hour shift, if shifts are scheduled so that there is not a rest period of 12 hours and between the end of one work period and the beginning of the next, overtime rates will apply to the hours worked on the succeeding work period within the 24-hour period.
- For those working a 11.5 hour shift, if shifts are scheduled so that there is not a rest period of 11 hours and 45 minutes between the end of one work period and the beginning of the next, overtime rates will apply to the hours worked on the succeeding work period within the 24-hour period.

Further to Article 17 (Paid Statutory Holidays) of the Master Agreement:

- Employees working a 7.78 hour shift pattern will take stats on the Calendar stat dates or on another mutually agreed upon date. As stats are paid at 7 hours per stat x 12 stats = 84 hours, the Regular Full Time employee would owe 9.36 hours per year (7.78 x 12 = 93.36 84 = 9.36). Employees will be scheduled to work one stat plus 1.58 hours. Or if a stat falls on a scheduled day off, they may elect to not take a day in lieu, they would then owe the balance of 1.58 hours.
- Employees working a 8.75 hour shift pattern will take stats on the Calendar stat or on another mutually agreed upon date. As stats are paid at 7 hours the Regular Full Time employee would owe 21.00 hours per year (8.75 x 12 = 105 84 = 21.00). Employees will be scheduled to work two stats plus 3.5 hours. Or if any stat falls on scheduled days off, they may elect to not take that day in lieu, they would then owe the balance.
- For those working 11.25 hours, stats are to be indicated as the calendar stat or on another mutually agreed date but must be identified for the purposes of potential overtime and paid at 7.0 hours to make up the annual FTE of 1827 hours.
- For those working 11.50 hours, stats are identified on a schedule day off for purposes of overtime and paid as per the Collective Agreement. If a stat

falls on a scheduled day the employee needs to work another day to maintain the FTE hours.

Further to Article 18 (Annual vacation) of the Master Agreement, the annual vacation entitlement shall be converted to hours on the basis of a seven hour day and deducted accordingly.

MEMORANDUM OF UNDERSTANDING #18 BASELINE STAFFING INFORMATION

The Employer will provide the Union with copies of the baseline staffing levels, the regular FTEs and total auxiliary hours for all units/wards/programs by March 31, 2013. This data shall also be provided on an annual basis thereafter.

If the Union has questions or concerns regarding the baseline staffing levels, the Employer will agree to discuss the Union's concerns.

*MEMORANDUM OF UNDERSTANDING #19 RECRUITMENT AND RETENTION COMMITTEE

The parties recognize the value of developing positive recruitment and retention practices within the Public Service.

As such, the Parties agree to establish a Joint Committee consisting of up to four representatives from the Union and up to four representatives from the Employer to review and propose strategies to address recruitment and retention challenges for both regular and auxiliary nurses.

The Committee may review, but are not limited, to the following topics:

- Establishing relationships with nursing education programs and encouraging preceptor opportunities
- Assessing options to market the BC Public Service nursing opportunities more broadly including other labour markets
- Identifying ways to be more flexible in recruiting and retaining auxiliaries
- Coordinating with ministry led recruitment and retention initiatives

The Committee will meet within 3 months of the collective agreement being ratified.

Meetings of the committee shall be held at the call of either party within 21 calendar days of such call.

Local level strategies may be implemented outside of this Committee.

The Recruitment and Retention Committee may make specific recommendations to the bargaining principals regarding strategies and processes to address recruitment challenges as outlined in this Memorandum and any other recommendations it may consider necessary to improve recruitment opportunities and identify creative solutions.

*MEMORANDUM OF UNDERSTANDING #20 RE: ECONOMIC STABILITY DIVIDEND

Definitions

1. In this Memorandum of Understanding:

"Collective agreement year" means each twelve (12) month period commencing on the first day of the renewed collective agreement. For example, the collective agreement year for a collective agreement that commences on April 1, 2014 is April 1, 2014 to March 31, 2015 and each period from April 1 to March 31 for the term of the collective agreement.

"Economic Forecast Council" means the Economic Forecast Council appointed under s. 4 of the *Budget Transparency and Accountability Act*, [S.B.C. 2000] c. 23;

"Forecast GDP" means the average forecast for British Columbia's real GDP growth made by the Economic Forecast Council and as reported in the annual February budget of the government;

"Fiscal year" means the fiscal year of the government as defined in the *Financial Administration Act* [1996 S.B.C.] c. 138 as 'the period from April 1 in one year to March 31 in the next year';

"Calendar year" is a twelve (12) month period starting January 1st and ending December 31st of the same year based upon the Gregorian calendar.

"GDP" or "Gross Domestic Product" for the purposes of this MOU means the expenditure side value of all goods and services produced in British Columbia for a given year as stated in the BC Economic Accounts;

"GWI" or "General Wage Increase" means a general wage increase resulting from the formula set out in this MOU and applied as a percentage increase to all wage rates in the collective agreement on the first pay day after the commencement of the eleventh (11th) month in a collective agreement year;

"Real GDP" means the GDP for the previous fiscal year expressed in constant dollars and adjusted for inflation produced by Statistics Canada's Provincial and Territorial Gross Domestic Product by Income and by Expenditure Accounts (also known as the provincial and territorial economic accounts) and published as "Real Gross Domestic Product at Market Prices" currently in November of each year.

The Economic Stability Dividend

- 2. The Economic Stability Dividend shares the benefits of economic growth between employees in the public sector and the Province contingent on growth in BC's real GDP.
- 3. Employees will receive a general wage increase (GWI) equal to one-half (1/2) of any percentage gain in real GDP above the forecast of the Economic Forecast Council for the relevant calendar year.
- 4. For greater clarity and as an example only, if real GDP were one percent (1%) above forecast real GDP then employees would be entitled to a GWI of one-half of one percent (0.5%).

Annual Calculation and publication of the Economic Stability Dividend

5. The Economic Stability Dividend will be calculated on an annual basis by the Minister of Finance for each collective agreement year from 2015/16 to 2018/2019 and published through the PSEC Secretariat.

- 6. The timing in each calendar year will be as follows:
 - (i) February Budget Forecast GDP for the upcoming calendar year;
 - (ii) November of the following calendar year Real GDP published for the previous calendar year;
 - (iii) November Calculation by the Minister of Finance of fifty percent (50%) of the difference between the Forecast GDP and the Real GDP for the previous calendar year;
 - (iv)Advice from the PSEC Secretariat to employers' associations, employers and unions of the percentage allowable General Wage Increase, if any, for each bargaining unit or group with authorization to employers to implement the Economic Stability Dividend.
- 7. For greater clarity and as an example only:

For collective agreement year 3 (2016/17):

- (i) February 2015 Forecast GDP for calendar 2015;
- (ii) November 2016 Real GDP published for calendar 2015;
- (iii) November 2016 Calculation of the fifty percent (50%) of the difference between the 2015 Forecast GDP and the 2015 Real GDP by the Minister of Finance through the PSEC Secretariat;
- (iv)Direction from the PSEC Secretariat to employers associations, employers and unions of the percentage allowable General Wage Increase, if any, for each bargaining unit or group with authorization to employers to implement the Economic Stability Dividend.

Availability of the Economic Stability Dividend

8. The Economic Stability Dividend will be provided for each of the following collective agreement years: 2015/16 (based on 2014 GDP); 2016/17 (based on 2015 GDP); 2017/18 (based on 2016 GDP); and, 2018/19 (based on 2017 GDP).

Note: The Economic Stability Dividend for 2015/16 was calculated at 0.45%; for 2016/17 at 0.35%; for 2017/2018 at 0.40% and for 2018/2019 0.75%.

INDEX

A

Academic Bonuses
Accessibility - Union14
Accessibility of Employer's Regulations14
Accommodation, Board and Lodging Allowance
Accumulation of Time95
Actions of Patients
Addressing Workplace Violence - Appendix #6
Administrative Provisions
Advance Payment of Group Life Benefits - Info
Appendix B173
Air Travel Insurance
Ambiguity with Master Agreement15
Amending Time Limits - Grievances
Amending Time Limits - Arbitration
Amendments
Ancillary Bonuses
Annual Vacation
Annual Vacation - Auxiliary117
Application of Agreement - Auxiliaries114
Appointment to the Public Service - Info
Åppendix A172
Appraisal Reports
Approved Leave of Absence With Pay During
Vacation
Arbitration
Area of Responsibility - Stewards
Assessment of Professional Competence28
Assistant Leader - Info Appendix F184
Attendance Security
Authorization and Application of Overtime53
Auxiliary Employees
Auxiliary Employees Hired for
Downsizing/Closing Facility – MOU#10196

B

Bargaining Agent	12
Bargaining Unit Defined	
Baseline Staffing Information - MOU#18	
Basic Medical Insurance	91
Benefit Waiting Period Allowance	75
Benefits Continuation	76
Benefits Upon Layoff	78
Bereavement Leave	64

Board & Lodging & Relocation Ex	penses –
MOU#14	
Board Procedure - Arbitration	
Bridging of Service	

С

Call-out Provision	
Chairperson of Union-Mangement Meeting	ng CC2
Chairperson of Local Standing Joint	
Committee	
Changes in Agreement	
Changes in Schedules	
Changes to Shifts and Rotations-MOU#2	HC17
Changing or Trading of Shifts, Rotations	of
Shifts, or Rest Days	HC4
Changing Status	
Check-Off of Union Dues	16
Child Care Expenses	106
Christmas or New Year's Day Off	58
Civil Emergency	69
Classification and Reclassification	107
Classification Appeal Procedure	108
Classification Grade Descriptors	
Classification Referee	
Clothing and Uniforms	128
Commencement of Bargaining	
Communicable Disease	
Compassionate Care Leave	73
Compensation-Escort Duty	HC1
Compensation for Statutory Holidays, Ov	vertime
and Standby	
Compensation for Surplus Time	
Component Structure & Jurisdictional Ar	eas-
Appendix #1	
Composition of the Board of Arbitration.	25
Consecutive Days Off	
Contact While Off-Duty	
Contracting Out	
Conversion of Auxiliary Employees	
Conversion of Hours& Part-Time Entitle	
Copies of Agreement	
Copies of Correspondence	
Copies of Regulations	

D

Death Benefit	
Decision of the Board	
Deemed Resignation	77

Deferred Salary Leave Program-*Info Ap	pen D
	177
Definitions	1
Demotions and Transfers	
Dental Plan	92
Designated Paid Holidays	56
Deviation from Grievance Procedure	23
Disagreement on Meaning of Decision	25
Disagreement Over Recommendations	80
Disciplinary Procedure	
Disciplinary Records	29
Discrimination and Harassment in the	
Workplace - Info Appendix E	184
Discrimination and Sexual Harassment	
Complaint Procedures	10
Dismissal	27
Dismissal or Suspension Grievance	22
Dismissal, Suspension, and Discipline	27
Donor Leave.	
Duties and Responsibilities - Stewards	19
*	

Е

Economic Stability Dividend – MOU#20219
Education Policy
Educational Leave and Assistance
Effective Date of Agreement
Elections
Eligibility Requirements for Benefits119
Employee and Family Assistance Program94
Employment Equity - MOU#8193
Employer's Rights
Employment Insurance
Employment of Relatives
Entitlement - Annual Vacation
Entitlements Upon Return to Work77
Establishment of Local Joint Standing
Establishment of Local Joint Standing CommitteesHC12
CommitteesHC12
CommitteesHC12 Establishment Union-Management Component
CommitteesHC12 Establishment Union-Management Component CommitteesCC2
CommitteesHC12 Establishment Union-Management Component CommitteesCC2 Excessive Work Load127
CommitteesHC12 Establishment Union-Management Component CommitteesCC2 Excessive Work Load127 Exclusions - Appendix 2131
CommitteesHC12 Establishment Union-Management Component CommitteesCC2 Excessive Work Load127 Exclusions - Appendix 2131 Expedited Arbitration26
CommitteesHC12 Establishment Union-Management Component CommitteesCC2 Excessive Work Load127 Exclusions - Appendix 2131 Expedited Arbitration26 Expenses of Board26
Committees.HC12Establishment Union-Management ComponentCommitteesCommitteesCC2Excessive Work Load127Exclusions - Appendix 2131Expedited Arbitration26Expenses of Board26Expiration of Agreement128
Committees
Committees
Committees

F

Failure to Act	20
Failure to Appoint	25
Family Illness	70
Forensic Allowance - Hospital	HC9
Formula for Hourly, Daily and Partial Mor	nth
Calculations	105

G

General Application Dispute	23
General Conditions	123
General Leave	69
Grievances	20
Group Grievance	22
Group Life	92

Η

Health and Convenience	HC11
Health and Welfare	91
Health and Welfare-Auxiliares	115
Her Majesty's Forces	68
Holiday Coinciding with a Day of Vacatio	n 58
Holiday Falling on a Day of Rest	57
Holiday Falling on a Scheduled Work Day	57
Holidays Falling on Saturday or Sunday	57
Hours and Overtime Rates	52
Hours of Work & Work Week-Community	.CC3
Hours of Work Committee	CC3
Hours of Work	49
Household Emergency	69
Human Rights Code	8
Human Rights Code – MOU#6	190
Hygiene Facilities	CC7

I

Identification of a Supernumerary Employe	e42
Identification of Employee to Receive Noti	ce of
Layoff	44
Illness, Injury, Rehabilitation and Workers	
Compensation	63
In-Charge-MOU#1	
Increment Dates	98
Increments - Auxiliaries	121
Increments - Eligibility	98
Indemnity	124
Individual Contracts	13
Info Appendix G - Re: 15th Master Append	ix 6
	186
In-hiring Rates of Pay	99

Injury Pay Provisions	81
Interview Expenses	35
Invalid Articles	16
Invalid Articles - Community	CC1
Invalid Articles - Hospital	HC8
Investigation of Accidents	86
Investigations and Documents	

J

Job Evaluation Plan	.107
Job Sharing - Info Appendix C	.174
Joint Committee - Layoff & Recall	48
Joint Education Committee and Jurisdiction	88
Jurisdictional Units and Seniority Blocks	40

L

Labour Adjustment-MOU#11197	7
Layoff and Recall - Auxiliaries114	ł
Layoff and Recall - Regular Employees)
Leave for Committee Meetings67	7
Leave for Court Appearances 67	/
Leave for Medical and Dental Care70)
Leave for Professional Association/College	
Duties64	ł
Limitations of the Local Standing Joint	
Committee HC14	ł
Local Joint Standing Committee	2
Loss of Seniority	3

Μ

Maternity and/or Parental Leave and/or
Pre-Adoption Leave Allowance Repayment78
Maternity Leave
Maternity Leave Allowance75
Maternity/Parental/Adoption Leave74
Maximum Combined Entitlement75
Meal Allowance102
Meal Allowance-Community CC7
Meal Breaks
Meal TimeHC1
Medical Examination
Meeting of Committee - Union/Management
Component Committe CC2
Meetings of Local Standing Joint
CommitteeHC12
Membership in Professional Bodies16
Ministries, Boards and Agencies - Appendix 4
Misuse of Managerial/Supervisory Authority5

N

New Classifications or Proposed Changes in
Existing Classifications108
New Facilities - MOU#1 188
No Discrimination for Union Activity13
No Employee Liability91
No Layoff to Compensate for Overtime55
Notice Boards
Notice of Layoff and Employee Options Upon
Receipt of Notice44
Notice of Promotions
Notice to Bargain129
Notification - Arbitration24
Notification of Unsuccessful Applicants
Nullification - Grievances24
Nursing Advocacy Committee – Professional
Responsibility- MOU#9 193
Nurses of the Ministry of Children & Family
Development - MOU#1CC9

0

Occupational First Aid Requirements and
Courses
Occupational Health and Safety78
Occupational Health and Safety Committees79
Orientation Period47
Out-of-Pocket ExpensesHC9
Overtime
Overtime-Application CC6
Overtime Entitlement
Overtime Meal Allowance
Overtime Meal Allowance- Hospital HC8
Overtime Meal Break and Allowance-
Community CC6
Overtime On Designated Paid Holidays
Overtime On Travel Status
Overtime-Community CC4

Р

Paid Holidays - Auxiliaries	116
Paid Statutory Holiday	56
Parental Leave	
Parental Leave Allowance	75
Parking	123
Pay Period	
Payment of Salaries and Allowances	96
Payroll Deductions	126
Personnel Files	
Personal Property Damage	91

\mathbf{D}_{1}
Placement and Duties of Employees HC12
Political Activity126
Postings
Postings, Transfers & Secondment
Pre Layoff41
Preamble1
Preamble-Community CC1
Preamble-HospitalHC1
Pre-Arranged Overtime
Precedence of Agreement
Pre-employment Eligibility & Check-off
Administration - MOU#4189
Pre-Placement Adoption Leave76
Previous Policies and Letters of Understanding
Principles - Layoff & Recall
Provincial Joint Occupational Health and Safety
Committee
Purpose of Agreement CC1

R

Recognition and Rights	12
Recognition of Stewards	
Recording of Overtime	
Recruitment & Retention Incentive Ad	justment
– *MOU#15	
Recruitment & Retention Committee -	MOU#19
	218
Reduction of Benefits	67
Redundant Positions	41
Re-employment	32
Rehabilitation Committee	
Rejection During Probation	
Relocation Expenses	
Relocations	
Replacement of Articles Held Invalid	НС8
Requests for Time Off Procedures	
Resignation Prompting - MOU#2	
Responsibilities of the Local Standing.	
Committee	
Responsibilities of the Union-Managen	
Committee	
Respectful Workplace	
Rest Interval	
Rest Periods	49
Resubmission to Arbitration	26
Retirement Allowance	
Return to Former Classification	27
Return to Higher-Paid Classification	
Review of Staffing Decisions	

Review of Workplace Safety Training and
Guidance – MOU#16
Revision of Benefits - MOU#7192
Right to Refuse Overtime
Right to Refuse Unsafe Work79
Rights of Witnesses 15
Role of the Rehabilitation Committee –MOU#13
Rotation Between Wards and Areas HC5
Rotations and AdjustmentHC3

S

Safeguarding Vulnerable People - MOU#3188Salaries	Salaries96Salary During Vacation Period62Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-Hospital112Scope of Agreement38Selection Panels35Seniority-Auxiliaries112Seniority List31
Salaries96Salary During Vacation Period62Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Shift and Rotation51Shift Designation51Shift Designation51Shift Differential51Shift Differential51Shift System51Shift System51Shift System51Shift System51Shifts, Shift Schedules and Shift Rotations. HC258Shifts, Variations and FlexibilityHC6	Salaries96Salary During Vacation Period62Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-Hospital112Scope of Agreement38Selection Panels35Seniority-Auxiliaries112Seniority List31
Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions.112Service Breaks72Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift System58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31
Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions.112Service Breaks72Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift System58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	Salary on Demotion100Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31
Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions.112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift Designation51Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift System58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	Salary on Promotion or Reclassification99Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31
Savings Clause.48Scheduling of Vacation60Scheduling of Vacation-Community.CC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions.112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift AssignmentHC2Shift Designation51Shift Designation51Shift Designation51Shift System51Shift System51Shift System51Shift System58Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Variations and FlexibilityHC6	Savings Clause48Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31
Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions.112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift AssignmentHC2Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift System51Shift Schedules and Shift Rotations.42Shifts, Shift Schedules and Shift Rotations.42Shifts, Shift Schedules and Shift Rotations.42Shifts, Variations and FlexibilityHC6	Scheduling of Vacation60Scheduling of Vacation-CommunityCC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority List31
Scheduling of Vacation-Community.CC6Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority Auxiliaries112Seniority List31Seniority on Applying for Regular Positions.112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift BreakHC2Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift System58Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Variations and FlexibilityHC6	Scheduling of Vacation-Community CC6Scheduling of Vacation-Hospital
Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority-Auxiliaries112Seniority List31Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker73Severance Pay47Sexual Harassment9Sharing of Overtime54Shift AssignmentHC3Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift System51Shift Schedules and Shift Rotations. HC258Shifts, Shift Schedules and Shift Rotations. HC258Shifts, Variations and FlexibilityHC6	Scheduling of Vacation-HospitalHC14Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority30Seniority List31
Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority-Auxiliaries112Seniority List30Seniority List31Seniority on Applying for Regular Positions.112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift And Rotation51Shift BreakHC2Shift BreakHC2Shift Designation51Shift Differential51Shift System51Shift System51Shift Schedules and Shift Rotations.58Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Variations and FlexibilityHC6	Scope of Agreement13Secondment38Selection Panels35Seniority-Auxiliaries112Seniority.30Seniority List31
Secondment38Selection Panels35Seniority-Auxiliaries112Seniority Auxiliaries112Seniority List30Seniority on Applying for Regular Positions.112Separation Allowance63Service Breaks72Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift BreakHC2Shift Designation51Shift Designation51Shift Differential51Shift System51Shift System51Shift Schedules and Shift Rotations.HC2Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Variations and FlexibilityHC6	Secondment38Selection Panels35Seniority-Auxiliaries112Seniority30Seniority List31
Seniority-Auxiliaries112Seniority30Seniority List31Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift BreakHC2Shift Designation51Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC258Shifts, Variations and Flexibility	Seniority-Auxiliaries
Seniority-Auxiliaries112Seniority30Seniority List31Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift BreakHC2Shift Designation51Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC258Shifts, Variations and Flexibility	Seniority-Auxiliaries
Seniority30Seniority List31Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift Designation51Shift Differential51Shift System51Shift System51Shift System58Shifts, Shift Schedules and Shift RotationsHC258Shifts, Variations and Flexibility	Seniority
Seniority List31Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift Designation51Shift Differential51Shift System51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC258Shifts, Variations and Flexibility	Seniority List
Seniority on Applying for Regular Positions. 112Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakShift CyclesShift Designation51Shift Differential51Shift System51Shift System51Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityShifts, Variations and Flexibility	
Separation Allowance63Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift Designation51Shift Differential51Shift System51Shift System51Shift System51Shift System51Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Variations and FlexibilityHC6	Seniority on Applying for Regular Positions. 112
Service Breaks72Service Seniority Tie Breaker33Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift Designation51Shift Designation51Shift System51Shift System51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC258Shifts, Variations and Flexibility	
Service Seniority Tie Breaker	
Severance Pay47Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Sexual Harassment9Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift CyclesHC3Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Sharing of Overtime54Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift CyclesHC3Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Shift and Rotation51Shift AssignmentHC3Shift BreakHC2Shift CyclesHC3Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Shift AssignmentHC3Shift BreakHC2Shift CyclesHC3Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Shift BreakHC2Shift CyclesHC3Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift Rotations.HC2Shifts, Variations and FlexibilityHC6	
Shift Cycles	
Shift Designation51Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Shift Differential51Shift System51Shift Workers58Shifts, Shift Schedules and Shift RotationsHC2Shifts, Variations and FlexibilityHC6	
Shift System	Shift Differential
Shift Workers	
Shifts, Variations and FlexibilityHC6	
	Shifts, Shift Schedules and Shift RotationsHC2
	Shifts, Variations and FlexibilityHC6
Short Term Illness & Injury Plan & LTD Plan -	Short Term Illness & Injury Plan & LTD Plan -
Appendix 3133	
Short-Term Illness and Injury and Long Term	
	Disability
Disability63	Special Leave
Disability	5

Special Leave Limitation	72
Standby Provisions	103
Standing Joint Committee	124
Statement of Duties and Responsibilities	127
Step 1 - Grievances	20
Step 2 - Grievances	21
Step 3 - Grievances	21
Stewards	18
Strain Injury Prevention	82
Strikes and Picket Lines	14
Supervisory Officials	18
Suspension	27
Suspensions Pending Investigation	24

Т

Technical Information	127
Technical Objections	24
Temporary Appointments	
Temporary Substitution	
Term of Agreement	
Termination of Employment	
Terminology	
Terms of Employment	
Time Limit - Grievances	
Time off for Meals	CC5
Training Credits	
Transfer Within Geographic Location	
Transfers Without Posting	
Transfers without Posting – MOU#5	
Transition Committee – MOU#12	

U

Unauthorized Absence
Union and Professional Security16
Union Business or Public Duties - Special Leave
Union Observer
Union to Interview New or Terminating
Employees18
Union-Management Component Committee CC2
Unjust Dismissal or Suspension
Use Personal Vehicle on Employer BusinessCC7

V

Vacation Leave on Retirement	62
Vehicle Allowance	101
Violence In The Workplace	84
Volunteering for Shifts	. HC4

W

Wage Schedules – Appendix - 5	153
Ward and Area Selection-Appendix 1	HC15
Weekly Indemnity	119
Work Day	
Work Day - Hospital	HC1
Workers' Compensation	64





COMMUNITY SERVICES NURSES COMPONENT

TABLE OF CONTENTS

Article		<u>Page</u>
ARTIC	LE 1 - PREAMBLE	1
*1.01 1.02	Purpose of Agreement Terminology	1 1
ARTIC	LE 2 - INVALID ARTICLES	1
ARTIC	LE 3 - UNION-MANAGEMENT COMPONENT COMMITTEE	2
3.01 3.02 3.03 3.04	Establishment Meeting of Committee Chairperson of Meeting Responsibilities of Committee	2 2
ARTIC	LE 4 - HOURS OF WORK AND WORK WEEK	
4.01 4.02 4.03	Hours of Work and Work Week Overtime Hours of Work Committee	4
ARTIC	LE 5 - TIME OFF FOR MEALS	5
ARTIC	LE 6 - OVERTIME	6
*6.01 6.02	Application Overtime Meal Break and Allowance	
ARTIC	LE 7 - SCHEDULING OF VACATION	6
ARTIC	LE 8 - HYGIENE FACILITIES	7
ARTIC	LE 9 - MEAL ALLOWANCE	7
ARTIC	LE 10 - USE OF PERSONAL VEHICLE ON EMPLOYER'S BUSINESS	7
ARTIC	LE 11 - TERM OF AGREEMENT	
*11.01 *11.02 *11.03 11.04 11.05	Expiration of Agreement Notice to Bargain Commencement of Bargaining Changes in Agreement Effective Date of Agreement	
	RANDUM OF UNDERSTANDING #1 NURSES OF THE MINISTRY OF DREN AND FAMILY DEVELOPMENT	9

ARTICLE 1 - PREAMBLE

***1.01 Purpose of Agreement**

(a) Subject to the provisions of the Master Agreement entered into between the Employer and the Union the purpose of this Agreement is to set out those terms and conditions of employment applicable only to employees included in the Community Services Nurses Component defined in Appendix 1(a) of the Master Agreement.

(b) This Agreement is subsidiary to and deemed part of the said Master Agreement.

(c) The provisions of the Master Agreement shall have the same force and effect in this Agreement as if they were included herein.

(d) The parties to this Agreement believe that the provisions thereof encourage the continuing development of an excellent standard of health services to the public.

(e) The salaries for employees covered by this Agreement as negotiated by the parties to the Master Agreement are recorded in Appendix **5** of the Master Agreement.

1.02 Terminology

(a) All terms used in this Agreement shall have the same meaning as defined in the Master Agreement unless otherwise specified in either Agreement.

(b) Whenever the term "Component Agreement" is used it shall be deemed synonymous to the term subsidiary agreement as used in the *Public Service Labour Relations Act*.

ARTICLE 2 - INVALID ARTICLES

In the event that legislation or a court of competent jurisdiction renders null and void or materially alters any provision of this Agreement, the remaining provisions of the Agreement shall remain in effect for the term of the Agreement and the parties hereto shall negotiate a mutually agreeable provision to be substituted for the provision so rendered null and void or materially altered.

ARTICLE 3 - UNION-MANAGEMENT COMPONENT COMMITTEE

3.01 Establishment

The Union and the Employer recognize the initial value of ongoing joint discussions and consultation in matters pertaining to working conditions, employment services and labour-management relations.

The Union-Management Component Committee shall be established to represent each unit specified in Appendix 1(a) of the Master Agreement consisting of equal numbers of employees nominated by the Union and Employer representatives with a minimum of two and a maximum of four from each party. The Committee may use additional persons for technical information or advice. The Committee may establish sub-committee or "ad hoc" committees as it deems necessary and shall set guidelines and operating procedures for such committees.

3.02 Meeting of Committee

The Committee should meet not less than four times each year or at the call of either party at the time and a place fixed by mutual agreement. However, such meetings must be held not later than 30 days after their request has been given. Employees shall not suffer loss of basic pay for time spent on the Committee.

3.03 Chairperson of Meeting

An Employer representative and an employee nominated by the Union shall be designated as joint chairpersons and shall alternate in presiding over meetings.

3.04 Responsibilities of Committee

The Committee shall deal with any matter relevant to the Community Component referred to the Committee by the bargaining principals. Notwithstanding the generality of the above, the Committee may:

(a) conduct studies and make recommendations on matters giving rise to grievances and misunderstandings;

(b) receive complaints with regards to, and make recommendations on matters of safe working conditions;

(c) deal with any other matters affecting the Community Services Nurses Component employees;

(d) the Committee shall not have jurisdiction over salaries or any matter of collective bargaining, including the administration of this Agreement, unless otherwise agreed by the bargaining principals.

ARTICLE 4 - HOURS OF WORK AND WORK WEEK

4.01 Hours of Work and Work Week

(a) Recognizing the particular and unique needs of clients dealing with government-community based health care services and that the provision of such services cannot always be predicted accurately in advance; the parties agree that work schedules for employees engaged in such activities will be scheduled on a flexible basis in the interest of patient care and/or efficiency.

(b) The scheduled hours of work for nurses within this component shall be flexible to a maximum of 140 hours within a four-week period. The Employer will identify each four-week period in advance. The establishment of work schedules shall be by mutual agreement between the Employer and the employees at the local level.

Scheduling Parameters

(1) It is intended that the base schedule to which flexibility is to be applied shall be a seven-hour work day.

(2) In planning the proposed schedule, the seven-hour work day may be altered by mutual agreement if it is in the interest of patient care and/or efficiency or to complete work due to exceptional circumstances. Such a schedule will ensure the work day does not exceed 10 hours per day excluding travel.

The planning may also include the identification of possible day(s) or partial day(s) off. These day(s) are scheduled in anticipation of the employee working sufficient flexible time in excess of the base seven-hour schedule. It is understood that such day(s) off or partial day(s) will in fact be earned.

(3) Once posted the proposed daily schedule of hours can also be altered by mutual agreement if it is in the interest of patient care and/or efficiency or to complete work due to exceptional circumstances. Such scheduling will ensure the work day does not exceed 10 hours per day excluding travel.

(4) The employee shall keep an accurate record of actual hours worked which will be submitted to their supervisor.

(5) The Employer shall make every effort to notify an employee of any anticipated changes to the length of the work day.

(6) The minimum scheduled hours for regular part-time employees shall be as outlined in Clause 29.01.

(7) Employees will not be required to work split shifts without their consent.

(8) To ensure adequate services for the Public and still maximize the number of employees with weekends scheduled off and evenings scheduled off, it may be necessary to schedule, by mutual agreement at the local level, six consecutive days.

(9) In order to provide the flexibility necessary to enable the completion of the required hours of work in each four-week period, it is agreed that no premium or penalty contemplated in Article 15 or 16 of the Master Agreement shall apply where it results from an employee exercising their right to flexible work arrangements pursuant to this Article.

(10) If mutual agreement cannot be achieved pursuant to (2), (3) and (8) above, the regular full-time employee will work seven consecutive hours each day exclusive of meal and rest periods between 8:00 a.m. and 12 midnight, five days a week. The regular part-time employee will be scheduled up to seven hours per day.

4.02 Overtime

(a) Authorized work performed in excess of 140 hours in a designated four-week period shall be compensated at the rate of time and one-half of the

employee's regular rate of pay. The overtime compensation may be taken in cash or time off pursuant to Article 26 of the Master Agreement.

(b) It is understood that every reasonable effort will be made to schedule earned time off within the proposed schedule. Notwithstanding (a) above, in the event that an employee is unable to do so, it will be carried over to the next four-week period where it shall be scheduled off at a mutually agreeable time.

4.03 Hours of Work Committee

(a) The Union and the Employer recognize the value of ongoing joint discussions pertaining to hours of work issues. The Committee shall consist of equal numbers nominated by the Union and the Employer with a minimum of two and a maximum of four from each party.

(b) The Committee shall meet at the call of either party at the time and place fixed by mutual agreement. However, such meetings must be held no later than 30 days after a request has been given.

(c) The Committee shall deal with any matter relevant to the administration of the Community Component Article 4 referred to the Committee by either party to this Agreement. The terms of reference of the Committee are to:

(1) determine whether the current provisions properly and adequately meet patient care interests and as well provide for employee personal interests;

(2) make recommendations to the bargaining principals on matters requiring clarification of the hours of work provisions. This may result in issuing documents of clarification to the field on matters causing misunderstanding.

(3) receive complaints with regard to, and make recommendations on matters pertaining to Hours of Work.

ARTICLE 5 - TIME OFF FOR MEALS

All employees covered by this Agreement shall receive not less than 30 minutes and not more than one hour meal time, scheduled at approximately the mid-point in the work day.

ARTICLE 6 - OVERTIME

*6.01 Application

All time worked in excess of the daily or weekly work schedules, as defined in Article 4 of this Agreement shall be compensated at applicable overtime rates in accordance with the Master Agreement.

6.02 Overtime Meal Break and Allowance

Where an employee is required to commence overtime work within two hours of the end of a regular work day, they shall be deemed to have worked overtime continuously. In addition, they shall be given a meal break of not less than one-half hour and be provided with a meal or \$15.30.

ARTICLE 7 - SCHEDULING OF VACATION

 (a) (1) Employees can take annual vacation during the entire calendar year. Subject to operational requirements all employees shall be allowed to take at least four weeks of their vacation entitlement during the period May 1 to September 30 inclusive, which shall be defined as the prime-time vacation period.

(2) For those employees who have more than four **weeks'** vacation entitlement, the Employer shall make every reasonable effort, subject to operational requirements, to allow such employees to take their complete vacation entitlement during the prime-time vacation period if they so desire.

(3) Employees may submit their requests for vacations **any time** prior to February 1, the date when the formal notice for vacation requests for the current year is posted. All employee requests for vacation selection shall be completed by March 1.

(4) All employees must have exercised their seniority rights by February 20th. An employee who does not exercise their seniority rights by February 20th shall not be entitled to exercise their rights in respect to any vacation time previously selected by an employee with less seniority.

(5) The final date for posting the completed vacation schedule shall be April 1st of each year.

(b) All sections of Clause 18.02 (b) (c) (d) and (e) of the Master Agreement will apply in the scheduling of vacation.

ARTICLE 8 - HYGIENE FACILITIES

(a) In the interest of public health and hygiene, washing facilities shall be provided in accordance with the Occupational Health and Safety Regulations.

(b) Lockers which may be locked shall be provided wherever practicable for all employees, required to change their clothes.

ARTICLE 9 - MEAL ALLOWANCE

Employees who are required to be away from their base facility on Employer's business shall receive a meal allowance when:

(a) they are required to work the major part of the work day beyond a radius of 32 kilometers from the facility; or

(b) at the time of the scheduled meal break they are more than 30 minutes travel time from their base facility; or

(c) they are required in the performance of their official duties to purchase a meal.

ARTICLE 10 - USE OF PERSONAL VEHICLE ON EMPLOYER'S BUSINESS

If a vehicle is not provided by the Employer, it is a condition of employment for certain community nurses to provide a personal vehicle for use on the Employer's business. Such nurses will be identified by the Employer.

ARTICLE 11 - TERM OF AGREEMENT

*11.01 Expiration of Agreement

This Agreement covers the period from and including **April 1, 2014** to and including midnight, March 31, **2019**. All terms and conditions of this Agreement shall remain in full force and effect after March 31, **2019** until the Union gives notice of strike and a strike occurs, or until the Employer gives notice of lockout and lockout occurs, or until a new or amended Agreement comes into force, whichever is earliest, and as may be provided by statute.

*11.02 Notice to Bargain

(a) This Agreement may be opened for collective bargaining by either party giving written notice to the other party on or after January 1, **2019** but in any event not later than midnight January 31, **2019**.

(b) Where no notice is given by either party prior to January 31, **2019**, Clause **11.03** of this Article applies, as if notice has been given.

*11.03 Commencement of Bargaining

Where a party to this Agreement has given notice under Clause **11.02** of this Article, the party shall within 30 calendar days after the notice was given commence collective bargaining.

11.04 Changes in Agreement

Any change deemed necessary in this Agreement may be made by mutual agreement at any time during the life of this Agreement.

11.05 Effective Date of Agreement

The provisions of this Agreement shall come into full force and effect on the date of signing of the Master Agreement, except where otherwise specified.

MEMORANDUM OF UNDERSTANDING #1 NURSES OF THE MINISTRY OF CHILDREN AND FAMILY DEVELOPMENT

Recognizing the unique nature of nursing services provided in correctional facilities, work schedules for nurses employed in such facilities may be developed by mutual agreement of the Employer and the employees at the local level subject to the approval of the bargaining principals. Shift schedules may be based on either a seven or seven and one-half hour day.

In the event that mutual agreement cannot be reached, employees shall continue to work shifts and rotations currently in effect.





HOSPITAL SERVICES NURSES COMPONENT

TABLE OF CONTENTS

<u>Article</u>	Pag	<u>e</u>
ARTICI	LE 1 - PREAMBLE AND DEFINITIONS	1
1.01	Preamble	1
ARTICI	LE 2 - HOURS OF WORK	1
2.01 2.02 2.03	Work Day Meal Time Compensation - Escort Duty	1
ARTICI	LE 3 - SHIFT AND ROTATION	2
3.01 3.02 3.03 3.04 *3.05 3.06 3.07 3.08 3.09	Shifts, Shift Schedules, and Shift Rotations	2 3 3 4 4 5
ARTICI	LE 4 - SAVING CLAUSE	8
4.01 4.02	Invalid Articles Replacement of Articles Held Invalid	
ARTICI	JE 5 - ALLOWANCES	8
5.01 5.02 *5.03	Overtime Meal Allowance Out-of-Pocket Expenses Forensic Allowance	9
ARTICI	LE 6 - EMPLOYEE/PATIENT WELFARE 1	1
6.01 6.02	Health and Convenience	1
ARTICI	LE 7 - LOCAL STANDING JOINT COMMITTEE 1	2
7.01 7.02 7.03 7.04 7.05	Establishment of Committees1Meetings of Local Standing Joint Committee1Chairperson of Meeting1Responsibilities of the Committee1Limitations1	223
*ARTIC	LE 8 – SCHEDULING OF VACATION 1	.4
ARTICI	LE 9 - TERM OF AGREEMENT 1	.4

*9.01	Expiration of Agreement	
	Notice to Bargain	
9.03	Commencement of Bargaining	15
9.04	Changes in Agreement	15
	Effective Date of Agreement	
*APPEND	DIX 1 WARD AND AREA SELECTION	15
	DIX 1 WARD AND AREA SELECTION RANDUM OF UNDERSTANDING #1 IN CHARGE	

ARTICLE 1 - PREAMBLE AND DEFINITIONS

1.01 Preamble

The purpose of this Agreement is to set out those terms and conditions of employment applicable only to employees included in the Hospital Services Nurses Component as described in Appendix 1 (b) of the Master Agreement.

ARTICLE 2 - HOURS OF WORK

2.01 Work Day

The hours of work shall be 7 1/2 consecutive hours, interrupted only by the meal break as provided in Clause 2.02.

2.02 Meal Time

Meal time of a minimum of 30 minutes to a maximum of 45 minutes shall be scheduled during an employee's work day.

2.03 Compensation - Escort Duty

An employee required to act as an escort to a patient will be compensated as follows:

(1) All hours in excess of seven and one-half in any one day (or multiples of seven and one-half hours for two or more days) will be paid at overtime premium rates, providing they return by the first available transport system, after surrendering their charge.

(2) Where the employee has elected and has been given permission to return at a later time or date, the employee shall be on unpaid leave of absence from the time of surrendering their charge to the time of commencing their return journey, at which time they will commence their regular rate of pay.

(3) In respect of (2) above, an employee may utilize vacation and/or Article 26 credits to avoid loss of earnings.

(4) In the event of the employee being unable to return by any transport system as a result of adverse weather conditions following the surrendering of their charge, they will be reimbursed for such accommodation and subsistence expenses in accordance with the rates applicable to travel status. Under such circumstance, overtime payments are inappropriate, but there shall be no loss of basic pay.

(5) Where, however, circumstances of (4) are appropriate but the employee's charge has not been surrendered, e.g. travel restrictions occur between connections en route, they are considered to be still on duty and therefore will be subject to compensation shown under subsection (1) above.

ARTICLE 3 - SHIFT AND ROTATION

3.01 Shifts, Shift Schedules, and Shift Rotations

(a) All employees covered by the provisions of this Agreement shall continue to work shifts and rotations currently in effect.

(b) Once an employee is assigned to a specific rotation of shifts and days off, the Employer will make every effort to ensure that this rotation is maintained.

(c) Notwithstanding the provisions of subsections (a) and (b), an employee may be required to work five consecutive days to provide for the changing of shifts and (or) rotations. It is agreed that the interruption to the four days on and two days off shift pattern shall be kept to a minimum.

(d) To meet the principle needs of the Service as defined under Clause 1.01 of the Master Agreement, changes in shifts, shift schedules, and shift rotations may be determined at Local Standing Joint Committee Level, subject to the written approval of the Bargaining Principals.

3.02 Shift Break

If shifts are scheduled so that there is not a period of at least 15 hours and 45 minutes between the end of one work period and the beginning of the next, overtime rates will apply to hours worked on the succeeding work period within the 24-hour period.

3.03 Shift Cycles

(a) All shift cycles shall not last less than 12 calendar days, nor more than 30 calendar days. The foregoing shift cycle lengths may be lessened or extended to allow a shift change on employees' scheduled days off.

(b) By mutual agreement between an employee and their supervisor, an employee may work two or more consecutive cycles on the night shift.

(c) Supervisory personnel (Nurse Level 7 and 9) on their own initiative and with agreement from local management, may reduce or increase their shift cycle lengths.

3.04 Shift Assignment

Rosters covering a minimum of 24 days shall be posted 14 calendar days in advance, and any subsequent change shall be subject to the provisions of Clause 15.05 of the Master Agreement.

***3.05** Rotations and Adjustment

(a) There shall be no pay back for short fall in the shift systems.

(b) Shift employees shall work a pattern of four **days'** work and two days off (4-2 shift pattern).

(c) All other employees shall work a pattern of five **days'** work and two days off (5-2 shift pattern).

(d) (1) Shift employees working the 4 - 2 shift pattern where the length of the scheduled work shift is 7 hours and 30 minutes, exclusive of meal break, will earn a surplus of time off the equivalent of 10 working days per year (in addition to any compensation or premium due under Clause 17.07 of the Master Agreement) which shall be paid in accordance with the provisions of Clause 26.02 of the Master Agreement. Effective January 1, 2018 the surplus time off will be the equivalent of 11 working days per year.

(2) Employees working the 5 - 2 shift pattern where the length of the scheduled work shift is 7 hours and 30 minutes exclusive of meal breaks

will earn a surplus of time off the equivalent of 17 working days per year (in addition to any compensation or premium due under Article 17 of the Master Agreement). The time shall be paid by permitting the employee to schedule off one of the days owing each month at a time convenient to the employee but subject that minimum operational requirements are maintained, and the remaining five days shall be paid in accordance with Clause 26.02 of the Master Agreement.

(3) Part-time employees will earn a surplus of hours and receive repayment on a prorated basis.

3.06 Changing or Trading of Shifts, Rotations of Shifts, or Rest Days

With suitable advance notice to their supervisor, employees shall normally be allowed to trade shifts, rotations, and (or) rest days, providing no extra cost to the Employer will result and that:

(a) the replacement employee is of a similar and (or) acceptable level of ability, qualifications, and experience; or

(b) that coverage of the ward or area will not be adversely affected.

3.07 Volunteering for Shifts

(a) Volunteers for any shift will be utilized whenever and wherever practicable, providing this does not involuntarily restrict the rotation of other employees, except where an employee is engaged in a professional development program.

(b) All shift staff who are prepared to remain indefinitely on the afternoon or night may be granted permission to do so, providing this:

(1) is administratively practicable;

(2) in no way adversely affects the coverage of the ward or unit, or patient treatment program;

(3) does not interfere with the employee's professional development and knowledge of various conditions and programs.

With 30 **days'** notice and after four months' service, including such notice, the Employer may recall employees to rotation.

3.08 Rotation Between Wards and Areas

(a) Employees can normally expect (and will be expected) to remain on the same ward or area for approximately 18 months. Ward staff members will be expected to rotate through all types of wards. No ward staff shall be permitted to remain on a ward if by doing so they restrict the rotation of others.

(b) Ward staff assigned as in (a) above may request a move at any time after three months after assignment to a given ward if the request is supported by sound reasons. Before such a request is refused, it shall be referred to the appropriate excluded manager. At the end of 18 months or more on a given ward, the employee requesting transfer to another ward or another type of ward may be transferred at the next shift change, but in any case not more than 60 days following the request. Any application issues arising from this Clause shall be referred to the Local Standing Joint Committee.

(c) Supervisors after 18 months in the area, and with 60 days' notice, may request assignment to other areas, and such requests shall be granted wherever possible.

(d) Requests made pursuant to (b) and (c) above shall not be unreasonably denied.

(e) Temporary reassignment to cover vacation, illness, and similar absence of employees shall be on a compulsory basis of rotating the employees involved, although timely volunteering will be considered.

(f) Save for the closing of a hospital, clinic or other facility, no employee shall be transferred or sent relieving without their consent to a ward or area in another hospital.

(g) The Employer shall notify an employee at least 14 calendar days in advance of any move to another ward or area except in the case of temporary assignment made necessary by operational requirements, in which case as much reasonable notice as possible will be provided.

3.09 Shifts, Variations and Flexibility

- (a) (1) Notwithstanding any provision to the contrary contained in Article3, the following hours of work provisions shall apply for the following enumerated classes of employees.
 - program nurse positions as agreed by the parties
 - Central Sterile Supply, all facilities
 - Admissions Nurses, all facilities
 - Staff Trainers, all facilities
 - (Any other position agreed to by the parties from time to time.)

(2) For existing incumbents of the topic functions there shall be two work schedule options, either current schedules in effect, or, by mutual agreement at the local level, a seven-hour day, day shift, Monday to Friday, with statutory holidays off.

(3) In the event of vacancies or newly created positions in subclause (1), the Employer at the local level may establish the work schedule as being a seven-hour day, day shift, Monday to Friday with statutory holidays off. This information will be included in any relevant posting.

(b) Notwithstanding any provisions to the contrary in Article 3, the representatives of the Employer and the Union at the local level may establish variations to the current shifts and rotations. Any varied work schedule shall meet the following criteria:

(i) All varied schedules must conform to the program requirements necessary for the delivery of service.

(ii) All work schedules shall meet the annualized hours of work.

(iii) Any work schedule shall not contain scheduled overtime.

(iv) Two-thirds of the affected incumbent nurses must agree to the varied work schedule.

(v) The varied work schedule shall be reflected in the form of a memorandum between the bargaining principles before it is implemented.

(vi) In the event that either the Employer or the Union at the local level, or either of the bargaining principles, determine that such a changed work schedule is no longer desirous, then said schedule shall be cancelled upon one month's notice.

(c) By mutual agreement between the individual employee(s) and the Employer at the local level, employees classified at Level 7 and 9 may schedule their work on a flexible basis. Scheduling of work on a flexible basis shall meet the following criteria:

(i) The program requirements necessary for the delivery of service are maintained.

(ii) The annualized hours of work must be met. The employee may be requested to keep an accurate record of actual hours worked which will be submitted to their supervisor.

(iii) No premium or penalty contemplated on Articles 15 (Shift and Rotation) or 16 (Overtime) of the Master Agreement shall apply where it results from an employee self scheduling their work on a flexible basis. Hours worked in excess of the seven or the seven and one-half in a work day initiated or scheduled by the Employer shall attract the premiums contemplated in Articles 15 and/or 16.

(iv) In the event either the employee(s) or the Employer at the local level, or either of the bargaining principles, determine such flexible scheduling of work is no longer desirous, then it shall be cancelled.

ARTICLE 4 - SAVING CLAUSE

4.01 Invalid Articles

If any article or section of this Agreement or of any riders hereto should be held invalid by operation of law or by any tribunal of competent jurisdiction, or if compliance with or enforcement of any article or section should be restrained by such tribunal pending a final determination as to its validity, the remainder of this Agreement and of any rider thereto, or the application of such article or section to persons or circumstance other than those as to which it has been held invalid or as to which compliance with or enforcement of has been restrained, shall not be affected.

4.02 Replacement of Articles Held Invalid

In the event that any article or section is held invalid, or enforcement of or compliance with which has been restrained, as above set forth, the parties affected thereby shall enter into immediate collective bargaining negotiations, upon the request of either party for the purpose of arriving at a mutually satisfactory replacement for such article or section during the period of invalidity or restraint. If the parties do not agree, they shall submit the dispute to the procedure as outlined in Article 9 of the Master Agreement.

ARTICLE 5 - ALLOWANCES

5.01 Overtime Meal Allowance

(a) An employee who is required to work a minimum of two and one-half consecutive hours overtime before or after, but joined to their scheduled hours of work, shall be provided with a meal or shall be reimbursed in the amount of \$15.30.

A meal break of one-half hour with pay at the straight time rates shall be provided.

(b) An employee who is required to work continuously six and one-half or more hours overtime, excluding the overtime meal break in clause (a), shall receive a further meal or allowance and break as in clause (a). This entitlement shall continue upon the completion of every three hours of overtime worked thereafter.

(c) This Clause shall not apply to an employee who is on travel status which entitles them to claim lodging and (or) meals.

5.02 Out-of-Pocket Expenses

Employees performing their duties within their headquarters area (not on travel status) but away from their hospital, clinic, or other facility, may claim reasonable out-of-pocket expenses, such as meals, subject to the approval of the Employer.

*5.03 Forensic Allowance

(a) Employees employed in the following wards or areas or any eventual successor wards or areas shall be paid a forensic allowance in accordance with the following:

(1) Maximum Security - A forensic allowance of one extra day off per month:

Youth Forensic Psychiatric Services - Inpatient Assessment Unit

(2) Combined Maximum/Medium Security - A forensic allowance of one extra day off every two months:

- Case Management Services Nurse Case Coordinators only
- Complex Coordinator (if established)

(3) Medium Security - A forensic allowance of \$80.00 per month:

(a) (i) Burnaby Youth Custody Services (ii) Prince George Youth Custody Services

(b) Maples Adolescent Treatment Centre

 Crossroads Forensic Treatment Program (when operating and clients are required to be "locked-in")

(b) Employees entitled to an extra day off shall take the same in conjunction with a regularly scheduled two-day break, and such time shall be posted on the roster at the time the regular schedule of work days is posted.

(c) Where an employee has worked part of a month on one security ward or area and part of a month on another security ward or area, the employee shall receive the allowance appropriate to the ward or area on which the greater length of time was worked. Where an employee has worked an equal amount of time between two security wards or areas, the higher security allowance shall apply.

(d) Where an employee under (a)(3)(b) has worked a full shift directly as a primary care provider with locked forensic clients, the employee shall receive a \$4.00 allowance per shift to a monthly maximum of \$80.00. Effective the first full pay period following date of ratification of the 16th Master Agreement, the employees shall receive a 53 cents allowance per hour to a bi-weekly maximum of \$36.80.

(e) (1) Authorized absences of less than five scheduled working days in a month, or 10 days in two months under (a)(2) shall not affect an employee's entitlement to the allowance. Should an employee be on authorized absence for more than five scheduled working days in any month, or 10 days over two months under (a)(2) the allowance shall be reduced on a pro rata basis and shall be the product of a numerator being the actual number of days worked, the denominator being the number of scheduled work days for that employee in the given month, and the multiplicand being the full amount of the appropriate allowance. In such instances, the maximum and combined security allowances shall be converted to a cash payment.

(2) No employee shall suffer any loss of allowance because of absence from their normal work site of five days or less in one month, or 10 days or less in two months under (a)(2), due to temporary assignment by the Employer or due to union business. An employee may combine the absences permitted in sub clauses (1) and (2) without reduction of the allowance; except that an employee may not combine the Employer assignment and union business in the same calendar month under the provisions of sub clause (2) without a prorated reduction in the allowance should the absences in sub clause (2) exceed five working days. (3) Employee entitled to and who take the security allowances of an additional day off shall not have the day charged against the five days of valid absence permitted in sub clause (1).

(f) No employee shall suffer loss of Forensic allowance because of absence from work due to a work-related injury. Employees in the maximum or combined security areas who are absent from work more than five days in a single month due to a work related injury shall receive their allowance as extra pay.

(g) Regular part-time and auxiliary employees, and employees who commence or terminate employment during a month shall receive the appropriate allowance on a pro rata basis and converted to cash, except that any such employees who work 15 full-time days or more in the month shall receive the full allowance.

(h) Should a pay equity program be agreed to between the parties and incorporate the basis for which these allowance are paid, this Article shall be deleted in its entirety upon implementation of any such pay equity program.

ARTICLE 6 - EMPLOYEE/PATIENT WELFARE

6.01 Health and Convenience

(a) Rooms suitable for consumption of beverages and meals will be provided.

In facilities where meal service is currently provided for employees, meal service shall continue to be provided and be available on the day and afternoon shift at times which are concurrent with the established kitchen hours. Cost for such meals shall be the basis of raw food plus services costs.

(b) A room(s) will be set aside away from the work station to provide staff privacy during breaks. The room shall be minimally equipped for making hot and cold beverages. Chattels will be provided at no cost to the employees. The room will be equipped with one or more couches and (or) chairs.

(c) Lockers which can be locked shall be provided for the safekeeping of personal belongings. These lockers may be located in the staff rooms.

6.02 Placement and Duties of Employees

(a) No employee shall be required to work alone on a ward if the employee indicates to the supervisor that the employee believes the situation may be unsafe.

(b) In concert with other affected bargaining units, placement of male and female employees on shift and rotations as well as wards and units shall, as a minimum, adhere to the following principles:

(1) The modesty, safety, and health of the patient must be assured;

(2) The safety and health, as well as reasonable modesty, of the employee must be assured.

(c) The local union organization or local management may refer differences of interpretation of modesty, safety, and (or) health to the Local Standing Joint Committee referred to in Article 7.

ARTICLE 7 - LOCAL STANDING JOINT COMMITTEE

7.01 Establishment of Committees

A Local Standing Joint Committee shall be established for each unit specified in Appendix 1(b) of the Master Agreement, consisting of equal numbers of employees nominated by the Union and Employer representatives with a minimum of two and a maximum of four from each party. The Committee may use additional persons for technical information or advice.

7.02 Meetings of Local Standing Joint Committee

The Committee shall meet at least once each month at a mutually agreeable time and place. Employees shall be without loss of basic pay for time spent on the Committee

7.03 Chairperson of Meeting

An Employer representative and an employee nominated by the Union shall be designated as joint chairpersons and shall alternate in presiding over meetings.

7.04 **Responsibilities of the Committee**

(a) <u>Assigned Responsibilities</u>

(1) The Committee shall deal with responsibilities specifically assigned to it under the terms of the Master and this Agreement.

(2) Where the Master Agreement or this Agreement use the terms "by mutual agreement", "local arrangements", or similar terms, and where it is evident that these terms do not apply to another committee(s) of the Union and the Employer nor to arrangements between an individual employee and their supervisor, these matters shall be referred to the Local Standing Joint Committee.

(3) Unless otherwise expressed in the Master or this Agreement, all recommendations of the Committee are subject to the approval of the Bargaining Principals. Where the members of the committee cannot come to an agreement on matters referred to it, the opposing positions will be reduced to writing and referred to the Bargaining Principals for solution.

Where the Bargaining Principals fail to agree, the question will be handled under the provisions of Article 8 of the Master Agreement.

(4) The Committee may appoint standing or ad hoc sub-committees to deal with specialized problems. The sub-committee(s) shall be responsible to the Local Standing Joint Committee.

(5) There is no restriction against the Local Standing Joint Committee acting as the Safety Committee, provided that the provisions of the Workers' Compensation Act and regulations are complied with.

(b) <u>Continuing Responsibilities of the Committee</u>

The Committee may concern itself with the following general matters:

(1) Reviewing suggestions from employees and questions of working conditions and service (but not grievances), so that better relations shall exist between the Employer and its employees.

(2) Correcting conditions causing grievances and misunderstandings.

(3) Studying issues arising from changing conditions and concepts in nursing care and treatment of patients.

7.05 Limitations

The Committee shall not have jurisdiction over wages or any other matter of collective bargaining, including the administration of this Agreement, except the recommendations to be made under Clause 7.04. The Committee shall not supersede the activities of any other committee of the Union or the Employer and does not have the power to bind either the Union or its members or the Employer to any decision or conclusions reached in their discussions. The Committee shall have the power to make recommendations to the Bargaining Principals.

*ARTICLE 8 – SCHEDULING OF VACATION

Employees can take annual vacation during the entire calendar year. Subject to operational requirements all employees shall be allowed to take at least four weeks of their vacation entitlement during the period May 1 to September 30 inclusive, which may be defined as the prime-time vacation period.

For those employees who have more than four weeks' vacation entitlement, the Employer shall make every reasonable effort, subject to operational requirements, to allow such employees to take their complete vacation entitlement during the prime-time vacation period if they so desire.

ARTICLE 9 - TERM OF AGREEMENT

***9.01** Expiration of Agreement

This Agreement covers the period from and including January 1, **2014** to and including midnight, March 31, **2019**. All terms and conditions of this Agreement shall remain in full force and effect after March 31, **2019** until the Union gives notice of strike and a strike occurs, or until the Employer gives notice of lockout and lockout occurs, or until a new or amended Agreement comes into force, whichever is earliest, and as may be provided by statute.

***9.02** Notice to Bargain

- (i) This Agreement may be opened for collective bargaining by either party giving written notice to the other party on or after January 1, **2019**, but in any event not later than midnight January 31, **2019**.
- (ii) Where no notice is given by either party prior to January 31, 2019, Clause 9.03 of this Article applies, as if notice has been given.

9.03 Commencement of Bargaining

Where a party to this Agreement has given notice under Clause **9.02** of this Article, the party shall within 30 calendar days after the notice was given commence collective bargaining.

9.04 Changes in Agreement

Any change deemed necessary in this Agreement may be made by mutual agreement at any time during the life of this Agreement.

9.05 Effective Date of Agreement

The provisions of this Agreement shall come into full force and effect on the date of signing of the Master Agreement, except where otherwise specified.

*APPENDIX 1 WARD AND AREA SELECTION

- 1. Subject to the limiting factors of Clause 3.08 of this Agreement every attempt will be made to place employees on the ward or in the area of their choice, subject to the provisions of proper and adequate patient coverage.
- 2. In hospitals where nursing procedures are required to be performed by members of a specific sex, every reasonable effort will be made to ensure that a member of the appropriate sex is available to carry out these procedures.

In cases where specific nursing procedures come into dispute as to the appropriateness of the assignment, such shall be referred to the Local Standing Joint Committee with views, if any, being received from the Human Rights Commission.

Failure to reach agreement shall be dealt with under the provisions of Clause 7.04(a)(3) of this Agreement.

- 3. Every attempt shall be made by the Employer to ensure that all employees for maximum security areas will be selected only from volunteers unless the employee was expressly hired for a maximum security area.
- 4. The Employer undertakes to work toward maximizing the professional nursing element in the job content for all employees in the bargaining unit.

*MEMORANDUM OF UNDERSTANDING #1 IN CHARGE

(a) For the facility specific designations as agreed and listed below, employees classified at the Nurse 4 level only, shall receive a special allowance of \$9.38 per shift worked where such employees are "in-charge". This allowance will apply when the majority of the shift is worked and shall not be pro-rated.

(b) Where the incumbent of a higher paying position is temporarily absent for three or more hours, employees classified at the Nurse 4 level only and "incharge" shall receive a special allowance of \$1.25 per hour worked.

(c) In the circumstances of (b) above, Clause 27.08 (Temporary Substitution) will only apply where the temporary absence is known in advance to exceed 16 consecutive shifts. If the Employer determines that it will not designate an employee pursuant to Clause 27.08, then (a) above will apply.

(d) Where a position is vacant, the Employer shall determine whether to designate an employee to substitute pursuant to Clause 27.08 or make temporary appointments pursuant to Clause 27.09. However, where an Employer decision in this regard results in non-compliance with the facility designations listed below, then (a) above will apply.

(e) An employee shall not receive both the "in-charge" special allowance and substitution pay on any given shift.

(f) "In-charge" means assuming supervisory and/or administrative duties over and above regular clinical duties, but is not a designation pursuant to Clause 27.08 to assume the principal duties of a higher paying position. (g) The "in-charge" special allowance payable under (a) and (b) shall be paid according to classification seniority within the facility designations described below.

(h) Organizational changes or changes to existing staffing models in the affected facilities, may necessitate revision of the facility designations as required. Discussions in this regard will commence at the call of either party.

(i) The Employer will consult with the Union respecting expected duties for employees who are assigned "in-charge" responsibilities as defined in (g) above.

***Facility Designations**

- 1. Nothing in this Memorandum of Understanding is intended to detract from the Employer's ability to assign nurse supervisors (defined as bargaining unit Nurse Levels 4, 5, 7, 9 or excluded nurse) to any shift in any facility, nor to designate Nurse Level 4 for temporary substitution pursuant to Clause 27.08 or make temporary appointments pursuant to Clause 27.09.
- 2. Inpatient Assessment Unit IAU

On each full day, afternoon or night shift where no nurse supervisor as defined in (1) above is scheduled to work, MOU#1 (a) or (b) shall apply for one Nurse Level 4 per shift.

3. Maples Adolescent Treatment Centre

The employer determines the staffing model and levels of staffing including the supervisory presence required on each shift. It is recognized that due to the staffing model utilized at this worksite, this Memorandum of Understanding would only apply to the night shift.

MEMORANDUM OF UNDERSTANDING #2 CHANGES TO SHIFTS AND ROTATIONS

A. The Union has identified a desire by some members, less than a majority, to have the opportunity to work additional day shifts and to have more weekends off. The Employer has identified a need to have a limited number of additional nurses on duty on Monday to Friday during the day shift in order

to better provide certain programs.

In view of the above,

1. the Parties agree to introduce pilot project(s) where a limited number of regular full time Nurse Level 4 and/or Nurse Level 5 employees will be given the opportunity to accept assignment of a combination of four **days**' work and five **days**' work with the equivalent number of annual days off as provided by a 4-2 shift pattern with the number of consecutive days off being two or more;

2. the maximum number of consecutive days of work shall be five;

all provisions of current Hospital Services Nurses Component Clause
 3.09 shall continue to apply;

4. by mutual agreement at the local level, ward(s) shall be chosen on which to introduce the pilot project;

5. the provisions of the pilot project shall be in writing;

6. the Parties agree to establish a Joint Committee to monitor the implementation and success of the pilot project;

7. should the pilot projects be a success, the Parties may agree to add additional wards to the pilot project.

B. The British Columbia Nurses' Union has identified a desire by some members to work an extended work day.

In view of the above, the BCNU and Employer shall investigate the feasibility of introducing mixed shift lengths on the number of wards on which there is a mutual agreement at the local level. The approach shall be that those employees who wish shall continue to work the 7.5 hour day, 4-2 shift rotation or a variation thereof, while others (volunteers) shall work an extended work day. The provisions of current Hospital Services Nurses Component Clause 3.09 shall continue to apply.